

Host-seeking behaviour of Anopheles mosquitoes in response to olfactory and visual cues

Margaret Mendi Njoroge

Propositions

- An odour-baited trap in close proximity of the eave diminishes the effectiveness of the repellent transfluthrin applied on eave fabric in reducing the number of outdoor-biting mosquitoes in the peri-domestic area. (this thesis)
- In the presence of potent spatial repellents, mosquito species differ in how they respond to attractive synthetic lures. (this thesis)
- 3. Application of agricultural pesticides increases the risk of schistosomiasis.
- 4. Aquaculture leads to reduction in natural fish populations in freshwater bodies.
- 5. Political affiliations diminish knowledge acquisition and application.
- 6. Accessibility to affordable internet services increases the risk of under-age pregnancies.

Propositions belonging to the thesis, entitled:

Host-seeking behaviour of *Anopheles* mosquitoes in response to olfactory and visual cues Margaret Mendi Njoroge Wageningen, 14 April 2021

Host-seeking behaviour of *Anopheles* mosquitoes in response to olfactory and visual cues

Margaret Mendi Njoroge

Thesis committee

Promotors

Prof. Dr W. Takken Personal chair at the Laboratory of Entomology Wageningen University & Research

Prof. Dr J.J.A. van Loon Personal chair at the Laboratory of Entomology Wageningen University & Research

Co-promotor

Dr A. Hiscox Head of Research Programmes, ARCTEC, London School of Hygiene and Tropical Medicine, UK

Other members

Prof. Dr M. Naguib – Wageningen University & Research Prof. J.G. Logan – London School of Hygiene and Tropical Medicine, UK Dr. N.O. Verhulst – University of Zürich, Switzerland Dr. J.G. de Boer – Netherlands Institute of Ecology, Wageningen

This research was conducted under the auspices of the C.T. de Wit Graduate School for Production Ecology & Resource Conservation

Host-seeking behaviour of Anopheles mosquitoes in response to olfactory and visual cues

Margaret Mendi Njoroge

Thesis

submitted in fulfilment of the requirement for the degree of doctor at Wageningen University by the authority of the Rector Magnificus, Prof. Dr A.P.J. Mol, in the presence of the thesis committee appointed by the Academic Board to be defended in public on Wednesday 14 April 2021 at 4 p.m. in the Aula.

Margaret Mendi Njoroge Host-seeking behaviour of *Anopheles* mosquitoes in response to olfactory and visual cues 174 pages. PhD thesis, Wageningen University, Wageningen, the Netherlands (2021) With references, with summary in English ISBN: 978-94-6395-725-0 DOI: htts://doi.org/10.18174/542244

TABLE OF CONTENTS

Chapter 1	7
General Introduction	
Chapter 2	17
Evaluating putative repellent 'push' and attractive 'pull' components for manipulating the	
odour-orientation of malaria vectors in the peri-domestic space	
Chapter 3	53
Sight vs. smell: Dynamics of malaria mosquito attraction response in the presence of spatial	
repellents	
Chapter 4	75
Less is more: repellent-treated eave wrap as a substitute for eave screen for indoor and	
outdoor protection of malaria mosquitoes	
Chapter 5	95
Assessment of push-pull mosquito control for indoor and outdoor protection	
against malaria vectors	
Chapter 6	115
General Discussion	
Summary	129
References	133
List of abbreviations	163
List of publications	165
Acknowledgements	167
Curriculum vitae	169
PE & RC Training and Education Statement	171

General Introduction

Introduction

Success in malaria vector control in recent years has been attributed largely to the widespread distribution and utilization of long lasting insecticide-treated nets (LLINs) and indoor residual spraying (IRS)(WHO, 2019). The disruption of contact between the human host and the vector remains a very effective mechanism of controlling mosquito-borne diseases when applied in combination with other tools such as effective diagnosis, prevention of mother to child transmission and prompt medical treatment (WHO, 1975; Lindsay *et al.*, 2002; Russell *et al.*, 2011; Bhatt *et al.*, 2015b; WHO, 2019).

While LLINs and IRS have contributed to the reduction in malaria disease burden, the widespread emergence of mosquitoes resistant to pyrethroids, the main insecticide group used on LLINs and IRS, and other insecticides threatens the continued success of these tools (Townson *et al.*, 2005; Tanner *et al.*, 2008; Sande *et al.*, 2015; Awolola *et al.*, 2018; Cook *et al.*, 2018; Deletre *et al.*, 2019; WHO, 2019; Hancock *et al.*, 2020). Increased pyrethroid resistance of mosquitoes in malaria endemic areas prompted the modification of formulations used on LLINs to include synergists such as piperonyl butoxide (PBO) that inhibit the enzymes that detoxify insecticides and have increased the effectiveness of existing insecticides (LeClair *et al.*, 2017; Matowo *et al.*, 2017; Gleave *et al.*, 2018; Rakotondranaivo *et al.*, 2018; Birhanu *et al.*, 2019). This solution, however, is considered to be a delay and not a preventive measure to the spread of insecticide resistance, with studies already showing decreasing efficiency of synergist-included insecticide formulations on bed nets (Djouaka *et al.*, 2016; Gleave *et al.*, 2018; Riveron *et al.*, 2019).

Another challenge in malaria control is the apparent shift of dominant malaria vector populations from predominantly indoor-biting and resting populations to those that rest and bite outdoors and/or bite earlier in the night when people are not yet protected by LLINs (Antonio-Nkondjio *et al.*, 2006; Moore *et al.*, 2012; Moiroux *et al.*, 2014; Sougoufara *et al.*, 2014; Moshi *et al.*, 2017; Limwagu *et al.*, 2019; Mburu *et al.*, 2019; Sherrard-Smith *et al.*, 2019). This change in vector behaviour is expected to increase the likelihood of receiving an infectious bite during the evening while conducting outdoor activities, before retiring indoors where protection from either LLINs or IRS is present (Killeen, 2014; Mathania *et al.*, 2016; Moshi *et al.*, 2017). Studies have shown that socio-economic activities such as fishing, cooking and harvesting in rural areas are associated with an increased risk of exposure to outdoor malaria transmission (Sande *et al.*, 2015; Moshi *et al.*, 2017; Moshi *et al.*, 2018). The need to identify alternative tools that can be applied promptly towards management of physiologically and behaviourally resistant outdoor-biting mosquitoes is urgent to further reduce malaria incidence (Sangoro *et al.*, 2014a; Sherrard-Smith *et al.*, 2019).

Mosquitoes utilize very specific chemical and physical cues when conducting various physiological functions such as mating (Vaníčková *et al.*, 2017), location of an ideal breeding habitat (Okal *et al.*, 2013; Lindh *et al.*, 2015; Asmare *et al.*, 2017), identification of a plant for sugar feeding (Foster, 2008; Nyasembe *et al.*, 2012; Nyasembe *et al.*, 2018) and the successful location of a host for a blood meal (Takken *et al.*, 1997; Takken *et al.*, 1999; Carde, 2015; Liu *et al.*, 2019). Understanding the very complex mechanisms involved in host-seeking (Hawkes *et al.*, 2016; Hawkes *et al.*, 2017) is key to developing tools that manipulate mosquitoes to move away from potential human hosts and on to non-human hosts (Donnelly *et al.*, 2015; Njoroge *et al.*, 2017) or to trap and kill devices that mimic human cues (Takken *et al.*, 1999; Mukabana *et al.*, 2002; Qiu *et al.*, 2007; Olanga *et al.*, 2010). Mosquito perception of olfactory cues takes place in receptors located on antennae and maxillary palps which inform the mosquito of the potential availability of the blood meal host prior to utilization of complementary cues of physical nature, *i.e.* temperature, moisture, and air speed, to guide landing and contact chemical cues perceived upon landing on the skin, in particular tastants, that elicit feeding (Takken *et al.*, 1999; Turner *et al.*, 2011; Webster *et al.*, 2015; Raji *et al.*, 2017).

Repellents are chemical compounds that have been in use since antiquity to deter biting insects away from humans usually by interfering with the normal perception and response of insects to humans (Charlwood, 2003; Paluch *et al.*, 2010; Maia *et al.*, 2011; Maia *et al.*, 2018). Usually applied either topically, on clothes or as volatile or spatial chemicals, repellents provide opportunities for controlling malaria (Wilson *et al.*, 2014; Maia *et al.*, 2018). Topical repellents exert their effect on their targets once direct interaction of mosquitoes with treated areas takes place (Norris *et al.*, 2017; Dennis *et al.*, 2019) and are characterized as being only slightly volatile which extends their effective action from a treated surface (usually: the skin)(Grieco *et al.*, 2007; Sathantriphop *et al.*, 2014). Insecticide treated clothes have been shown to reduce the risk of malaria by 50% and can provide protection in areas where LLINs cannot be in use such as in disaster areas (Maia *et al.*, 2018). Volatile or spatial chemicals represent a promising solution in reducing the spread of mosquito-borne disease due to their ability to disrupt normal host-vector interactions within a particular area (Achee *et al.*, 2012; Sangoro *et al.*, 2014a; Norris *et al.*, 2017; Bowman *et al.*, 2018).

Some insecticides have lethal, sub-lethal and excito-repellent properties though the effects of these in vector control are understudied (Chareonviriyaphap *et al.*, 2013). In the face of insecticide resistance, vectors may respond less to toxicity of insecticides due to innate mechanisms that reduce the toxic effects but may still respond to sub-lethal and excito-repellent effects of the same insecticides (Guedes *et al.*, 2017; Cook *et al.*, 2018; Thievent *et al.*, 2019). Sub-lethal concentrations of insecticides present opportunities for application as they reduce host-vector contact even when mosquito populations are

predominantly resistant to that group of insecticides (Cohnstaedt *et al.*, 2011; Bowman *et al.*, 2018). Recently, however, concerns have been raised on how lower insecticide concentrations may still select populations that are more resistant by either increasing fitness or selecting a wide-range of resistance-linked genes that will increase dominance in the population (Guedes *et al.*, 2017; Margus *et al.*, 2019). Insect repellents act by diverting mosquitoes away from attractive targets, thereby interfering with responses to the attractive cues emanated by the host. Repellency reduces vector-host contact and subsequently diminishes the chances of transmission of parasites and viruses to human (Norris *et al.*, 2017).

Results from field studies testing topical repellents show differences in outcome and protection (Maia et al., 2018). In a study done in Senegal, the effect of topical repellents (Para-menthane-3,8-diol (PMD), DEET and icaridine) on the biting rates of mosquitoes was shown to be significant in providing protection on exposed human subjects compared to the control groups when mosquito biting rates were monitored for nine hours after application (Uzzan et al., 2009). However, a study in Tanzania showed that 15% DEET used as a topical repellent in the early evenings was not effective in reducing malaria prevalence (Sangoro et al., 2014a; Sangoro et al., 2014b). The reasons for this were attributed to lack of compliance in application; both in technique and in frequency therefore rendering what could have been an effective tool in preventing host-vector contact to be ineffective for disease prevention and control in this setting (Sangoro et al., 2014a). Frequent re-application of a topical repellent can be impractical or impossible in areas where supplies are limited or where there is lack of proper education and training on proper use. In some cases, even with adequate education, adherence to application regime is low, therefore reducing effectiveness of contact repellents (Sangoro et al., 2014a; Lalani et al., 2016). Moreover, in many tropical areas where mosquito-borne diseases are endemic, high levels of perspiration remove the repellent from the skin, making this approach less effective in the long-run (Barnard, 2005; Wilson et al., 2014; Norris et al., 2017).

Use of spatial repellents instead of topical repellents reduces the impracticality of regular application in favour of active or passive emanation leading to continued action with minimal labour input (Achee *et al.*, 2012; Lynch *et al.*, 2016; Norris *et al.*, 2017). Spatial repellent compounds are highly volatile and capable of diffusing through the air and acting on targets at a distance from the point of release (Achee *et al.*, 2012; Norris *et al.*, 2017) making them ideal for use both indoors and outdoors (Grieco *et al.*, 2007; Lynch *et al.*, 2016; Norris *et al.*, 2017). The vaporization of these compounds, either passively or through the use of a fan or heat-based emanator, creates saturated pockets of air which induce an aversive behaviour from the malaria vectors (Achee *et al.*, 2012; Ogoma *et al.*, 2014b). Compounds such as transfluthrin, delta-undecalactone, dichlorodiphenyltrichloroethane (DDT) among many others have been

shown to have a spatial repellent effect on malaria mosquitoes (Barasa *et al.*, 2002; Pates *et al.*, 2002; Grieco *et al.*, 2007; Menger *et al.*, 2014; Ogoma *et al.*, 2014b; Sangoro *et al.*, 2014b; Deletre *et al.*, 2019). Transfluthrin has been mostly used indoors and has either been applied on surfaces or fabric, evaporated from heated devices or burning coils, or dispensed by devices with fans (Ogoma *et al.*, 2014b; Menger *et al.*, 2016; Ogoma *et al.*, 2017). Interestingly, some insecticides such as transfluthrin and chlorpyrifos methyl exhibit spatial repellent tendencies at sub-lethal concentrations (Achee *et al.*, 2009; Ogoma *et al.*, 2014b; Ogoma *et al.*, 2017) while other insecticides such as DEET have a stronger repellent effect than a toxic or killing effect on mosquitoes (Dennis *et al.*, 2019).

Many spatial repellents that have been developed for market use have been in the form of electrical vaporisers or mosquito coils (Ramesh *et al.*, 2001; Pates *et al.*, 2002; Nazimek *et al.*, 2011; Maia *et al.*, 2018). As many malaria endemic areas in rural Africa lack electricity, mosquito coils have been more commonly used, though their contribution to environmental pollution and respiratory illness have led to public health calls for their discontinued use (Hogarh *et al.*, 2016; Madhubabu *et al.*, 2017; Hogarh *et al.*, 2018; Tangena *et al.*, 2018).

Alternative methods of availing spatial repellents have been considered and suggestions made for the use of treated outdoor decorations, containers and eave fabrics particularly when produced using locally available fabrics such as hessian (burlap) fabric (Ogoma et al., 2012b; Masalu et al., 2017; Ogoma et al., 2017; Masalu et al., 2018). Many rural houses in Africa have open eaves as part of the house design to increase air circulation. These open eaves greatly contribute to increased risk of house-entry by malaria vectors and open eaves rank highly as a risk factor for malaria transmission (Ghebreyesus et al., 2000; Njie et al., 2009; Jatta et al., 2018b; Kaindoa et al., 2018; Mburu et al., 2018). Screening of exposed or open eaves, even with untreated materials, has been shown to play a major role in the reduction in indoor mosquitoes in various studies (Lindsay et al.; Ogoma et al., 2010; Menger et al., 2016; Mburu et al., 2018). Addition of insecticides on the eave fabric further reduces house entry behaviour of mosquitoes by up to 99% (Menger et al., 2016). Using repellent-treated eave fabric on such houses can greatly reduce the number of host-seeking mosquitoes both indoors and outdoors through effective passive release yet without requiring separate electrically powered devices (Ogoma et al., 2012b; Govella et al., 2015; Masalu et al., 2017; Ogoma et al., 2017; Mmbando et al., 2018; Mwanga et al., 2019). It would be, however, interesting to explore the possibility of using a treated eave strip or ribbon as opposed to an eave screen for consideration to use as the push component in the proposed push-pull strategy. An eave strip would utilize less fabric than full eave screening with fabric and would therefore reduce the initial costs to be incurred in the production of a push component for field application. Application in the field would be simpler than a full eave screen. Additionally, the variability of eave width in rural homes in western

Kenya would make it impossible to obtain a fabric that would cover all eaves as had been done in previous studies (Ogoma *et al.*, 2010; Menger *et al.*, 2016).

Application of spatial repellents has shown success in reducing host-vector interaction among protected hosts, although it is possible to unintentionally redirect mosquitoes to homesteads and persons that are not protected by a repellent (Maia *et al.*, 2016). To reduce this diversionary effect of spatial repellents on malaria vectors from protected populations to unprotected populations, a supplementary system could be added that would reduce the biting fraction of the mosquitoes by removal trapping thus reducing disease transmission (Cook *et al.*, 2007; Wagman *et al.*, 2015b).

The use of outdoor bait stations (traps or landing boxes) has been proposed in combination with spatial repellents in a push-pull system (Takken, 2010; Menger *et al.*, 2015; Wagman *et al.*, 2015b; Menger *et al.*, 2016). This concept proposes that the application of both push (spatial repellent) and pull (odourbaited stations) systems offers greater protection against mosquito bites than repellents alone or traps alone (Dekker *et al.*, 2001; Njiru *et al.*, 2006; Okumu *et al.*, 2010c; Revay *et al.*, 2013; Obermayr *et al.*, 2015) especially in the peridomestic area (*i.e.* the area in the direct circumference of a house; Figure 1) (Menger *et al.*, 2015). Once mosquitoes come into contact with spatial repellents, they are diverted away from the target area (Menger *et al.*, 2015). As they fly away from the repellent plume, mosquitoes perceive synthetic lures set up in either odour-baited traps or landing boxes and are attracted to them (Takken *et al.*, 2012b; Menger *et al.*, 2006; Qiu *et al.*, 2007; Okumu *et al.*, 2010b; Okumu *et al.*, 2010c; Mukabana *et al.*, 2012b; Menger *et al.*, 2015).



Figure 1. Proposed application of protection in the peridomestic area (encircled in blue) by the combination of push (red arrows), provided by the spatial repellent, and pull (green arrows), provided by odour-baited trap.

A key attractant used by mosquitoes to identify vertebrate hosts is carbon-dioxide (Takken *et al.*, 1989; Costantini *et al.*, 1996; Gillies, 2009). Traps utilizing attractive lures usually incorporate CO_2 for increased effectiveness (Gillies *et al.*, 1987; Takken *et al.*, 1989; Costantini *et al.*, 1996; Mboera *et al.*, 1997; Takken *et al.*, 1997; Mboera *et al.*, 2000; Gillies, 2009; Smallegange *et al.*, 2010; Mweresa *et al.*, 2014). When used alone or with odour lures, CO_2 activates host-seeking mosquitoes and induces them to fly upwind toward the trap. The odour plumes alone are not, however, sufficient to induce a mosquito to enter a trap, necessitating powered fans which suck mosquitoes into the trap (Cardé *et al.*, 2010a; Hiscox *et al.*, 2014). Synthetic lures usually play a major role in close-range attraction and landing (Takken *et al.*, 1997; Gibson *et al.*, 1999; Takken *et al.*, 1999; Dekker *et al.*, 2001; Foster *et al.*, 2004).

Provision of carbon dioxide can be costly as it requires special packaging of dry ice or metal tanks containing carbon dioxide under high pressure. To overcome this obstacle, an alternative source of carbon dioxide can be fermentation of sugar or molasses with yeast (Saitoh *et al.*, 2004; Oli *et al.*, 2005; Patrascu *et al.*, 2009; Smallegange *et al.*, 2010; Mweresa *et al.*, 2014). While designing the pull component of a push-pull product, it is crucial to explore possibilities of obtaining a carbon dioxide replacement for ease of field application due to the impracticality of generating sufficient carbon dioxide in the field (Oli *et al.*, 2005; Mburu *et al.*, 2017). One chemical that has been proposed as a possible alternative to carbon dioxide in attracting host-seeking malaria mosquitoes is 2-butanone (Mburu *et al.*, 2017), which was reported to activate the cpA neuron on the maxillary palp; in mosquitoes this olfactory receptor is the principal ligand of carbon dioxide mimic. In a field study the compound was shown to be an effective replacement of carbon dioxide in attracting the malaria vectors using odour-baited traps supplemented with 2-butanone led to a reduction of over 69% *Anopheles funestus* population and a reduction of over 30% malaria prevalence (Hiscox *et al.*, 2016; Homan *et al.*, 2016).

As an alternative to the use of odour baits in the pull system, traps that utilize other host-seeking cues or attractive cues other than olfactory ones could be explored which can be used to catch mosquitoes that have been diverted by spatial repellents (Joshi *et al.*, 1975; Mathenge *et al.*, 2005; Murchie *et al.*, 2016). The use of non-odour traps may be necessary if the repellent insecticide affects odorant receptors leading to a reduction in perception of host odours (Ditzen *et al.*, 2008; Murphy *et al.*, 2013; Tsitoura *et al.*, 2015; Ponlawat *et al.*, 2017; Tzotzos *et al.*, 2018). Tapping into the mosquito visual cues for example, may be explored as an option to ensure that in case mosquitoes odorant receptors are compromised following

exposure to repellents, the mosquitoes would still be able to respond to light cues (Murchie *et al.*, 2016; LeClair *et al.*, 2017; Ponlawat *et al.*, 2017).

Push-pull strategies have been used to control disease-transmitting insects of both agricultural and medical importance by using repellents identified among host odours combined with killing agents (Cook *et al.*, 2007; Takken, 2010; Wachira B.M., 2016). In principle the strategy may have the potential to control outdoor transmission of malaria in many communities, since host-seeking mosquitoes that are repelled from their intended hosts will be eventually lured towards specific lethal sites, therefore reducing chances of transmitting diseases through reduced host-vector contact and population suppression (Cook *et al.*, 2007). Laboratory and semi-field studies in Germany that targeted *Aedes* mosquitoes showed a reduction of up to 50% of bites on human volunteers when a spatial repellent (catnip oil) was used in combination with BG-sentinel traps (Obermayr *et al.*, 2015).

Another study targeting malaria vectors in central America showed a reduction in house entry behaviour of *An. vestitipennis* when transfluthrin was applied indoors, with a simultaneous increase in mosquito catches in the outdoor traps (Wagman et al., 2015b). Other studies done in Kenya showed reduction in house-entry behaviour of malaria vectors of between 61% and 99% and increased mosquito capture in the outdoor baited traps when several spatial repellents were applied indoors when house entry behaviour was monitored overnight (Menger et al., 2015; Menger et al., 2016).

More knowledge is required for the development of a 'push-pull' product for large scale roll out specifically targeting outdoor-biting mosquitoes. The product should target the mosquito's response to human odours through the release of both spatial repellents and synthetic lures mimicking human scent released from traps (Dia *et al.*, 2005; Qiu *et al.*, 2010). The ideal mosquito response to the push-pull strategy would be initial diversion from the protected human but a continued host-seeking behaviour to allow for the detection of the synthetic lures once they are in proximity to the odour plumes emitted by traps or other lure-and-kill devices. Quantification of the chemicals present in the air surrounding the system will give an indication of the amount of active ingredients that mosquitoes are exposed to, while also providing data for health and environmental safety assessment of the system (Martin *et al.*, 2013; Hogarh *et al.*, 2018; Kwan *et al.*, 2018). Understanding the basis of the decision made by mosquitoes on how to proceed following exposure to this cocktail of cues is important in determining the effectiveness of the proposed push-pull tool in controlling outdoor mosquitoes (Cardé *et al.*, 2010b; Suh *et al.*, 2014).

My study seeks to characterize the mosquito response to human hosts in the presence of spatial repellents and synthetic odours while determining effective concentrations required to reduce human bites by hostseeking anopheline mosquitoes. Additionally, the dissemination of repellents using fabrics such as eave

wraps will be explored in comparison to eave screening. The potential of 2-butanone as a carbon dioxide replacement will be explored in the pull system, and light traps explored as possible alternatives in reducing the population of diverted mosquitoes. The findings aim to further develop the application of the push-pull strategy for anopheline mosquito vector control.

This thesis:

This thesis describes the host-seeking behaviour of *Anopheles* mosquitoes in the presence of both spatial repellents and synthetic lures and seeks to define conditions that are necessary to achieve protection to susceptible humans outdoors. Data generated on the selection of the constituents of the push-pull system through systematic examination of various candidate components offers information on the process behind this control tool to define parameters within which protection is conferred to humans outdoors and indoors. Mosquito behavioural responses to human odours compared to light cues are studied in the presence of transfluthrin as a guide to optimise tools needed to work in synergy to provide maximum host protection.

The work described in this thesis is based on the following specific objectives:

- 1. Develop a push-pull system that is protective against anopheline mosquitoes to humans outdoors under semi-field conditions
 - a. Systematically test spatial repellents and odour-baited traps and determine the combination that offers the best protection to humans outdoors.
 - b. Quantify the chemical components of the push-pull system in air samples.
- 2. Evaluate the efficacy of 2-butanone as a replacement for CO₂ in odour-baited traps designed to attract host-seeking *Anopheles arabiensis* mosquitoes.
- 3. Determine response of host-seeking *Anopheles* mosquitoes to synthetic lures in comparison to UV light traps both in the absence and the presence of transfluthrin.
- 4. Investigate the impact of transfluthrin-treated fabric used as an 'eave wrap' on outdoor and indoor human biting rates compared to an 'eave screen'
- 5. Determine effect of the push-pull system on indoor and outdoor-biting natural mosquitoes in comparison to transfluthrin only (push) and an odour-baited trap only (pull).

Chapter 2 provides a description of the development of the push-pull system and provides data that show when outdoor-protection was achieved. Details are provided on the various repellents that were tested for consideration as the push component and how they compared to each other, while the pull component was tested on its ability to provide protection to humans from outdoor-biting mosquitoes. The best performing

push and pull systems were combined and the sum of their protection on humans against outdoor-biting mosquitoes quantified. The concentration of the constituent chemicals in the air was measured at different heights and distances from point of release to describe spatial variation important to explain mosquito behaviour.

Chapter 3 provides empirical evidence of changes in preference of host-seeking mosquitoes to either odour-baited traps or CDC UV light traps when transfluthrin is present in relatively close proximity compared to when it is absent, and attempts to inform future developments of the push-pull set up where both the push and the pull components are set up in close proximity.

Chapter 4 compares the protection of transfluthrin-treated eave wraps on both outdoor and indoor biting mosquitoes to that conferred by eave screens. Additional comparisons are made to their untreated counterparts as a measure of any inferiority in protection between the two and to define the scope of protection provided by eave wraps compared to eave screens.

Chapter 5 describes afield assessment of the push-pull system by providing data on the response of natural mosquitoes and protection conferred both indoors and outdoors by the push-pull system. The data also allow comparisons for degree of protection between the repellent eave wrap applied alone compared to the odour-baited trap applied alone to assess if synergistic protection in the push-pull system occurred.

Chapter 6 provides a summary and discussion of findings that aims to guide the future of push-pull mosquito control in Kenya.

Evaluating putative repellent 'push' and attractive 'pull' components for manipulating the odour-orientation of host-seeking malaria vectors in the peridomestic space

Margaret M. Njoroge, <u>Ulrike Fillinger</u>, Adam Saddler, Sarah Moore, Willem Takken, Joop J.A. van Loon, Alexandra Hiscox

Published in Parasites & Vectors (2021)

DOI: 10.1186/s13071-020-04556-7

Abstract

Background Novel malaria vector control approaches aim to combine tools to work in synergy for maximum protection. This study aimed to evaluate novel and re-evaluate existing, putative repellent 'push' and attractive 'pull' components for manipulating the odour-orientation of malaria vectors in the peri-domestic space.

Methods *Anopheles arabiensis* outdoor human landing catches and trap comparisons were implemented in large semi-field systems to (1) test the efficacy of Citriodiol® or transfluthrin-treated fabric strips positioned in house eave gaps as push components for preventing bites; (2) understand the efficacy of an MB5-baited Suna-trap in attracting vectors in the presence of a human being; (3) assess 2-butanone as a CO₂ replacement for trapping; and (4) determine the protection provided by a full push-pull set up. The air-concentrations of the chemical constituents of the push-pull mosquito control tool were quantified.

Results Microencapsulated Citriodiol® eave strips did not provide any outdoor protection against hostseeking *An. arabiensis*. Transfluthrin-treated strips significantly reduced the odds of a mosquito landing on the human volunteer (OR 0.17; 95% CI 0.12-0.23). This impact was lower (OR 0.59; 95% CI 0.52-0.66) during the push-pull experiment which was associated with low night-time temperatures likely affecting the transfluthrin vaporisation. The MB5-baited Suna trap supplemented with CO₂ attracted only a third of the released mosquitoes in the absence of a human being, however, with a human volunteer in the same system, the trap caught less than 1% of all released mosquitoes. The volunteer consistently attracted over two-thirds of all mosquitoes released. This was the case in the absence ('pull' only) and in the presence of a spatial repellent ('push-pull'), indicating that in its current configuration the tested 'pull' does not provide a valuable addition to a spatial repellent. The chemical 2-butanone was ineffective in replacing CO₂. Transfluthrin was detectable in the air space but with a strong linear reduction in concentrations over 5 metres from release. The MB5 constituent chemicals were only irregularly detected, potentially suggesting insufficient release and concentration in the air for attraction.

Conclusion This step-by-step evaluation of the selected 'push' and 'pull' components led to a better understanding of their ability to affect host-seeking behaviours of the malaria vector *Anopheles arabiensis* in the peri-domestic space and helps to gauge the impact such tools would have when used in the field for monitoring or control.

Keywords: Malaria, Vector control, Outdoor-biting, Spatial repellent, PMD, Citriodiol®, Transfluthrin, GC-FID, semi-field study

Background

In spite of the impressive efforts made in the past two decades, progress in the fight against malaria has stagnated in recent years(Bhatt *et al.*, 2015a; Bhatt *et al.*, 2015b; WHO, 2019). A large proportion of the reduction in malaria has been attributed to vector control, yet research and operational practice have concentrated on the development of chemotherapy and vaccines, with vector control not expanding its arsenal beyond long-lasting insecticidal nets (LLINs) and indoor residual spraying (IRS)(Hemingway *et al.*, 2016). Increased pyrethroid resistance in malaria vectors(Churcher *et al.*, 2016; Ranson *et al.*, 2016), shifts in mosquito biting behaviour from predominately endophagic to more exophagic populations (Meyers *et al.*, 2016; Limwagu *et al.*, 2019) and earlier biting (Mathania *et al.*, 2016) demand the re-evaluation of contemporary practices and the development of additional tools addressing current limitations. The World Health Organization (WHO) endorsed the universal use and application of LLINs and IRS as tools in the fight against malaria(WHO, 2014). Both of these tools primarily target indoorbiting mosquitoes which contribute to almost 80% of all malaria transmission (Saavedra *et al.*, 2019). Whilst the remaining outdoor transmission increases in importance once the indoor tools are effectively applied(Killeen *et al.*, 2012; Killeen, 2014), no outdoor tools have been approved by WHO for supplementary mass application(WHO, 2019).

The use of spatial repellents has been proposed to provide protection against bites at a distance from the point of application which could not only provide potential protection to multiple persons but may also lead to higher compliance due to reduced need for reapplication which is a barrier to effective use of tropical repellents (Grieco et al., 2005; Achee et al., 2012; Lynch et al., 2016; Norris et al., 2017). The ability to produce vector-free spaces would make spatial repellents ideal for application in the peridomestic space, defined as in-and around the outside of the house(Achee et al., 2012). Several insecticides already used in public health have, to varying degrees, spatial repellent effects on various mosquito species (Bibbs et al., 2018). These insecticides volatilize more readily than other adulticides and repel, even in instances when the vectors are intrinsically resistant to pyrethroids (Bowman et al., 2018; Deletre et al., 2019). One pyrethroid that exhibits spatial repellent properties against mosquitoes at sublethal concentrations is transfluthrin (Ogoma et al., 2012b; Bibbs et al., 2018; Martin et al., 2020). However, in the light of growing pyrethroid resistance it would also be desirable to search for novel active compounds. For example, Citriodiol® sourced from Eucalyptus citriodora oil, which includes a minimum 64% para menthane-3, 8-diol (PMD) as the active ingredient, is used in topical skin repellents (Barasa et al., 2002; Carroll et al., 2006; Carroll et al., 2019) and has been suggested to have spatial repellent properties(Menger et al., 2014).

There is a possibility that, when used on their own, spatial repellents might lead to increased biting on unprotected persons through diversion of host-seeking vectors from treated to untreated spaces(Maia *et al.*, 2016). To prevent diverted vectors from finding alternative hosts, supplementary tools such as odourbaited traps might be combined with spatial repellents. Odour-baited mass trapping, as a single tool, has shown to reduce *An. funestus* densities indoors in a recent field trial (Homan *et al.*, 2016). Spatial repellents and odour-baited traps target opposing odour-mediated orientations of the mosquito and therefore may work synergistically in a 'push-pull' system (Takken, 2010; Menger *et al.*, 2014; Menger *et al.*, 2015; Menger *et al.*, 2016).

The term 'push-pull' was first conceived as a strategy for insect pest management in Australia in 1987(Pyke et al., 1987) and the concept is now frequently applied in the control of agricultural pests(Cook et al., 2007; Yan et al., 2015). The intervention not only offers repulsion from the intended host, but rather redirects them to an alternative that does not lead to disease(Cook et al., 2007; Takken, 2010; Njihia et al., 2014; Wagman et al., 2015b). An adaptation of this tool for vector control was developed to curb transmission of trypanosomiasis. Cattle provided with a repellent worn on the neck as a push, were supplemented with insecticide-treated targets which acted as attractive pull components that killed the flies that landed on them (Saini et al., 2017). The reduction in tsetse fly populations was more strongly associated with a combined push-pull set up than with the repellent and attractant when used separately or not at all(Saini et al., 2017). To develop such a 'push-pull' strategy for malaria vector control, it is necessary to determine the efficacy of the potential components individually and in combination to understand their contribution to protecting human hosts from bites. The push-pull strategy for malaria vector control targets the odour-mediated orientation of female mosquitoes when searching for a human host and aims to manipulate this behaviour. This requires that effective quantities of the repellent and odour attractants are perceived by the targeted mosquito species within the space that should be protected(Martin et al., 2013). Quantification of the airborne concentrations of the chemical constituents of the push-pull control tool might help interpret behavioural responses recorded in bioassays and gauge the influence of weather conditions (Ramesh et al., 2001; Martin et al., 2013; Kwan et al., 2018). Such information might inform the spatial arrangement of the push-pull system and assist in identifying needs for improvements of release rates of individual components. Importantly, quantification of chemicals in the air allows for monitoring of safe levels, especially amounts inhaled by humans or levels available to susceptible non-target hosts (Ramesh et al., 2001; Nazimek et al., 2011).

This study aimed to evaluate novel and re-evaluate existing, putative repellent 'push' and attractive 'pull' components for manipulating the odour-orientation of malaria vectors in the peri-domestic space with the aim to develop a 'push-pull' system that reduces bites and kills vectors. Five objectives were pursued: (1)

To test the efficacy of fabric strips treated with either microencapsulated Citriodiol® or with an emulsified concentrate of transfluthrin positioned in open eave gaps on houses as a push component for preventing *Anopheles arabiensis* bites outdoors; (2) To understand the efficacy of an MB5-blend baited Suna-trap in attracting (pulling) *An. arabiensis* to the trap in the presence of a human being; (3) To assess the possibility of replacing CO_2 produced from yeast-sugar fermentation with the putative CO_2 replacement, 2-butanone, in the Suna trap; (4) To determine the degree of protection for a human host against mosquito bites by combining push and pull components; and (5) To quantify the air concentrations of the chemical constituents of the push-pull mosquito control tool.

Methods

Study site

All experiments were carried out in semi-field systems made up of four netting-screened green-houses located at the International Centre of Insect Physiology and Ecology's Thomas Odhiambo Campus (*icipe*-TOC) at Mbita, in Homabay County, western Kenya (0°26'06.19"S, 34°12'53.13"E; altitude 1,137 m). The majority of experiments (Table 1) were carried out in two large semi-field systems (Amiran Ltd, Nairobi, Kenya) measuring 27 m in length, 11 m in width and 4.3 m at the highest midpoint (Figure 1).

The two large systems were located in parallel, 10 m apart from each other. The roof covers were made from translucent water-proof SolarigTM material (Amiran Kenya Ltd.) and the sides were made of a 17-mesh netting material (17 apertures per every linear inch of mesh). One wooden make-shift hut made from plywood walls attached to angle irons, with grass thatch applied on an open gable roof, was included in each system at opposite ends approximately 5 m away from the shorter walls.

The huts were 6.5 m long and 3.5 m wide with a maximum height of 2.5 m (Figure 1). Between the roof and the walls was a 0.1 m eave gap; a size that was representative of the open eave gaps typical in traditional western Kenyan houses and in other rural African areas (Njie *et al.*, 2009; Wanzirah *et al.*, 2015; Jatta *et al.*, 2018a). The doors and windows of the experimental huts were fully mesh-screened. Mosquitoes could only enter and exit the huts through the eave gaps during experiments.





Figure 1. Pictorial presentation of the experimental set ups in the semi-field systems. A. View into the large tunnel-shaped semi-field system; 11 m wide and 27 m long. **B.** Volunteer implementing human landing collections between the experimental hut and the Suna trap in the larger system. **C.** Schematic description of experiments including HLC outside the hut 2.5 m away from the hut (eave treatments) and the Suna trap. Colour-coded mosquitoes were released from all four corners of the system. **D.** Schematic description of experiments in the small semi-field system, 11 m long and 7 m wide, where different trap configurations were tested with two traps included in the system.

Experiment . With a besine of any treatment? • White is the human bring rate of Anopheles architensis released in the semi-field systems between 19.00 and 23.00 h in the absence of any treatment? • Are the two semi-field systems comparable in the results they generate? • Are the two semi-field systems comparable in the results they generate? • Are the two semi-field systems comparable in the results they generate? • Are there any differences in catching efficiency/attractiveness of HLC volunteers? 11 no treatment no treatment yes Experiment 2 • Can Gitriotolol® untreated cotto fabric yes independent 213 1135 yrf transfluthrin fabric untreated cotto fabric yes independent 213 1135 yrf transfluthrin fabric untreated cotto fabric yes independent 213 1135 yrf transfluthrin fabric untreated lossian fabric yes independent 213 1135 yrf transfluthrin fabric untreated lossian fabric yes independent 213 1135 yrf transfluthrin fabric untreated lossian fabric yes independent 213 1135 yrf transfluthrin fabric untreated lossian fabric yes	Experiment 1 What is the human biting rate of <i>Anopheles arabiensis</i> released in the semi-field systems between 19.00 and 23.00 h in the absence -		TEST Treatment CC	VTROL Treatment	Human landing catch	Test & contre independent o in competition*
1.1 no treatment vis independent 2.2 1.2%min:renearballated Citrioidol® untreated cotton fibric vis independent 2.3 1.2%min:renearballated Citrioidol® untreated cotton fibric vis independent 2.4 1.2%min:renearballated Citrioidol® untreated cotton fibric vis independent 2.3 1.2%min:renearballated Citrioidol® untreated cotton fibric vis independent 2.3 1.2%min:renearballated Citrioidol® untreated cotton fibric vis vis independent 2.4 How effective is the MBS baited Suna trap in attracting inscetary-reared An. arabiensis in a large semi-field system in the absence of a human being? vis	1.1 no treatment 1.1 no treatment 2.1 1 g/m ² microeneapsulated Cirinoiol@ untreated cotton fabric 2.1 1 g/m ² microeneapsulated Cirinoiol@ untreated cotton fabric 2.1 1 g/m ² microeneapsulated Cirinoiol@ untreated cotton fabric 2.1 1 g/m ² microeneapsulated Cirinoiol@ untreated cotton fabric 2.2 1.55g/m ² transfluthrin fabric untreated hessian fabric 2.3 1.25g/m ² transfluthrin fabric untreated hessian fabric 2.4 2.5g/m ² transfluthrin fabric untreated hessian fabric 2.3 1.25g/m ² transfluthrin fabric untreated hessian fabric 2.4 2.5g/m ² transfluthrin fabric untreated hessian fabric 2.4 How does the MBS attridge perform in comparison to nylon strips impregnated by investigator? a large semi-field system in the absence a far archios in sectary-teared An. archiorsis in a large semi-field system in the system in the system sectary-teared An. archiorsis in a large semi-field system in the system sectary-teared An. archiorsis in a large semi-field system in the absence a bit of works semi-field system in the absence a bit of works semi-field system sectary-teared An. archiorsis in a large semi-field system in the absence a bit of works semi-field system sectary-teared An. archiorsis in a large semi-field system sectary-teared An. archiorsis in a large semi-field system sectary-teared An. archiorsis in a large semi-field system sectary sectary sectary sectary sectary sectary-teared An. archiorsis in a large	Expe	riment 1 What is the human biting rate of <i>Anopheles arabiensis</i> released in the se Are the two semi-field systems comparable in the results they generate? Are there any differences in catching efficiency/attractiveness of HLC v	ni-field systems between 19.00 and 23.00 h in the absence dunteers?	of any treatment'	~
Experiment 2 • can Circiodiol® and transfluthrin-tealed strips located at eave gaps reduce An arabiensis biting rates compared to untreated controls? • independent 2.1 1 g/m ² microencepalated Circiodiol® untreated octon fabric yes independent 2.3 1.55g/m ² transfluthrin fabric untreated hessian fabric yes independent 2.4 1.96m ² microencepsulated Circiodiol® untreated hessian fabric yes independent 2.3 2.5g/m ² transfluthrin fabric untreated hessian fabric yes independent 2.4 How effective is the MBS bailed Suma trap in attracting insectary-reared An. arabiensis in a large semi-field system in the absence of a human being? yes independent 2.5g/m ² transfluthrin fabric Does 3-batanone combined with MBS perform equally well in attracting insectary-reared An. arabiensis in a large semi-field system as CO ₂ produced from molasse formation? the work compared compared on the system as CO ₂ produced from molasse formation? 3.1 MBS-eartridge baited Sum trap supplemented with CO ₂ MBS-eartridge no comptring 3.2 MBS-eartridge baited Sum trap supplemented with CO ₂ NBS-eartridge no comptring 3.3 MBS-eartridge Unbaited Sum	 Experiment 2 Can Circiodiol® and/or transfluthrin-treated strips located at eave gaps reduce <i>An. arabiensis</i> biting rates compared to untreated cortin fabric 1g/m² microencapaulated Circiodiol® untreated exten fabric 1g/m² microencapaulated Circiodiol® untreated exten fabric 1g/m² microencapaulated Circiodiol® untreated hessian fabric 2.5g/m² transfluthrin fabric 2.4 2.5g/m² transfluthrin fabric 2.5g/m² transfluthrin fabric 2.6g/m² transfluthrin fabric 3.6 MB5-cartridge bailed Suna trap suplemented with CO₂ MB5-mylon strip bailed Suna trap with CO₂ only suction fan Suna trap with 2-butanone 3.1 MB5-cartridge bailed Suna trap suplemented with CO₂ MB5-mylon strip bailed Suna trap (no MB5, no CO₂, only suction fan Suna trap with 2-butanone 3.6 MB5-cartridge bailed Suna trap with CO₃ 3.6 MB5-cartridge bailed Suna trap suplemented with CO₃ 3.6 MB5-cartridge bailed Suna trap with 2-butanone 3.6 MB5-cartridge bailed Suna trap with 2-butanone 3.6 MB5-cartridge bailed Suna trap with 2-butanone 3.6 MB5-cartridge bailed Suna trap fuo MB5, no CO₂, only suction fan Suna trap with CO₃ 3.6 MB5-cartridge bailed Suna trap with CO₃ 3.6 MB5	1.1	no treatment no	eatment	yes	independent
 Can Chrodong and ordination activity located at eavy gas reduce <i>An</i>. <i>arabiensis</i> biting rates compared to untreated conto fabric yes independent 1 g/m² microencepsulated Citriodiol® untreated cotton fabric yes independent 1.25g/m² transfluthrin fabric untreated cotton fabric yes independent 1.25g/m² transfluthrin fabric interacted nexisinal fabric yes independent 1.25g/m² transfluthrin fabric interacting insecary-reared <i>An</i>. <i>arabiensis</i> in a large semi-field system in the absence and presence of a human being? How effective is the MBS bailed Sum trap in attracting insecary-reared <i>An</i>. <i>arabiensis</i> in a large semi-field system in the absence and presence of a human being? How does the MBS partied Number of the Number of the	 Can Chronogous and or transmomtument of the control fabric untreated conton fabric untreated control fabric untreated context and fabric untreated conton fabric untreated conton fabric untreated conton fabric untreated context and fabric untreated f	Expe	iment 2		c	
 11 g/m² microencepsulated Citriodiol® untreated cotion fabric yes independent 125g/m² microencepsulated Citriodiol® untreated hessian fabric yes independent variant lubric tansfluthrin fabric untreated hessian fabric yes independent yes independent 12.5g/m² transfluthrin fabric yes independent untreated hessian fabric yes independent the weffective is the MBS baited Sum trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system as CO₂ produced from molasse fermonation? How effective is the MBS baited Sum trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system as CO₂ produced from molasse fermentation? How does the MBS baited Sum trap suplemented with CO₂ MBS-arrindge setied with the synthetic MBS Iure in addition to CO₂? MBS-arrindge baited Sum a trap supplemented with CO₂ MBS-arrindge sum trap suplemented with CO₂ MBS-arrindge setied with CO₂ MBS-arrindge setied sum at rap with CO₃ mo competing Sum a trap with 2-butanone baited sum at rap with CO₃ MBS-carrindge baited Sum at rap with 2-but anone MBS-carrindge baited Sum at rap with 2-but anone MBS-carrindge baited Sum at rap with CO₃ MBS-carrindge baited Sum at rap with 2-but anone MBS-carrindge baited Sum at rap with CO₃ MBS-carrindge baited Sum at rap with 2-but anone What is the impact of a c	 11g/m² microencapsulated Citriodiol® untreated cotion fabric 2.5g/m² transfluthrin transfluthrin transfluthrin transfluthrin transfluthrin transfluthrin 2.5g/m² transfluthrin transfluthrin 2.5g/m² transfluthrin fabric 2.5g/m² transfluthrin transfluthrin 2.5g/m² transfluthrin 3.6 MIB5-cartridge baited Suna trap with CO2 3.7 MIB5-cartridge baited Suna trap with CO2 3.8 MIB5-cartridge baited Suna trap with CO3 3.9 MIB5-cartridge baited Suna trap with CO3 3.1 MIB5-cartridge bai	2.1	Can Utriodiol® and/or transilutinin-treated strips located at cave gaps in 2 m/2 microencansulated Citriodiol®	couce An. arabiensis bitting rates compared to unireated con eated cotton fabric	trols? ves	independent
 1.25g/m² transfluthrin fabric 1.25g/m² transfluthrin fabric 2.4 2.5g/m² transfluthrin fabric Experiment 3 Experiment 3 Experiment 4 How does the MBS bailed Suma trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence and presence of a human being? How does the MBS cartridge perform in comparison to nylon strips impregnated by investigator? Does 2-butanone combined with MB5 perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence and presence of a human being? How does the MBS cartridge perform in comparison to nylon strips impregnated by investigator? Does 2-butanone combined with MB5 perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system as CO₂ produced from molasse formentation? How does the MBS cartridge baited Suma trap supplemented with CO₂ motored from molasse Suma trap supplemented with CO₂ MB5-cartridge baited Suma trap with CO₂ MB5-cartrid	 2.3 1.25g/m² transfluthrin fabric untreated hessian fabric 2.4 2.5g/m² transfluthrin fabric 2.4 2.5g/m² transfluthrin fabric 2.4 2.5g/m² transfluthrin fabric 2.4 2.5g/m² transfluthrin fabric 2.5 How effective is the MB5 baited Suma trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence a How offsective is the MB5 baited Suma trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence a How does the mB5 cartridge perform in comparison to nylon strips impregnated by investigator? a. How does the trapping efficacy compare between Suma traps baited with CO₂ MB5-artridge baited Suna trap suplemented with CO₂ MB5-artridge baited Suna trap suplemented with CO₂ MB5-artridge baited Suna trap with CO₂ MB5-artridge baited Suma trap with CO₂ 3.3 MB5-cartridge baited Suma trap with CO₂ MB5-artridge baited Suma trap with CO₂ 3.4 MB5-cartridge baited Suma trap with CO₂ 3.5 MB5-cartridge baited Suma trap with CO₂ 3.6 MB5-cartridge baited Suma trap with	2.2	11 g/m ² microencapsulated Citriodiol®	eated cotton fabric	yes	independent
 2.4 2.5 g/m² transfluthrin fabric untreated hessian fabric joint independent Experiment 3 Experiment 3 Experiment 3 How effective is full BIS bailed Suma trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence and presence of a human being? How offective is MBS carriedge perform in comparison to nylon strips upregnated by investigator? Does 2-buttanone combined with MBS perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence and presence of a human being? How does the trapping efficacy compare between Suma traps bailed with CO₂ only and Suma trap supplemented with CO₂ mBS-artridge bailed Suma trap supplemented with CO₂ mBS-artridge bailed Suma trap supplemented with CO₂ MBS-artridge bailed Suma trap with 2-butanone MBS-cartridge bailed Suma trap with 2-butanone MBS-cartridge bailed Suma trap with 2-butanone MBS-cartridge bailed Suma trap with CO₂ mubailed Suma trap (no MBS, no CO₂, only suction fan) MBS-cartridge bailed Suma trap with CO₂ MBS-cartridge bailed Suma trap (no MBS, no CO₂, only suction fan) No Suma trap with CD₂ MBS-cartridge bailed Suma trap with CO₂ MBS-cartridge bailed Suma trap with 2-butanone MBS-cartridge bailed Suma trap with CO₂ MBS-cartridge bailed Suma trap with CO₂ MBS-cartridge bailed Suma trap (no MBS, no CO₂, only suction fan) No Suma trap with 2-butanone MBS-cartridge bailed Suma trap (no MBS, no CO₂, only suction fan) MBS-cartridge bailed Suma trap with 2-butanone MBS-cartridge bailed Suma trap wi	 2.4 2.5g/m² transfluthrin fabric 2.4 2.5g/m² transfluthrin fabric Experiment 3 Experiment 3 How effective is the MB5 baited Suma trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence a How does the MB5 cartridge perform in comparison to nylon strips impregnated by investigator? How does the MB5 cartridge perform in comparison to nylon strips impregnated by investigator? Does 2-butanone combined with MB5 perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field systems the trapping efficacy compare between Suma traps baited with CO₂ MB5-nylon strip baited Suma trap supplemented with CO₂ MB5-nylon strip baited Suma trap with CO₂ MB5-cartridge baited Suma trap supplemented with CO₂ Suma trap with CO₂ MB5-cartridge baited Suma trap supplemented with CO₂ Suma trap with CO₂ MB5-cartridge baited Suma trap supplemented with CO₂ Suma trap with CO₂ MB5-cartridge baited Suma tra	2.3	1.25g/m ² transfluthrin fabric	eated hessian fabric	yes	independent
 Experiment 3 How effective is the MBS baited Sum trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence and presence of a human being? How effective is the MBS cartridge perform in comparison to nylon strips impregnated by investigator? Does 2-butonoe combined with MBS perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system as CO₂ produced from molasse fermentation? How does the trapping efficacy compare between Sum a traps baited with CO₂ only and Suma traps baited with the synthetic MBS lure in addition to CO₂? MBS-cartridge baited Sum a trap supplemented with CO₂ on MBS-nylon strips intered Sum a trap supplemented with CO₂ more competing Sum a rap with 2-butanone MBS-cartridge baited Sum a trap supplemented with CO₂ more competing Sum a rap with 2-butanone MBS-cartridge baited Sum a trap supplemented with CO₂ more competing Sum a rap with 2-butanone MBS-cartridge baited Sum a trap supplemented with CO₂ more competing Sum a rap with 2-butanone MBS-cartridge baited Sum a trap supplemented with CO₂ more competing Sum a rap with 2-butanone MBS-cartridge baited Sum a trap supplemented with CO₂ more competing Sum a rap with 2-butanone MBS-cartridge baited Sum a trap supplemented with CO₃ more CO₃ only suction fan) MBS-cartridge baited Sum a trap with 2-butanone MBS-cartridge baited Sum a trap with 2-butanone MBS-cartridge baited Sum a trap supplemented with CO₃ more CO₃ only suction fan) MBS-cartridge baited Sum a trap with 2-butanone MBS-cartridge baited Sum a trap (no MBS, no CO₃, only suction fan) MBS-cartridge baited Sum a trap with 2-butanone WBS-cartridge baited Sum a trap with 2-butanone WBS-cartridge baited Sum a trap more CO₃ MBS-cartridge baited Sum a trap with 2-butanone WBS-cartridge baited Sum	 Experiment 3 How effective is the MB5 baited Suna trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence a How does the MB5 cartridge perform in comparison to nylon strips impregnated by investigator? How does the MB5 cartridge perform in comparison to nylon strips impregnated by investigator? Does 2-butanone combined with MB5 perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence a formation? How does the trapping efficacy compare between Suna traps baited with CO₂ MB5-vartridge baited Suna trap supplemented with CO₂ MB5-cartridge baited Suna trap supplemented with CO₂ MB5-cartridge baited Suna trap with 2-butanone MB5-cartridge baited Suna trap supplemented with CO₂ MB5-cartridge MB5-cartridge baited Suna trap supplemented with CO₂ MB5-cartridge Suna trap with 2-butanone MB5-cartridge baited Suna trap supplemented with CO₂ Suna trap with CO₂ Suma trap with CO₂ MB5-cartridge baited Suna trap supplemented with CO₂ MB5-cartridge baited Suna trap with 2-butanone MB5-cartridge baited Suna trap supplemented with CO₂ MB5-cartridge baited Suna trap with CO₂ MB5-cartridge ba	2.4	2.5g/m ² transfluthrin fabric	eated hessian fabric	yes	independent
 How effective is the MBS batied Sum a tracting inscetary-reared <i>An. arabiensis</i> in a large semi-field system in the absence of a human being? How does the MBS cartridge perform in comparison to nylon strips impregnated by investigator? How does the trapping efficacy compare between Sum a tracting inscetary-reared <i>An. arabiensis</i> in a large semi-field system as CO₂ produced from molasse fermenation? How does the trapping efficacy compare between Sum a traps impregnated by investigator? How does the trapping efficacy compare between Sum a traps bailed with CO₂ mB5-nylon strip baited Sum a traps supplemented with CO₂ no competing baited Sum a trap supplemented with CO₂ mB5-cartridge baited Sum a trap with CO₂ mB5-cartridge baited Sum a trap with CO₂ mB5-cartridge baited Sum a trap with CO₂ motion traps in trap with CO₂ mB5-cartridge baited Sum a trap with CO₂ mbsited Sum a trap (no MB5, no CO₂, only suction fan) no competing Sum a trap with CO₂ mB5-cartridge baited Sum a trap (no MB5, no CO₂, only suction fan) no competing Sum a trap with CO₂ mB5-cartridge baited Sum a trap with CO₂ mbsited Sum a trap (no MB5, no CO₂, only suction fan) yes independent texperiment 4 MB5-cartridge baited Sum a trap supplemented with CO₂ unbaited Sum a trap (no MB5, no CO₂, only suction fan) yes independent texperiment 4 MB5-cartridge baited Sum a trap with CO₂ unbaited Sum a trap (no MB5, no CO₂, only suction fan) yes independent texperiment 4 MB5-cartridge baited Sum a trap with CO₂ unbaited Sum a trap (no MB5, no CO₂, only suction fan) yes independent texperiment 4 MB5-cartridge Baited Sum a trap with CO₂ unbaited Sum a trap (no MB5, no CO₂, only suction fan) yes independent texperiment 4 MB5-cartridge Baited Sum a trap with CO₂	 How effective is the MBS baited Sum trap in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system in the absence a How does the MBS cartridge perform in comparison to nylon strips impregnated by investigator? Does 2-butanone combined with MBS perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field systementation? How does the trapping efficacy compare between Suna traps baited with CO₂ only and Suma traps baited with the synthetic MBS lure MBS-eartridge baited Suma trap supplemented with CO₂ MBS-survige Suma trap with 2-butanone MBS-cartridge baited Suma trap supplemented with CO₂ baited MBS-survige MBS-cartridge baited Suma trap supplemented with CO₂ baited MBS-survige MBS-cartridge baited Suma trap supplemented with CO₂ baited MBS-survice a suma trap with CO₂ MBS-cartridge MBS-cartridge baited Suma trap supplemented with CO₂ baited Suma trap with CO₂ MBS-cartridge MBS-cartridge baited Suma trap supplemented with CO₂ baited unbaited Suma trap (no MBS, no CO₂, only suction fan) suma trap with CO₂ MBS-cartridge baited Suma trap with CO₂ MBS-cartridge baited Suma trap with CO₂ MBS-cartridge baited Suma trap supplemented with CO₂ unbaited Suma trap (no MBS, no CO₂, only suction fan) suma trap with CO₂ MBS-cartridge baited Suma trap supplemented with CO₂ unbaited Suma trap (no MBS, no CO₂, only suction fan) texperiment 4 What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? What is the impact of a complete push	Expe	riment 3			
 How does the MBS cartridge perform in comparison to nylon strips impregnated by investigator? Dees 2-butanone combined with MBS perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system as CO₂ produced from molass fermentation? How does the trapping efficacy compare between Suna traps baited with CO₂ most traps baited with the synthetic MBS lure in addition to CO₂? MBS-cartridge baited Suna trap supplemented with CO₂ MBS-fartidge baited Suna trap supplemented with CO₂ mo competing baited multices and trap with CO₂ most trap supplemented with CO₂ mo competing Suna trap with CO₂ MBS-cartridge baited Suna trap with CO₂ MBS-cartridge baited sum at trap with CO₂ MBS-form trap (no MBS, no CO₂ only suction fan) no competing Suna trap with CO₂ MBS-cartridge baited Suna trap (no MBS, no CO₂ only suction fan) no competing Suna trap with CO₂ MBS-cartridge baited Suna trap (no MBS, no CO₂ only suction fan) no competing Suna trap with CO₂ MBS-cartridge baited Suna trap (no MBS, no CO₂ only suction fan) no competing Suna trap with CO₂ MBS-cartridge baited Suna trap (no MBS, no CO₂ only suction fan) no competing Suna trap with CO₂ MBS-cartridge baited Suna trap (no MBS, no CO₂ only suction fan) yes independent Experiment 4 MBS-cartridge baited Suna trap (no MBS, no CO₂ only suction fan) yes independent CO₂ MBS-cartridge baited Suna trap (no MBS, no CO₂ only suction fan) yes independent tarperiment 4 WBS-cartridge baited Suna trap with CO₂ unbaited Suna trap (no MBS, no CO₂ only suction fan) yes independent CO₂ mat supplemented with CO₂ most or supplemented with mos	 How does the MB5 cartridge perform in comparison to nylon strips impregnated by investigator? Does 2-butanone combined with MB5 perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field sys fermentation? How does the trapping efficacy compare between Sun a traps baited with CO₂ only and Suma traps baited with the synthetic MB5 lure MB5-cartridge baited Sun a trap supplemented with CO₂ MB5-nylon strip baited Suna trap supplemented with CO₂ MB5-cartridge baited Suna trap with CO₂ MB5-cartridge baited Suna trap with CO₂ Suma trap with CO₂ Suma trap with CO₂ MB5-cartridge baited Suna trap with CO₂ MB5-cartridge baited Suna trap with CO₂ Sum a trap with CO₂ Suma trap with CO₂ MB5-cartridge baited Suna trap with CO₂ MB5-cartridge baited Suna trap with CO₂ Sum a trap with CO₂ MB5-cartridge baited Suna trap with CO₂ with unbaited Suna trap (no MB5, no CO₂, only suction fan) Sum atrap with CO₂ MB5-cartridge baited Suna trap with CO₂ unbaited Suna trap (no MB5, no CO₂, only suction fan) texperiment 4 What is the impact of a complete push-pull set-up on the An. arabiensis biting rate? What is the impact of a complete push-pull set-up on the An. arabiensis biting rate? What is the impact of a complete push-pull set-up on the An. arabiensis biting rate? What is the impact of a complete push-pull set-up on the An. arabiensis biting rate? What is the impact of a complete push-pull set-up on the An. arabiensis biting rate? What is the impact of a complete push-pu	•	How effective is the MB5 baited Suna trap in attracting insectary-reared	An. arabiensis in a large semi-field system in the absence a	nd presence of a	human being?
 Does 2-butanone combined with MB5 perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field system as CO₂ produced from molasse fermenation? How does the trapping efficacy compare between Suna traps bailed with CO₂ mol solution to CO₂? MB5-cartridge baited Suna trap supplemented with CO₂ MB5-nylon strip baited Suna trap supplemented with CO₂ no competing baited Suna trap supplemented with CO₂ mol competing baited Suna trap with 2-butanone MB5-cartridge baited Suna trap supplemented with CO₂ MB5-nylon strip baited Suna trap supplemented with CO₂ mol competing baited Suna trap with CO₂ MB5-cartridge baited Suna trap with 2-butanone MB5-cartridge baited Suna trap with CO₂ MB5-cartridge baited Suna tr	 Does 2-butanone combined with MB5 perform equally well in attracting insectary-reared <i>An. arabiensis</i> in a large semi-field systementation? How does the trapping efficacy compare between Suna traps baited with CO₂ MB5-nylon strip baited Suna trap supplemented with CO₂ MB5-sertridge baited Suna trap supplemented with CO₂ MB5-sertridge baited Suna trap with 2-butanone MB5-cartridge baited Suna trap supplemented with CO₂ MB5-sertridge Suna trap with CO₂ MB5-cartridge baited Suna trap with CO₂ MB5-cartridge baited Suna trap supplemented with CO₂ baited MB5-cartridge Suna trap with CO₂ MB5-cartridge baited Suna trap supplemented with CO₂ MB5-cartridge baited Suna trap with CO₂ MB5-ca	•	How does the MB5 cartridge perform in comparison to nylon strips imp	egnated by investigator?	1	
 How does the trapping efficacy compare between Suma traps baited with CO₂ models fure in addition to CO₂? 3.1 MB5-cartridge baited Suma trap supplemented with CO₂ MB5-nylon strip baited Suna trap supplemented with CO₂ no competing baited suma trap with 2-butanone 3.3 MB5-cartridge baited Suma trap with 2-butanone 3.4 MB5-cartridge baited Suma trap with CO₂ 3.5 MB5-cartridge baited Suma trap with CO₂ 3.6 MB5-cartridge baited Suma trap with CO₂ 3.7 MB5-cartridge baited Suma trap with CO₂ 3.8 MB5-cartridge baited Suma trap with CO₂ 3.9 MB5-cartridge baited Suma trap with CO₂ 3.6 MB5-cartridge baited Suma trap with CO₂ 3.6 MB5-cartridge baited Suma trap supplemented with CO₂ 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suma trap with CO₂ 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 WB5-cartridge baited Suma trap with 2-butanone 3.6 WB5-cartridge baited Suma trap with CO₂ 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 WB5-cartridge baited Suma trap with CO₂ 3.6 What is the impact of a complete push-pull set-up on the An. arabitensis biting trat² 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suma trap 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suma trap 	 How does the trapping efficacy compare between Sum a traps bailed with CO₂only and Suma traps bailed with the synthetic MB5 lurc MB5-cartridge bailed Suma trap with 2-butanone 3.2 MB5-cartridge bailed Suma trap supplemented with CO₂ MB5-cartridge bailed Suma trap with 2-butanone 3.3 MB5-cartridge bailed Suma trap with CO₂ 3.3 MB5-cartridge bailed Suma trap with CO₂ 3.4 MB5-cartridge bailed Suma trap with CO₂ 3.5 MB5-cartridge bailed Suma trap with CO₂ 3.6 MB5-cartridge bailed Suma trap with CO₂ 3.7 MB5-cartridge bailed Suma trap with CO₂ 3.8 MB5-cartridge bailed Suma trap with CO₂ 3.9 MB5-cartridge bailed Suma trap with CO₂ 3.4 MB5-cartridge bailed Suma trap with CO₂ 3.5 MB5-cartridge bailed Suma trap supplemented with CO₂ 3.6 MB5-cartridge bailed Suma trap supplemented with CO₂ 3.6 MB5-cartridge bailed Suma trap with 2-butanone 3.6 MB5-cartridge bailed Suma trap supplemented with CO₂ 4.1 Transfiltuthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbailed Suma trap *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment is independently but concurrently. The treatment independent is independently but concurrently. The treatment is indepen	•	Does 2-butanone combined with MB5 perform equally well in attracti fermentation?	g insectary-reared An. arabiensis in a large semi-field sys	tem as CO ₂ proc	luced from molasse
 MB5-cartridge baited Suna trap supplemented with CO₂ MB5-nylon strip baited Suna trap supplemented with CO₂ no competing MB5-cartridge baited Suna trap supplemented with CO₂ MB5-artridge baited MB5-cartridge baited with CO₂ Suna trap with CO₂ MB5-cartridge baited MB5-cartridge baited Suna trap with CO₂ NB5-cartridge baited with CO₂ Suna trap with CO₂ MB5-cartridge baited with CO₂ MB5-cartridge baited with CO₂ NB5-cartridge baited Suna trap with CO₂ NB5-cartridge baited Suna trap with CO₂ NB5-cartridge baited Suna trap with CO₂ MB5-cartridge baited Suna trap with CO₂ NB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) ves independent Si MB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) ves independent Lexperiment 4 What is the impact of a complete push-pull set-up on the An. arabitansis biting tate? What is the impact of a complete push-pull set-up on the An. arabitansis biting tate? 	 MB5-cartridge baited Suna trap supplemented with CO2 MB5-artridge baited Suna trap with 2-butanone MB5-cartridge baited Suna trap with CO2 MB5-cartridge MB5-cartridge baited Suna trap with CO2 MB5-cartridge baited Suna trap with CO2 Suna trap supplemented with CO3 Suna trap with CO2 Suna trap with CO2 Suna trap with CO2 Suna trap with CO2 Suna trap with CO3 Suna trap supplemented with CO3 Suna trap supplemented with CO3 Suna trap supplemented with CO3 Suna trap (no MB5, no CO3, only suction fan) Subscartridge baited Suna trap supplemented with CO3 Suna trap (no MB5, no CO3, only suction fan) Subscartridge baited Suna trap supplemented with CO3 Suna trap (no MB5, no CO3, only suction fan) Subscartridge baited Suna trap with Subscartridge & with untreated cave wrap + mbaited Suna trap (CO3 * Was subsciences and control treatments independently but concurrently. The treatment CO3 * Subsciences Supervised for testing test and control treatments independently but concurrently. The treatment is independently but concurrently. The	•	How does the transing efficacy compare between Suna trans baited with	CO ₂ only and Suna trans haited with the synthetic MB5 lur	e in addition to C	0,2
 3.2 MB5-cartridge 3.3 MB5-cartridge Suma trap with 2-butanone 3.3 MB5-cartridge Suma trap with 2-butanone 3.3 MB5-cartridge baited 3.4 MB5-cartridge baited 3.5 MB5-cartridge baited 3.6 MB5-cartridge baited Suma trap (no MB5) supplemented with CO₂ 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suma trap with CO₂ 3.6 MB5-cartridge baited Suma trap with CO₂ 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suma trap (no MB5, no CO₂, only suction fan) 3.6 What is the impact of a complete push-pull set-up on the An. arabitensis biting trat² 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suma trap 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suma trap 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suma trap 	 3.2 MB5-cartridge 3.3 MB5-cartridge Suma trap with 2-butanone 3.3 MB5-cartridge Suma trap with 2-butanone 3.4 MB5-cartridge baited 3.4 MB5-cartridge baited 3.4 MB5-cartridge baited 3.5 MB5-cartridge baited 3.6 MB5-cartridge baited Suma trap supplemented with CO2 4.1 Experiment 4 • What is the impact of a complete push-pull set-up on the An arabitans biting rate? *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment 	3.1	MB5- cartridge baited Suna trap supplemented with CO ₂ MI	5-nylon strip baited Suna trap supplemented with CO ₂	no	competing
Suma trap with 2-butanoneSuma trap with CO23.3MB5-cartridgebaitedunbaited Suma trap (no MB5, no CO2, only suction fan)nocompeting3.4MB5-cartridge baitedunbaited Suma trap (no MB5) supplemented with CO2nocompeting3.5MB5-cartridge baited Suma trap with CO2unbaited Suma trap (no MB5) supplemented with CO2nocompeting3.5MB5-cartridge baited Suma trap with CO2unbaited Suma trap (no MB5, no CO2, only suction fan)yesindependent3.6MB5-cartridge baited Suma trap with 2-butanoneunbaited Suma trap (no MB5, no CO2, only suction fan)yesindependent3.6MB5-cartridge baited Suma trap with 2-butanoneunbaited Suma trap (no MB5, no CO2, only suction fan)yesindependent4.1transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suma trapyesindependent4.1CO2complete push-pull set-up on the An. arabiting tate?yesindependent	Suma trap with 2-butanone Suma trap with CO2 3.3 MB5-cartridge baited unbaited Suma trap (no MB5, no CO2, only suction fan) Suma trap with 2-butanone baited unbaited Suma trap (no MB5, no CO2, only suction fan) 3.4 MB5-cartridge baited unbaited Suma trap (no MB5) supplemented with CO2 3.4 MB5-cartridge baited unbaited Suma trap (no MB5) supplemented with CO2 3.5 MB5-cartridge baited Suna trap supplemented with CO2 unbaited Suma trap (no MB5, no CO2, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suma trap (no MB5, no CO2, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suma trap (no MB5, no CO2, only suction fan) 4.1 Experiment 4 . . • What is the impact of a complete push-pull set-up on the An arabiensis biting rate? 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment	3.2	MB5-cartridge MI	5-cartridge baited	no	competing
 3.3 MB5-cartridge 3.3 MB5-cartridge 8.una trap with 2-butanone 3.4 MB5-cartridge baited 3.4 MB5-cartridge baited 3.5 MB5-cartridge baited 3.6 MB5-cartridge baited Suna trap with CO2 3.6 MB5-cartridge baited Suna trap with 2-butanone 3.6 MB5-cartridge baited Suna trap (no MB5, no CO3, only suction fan) 3.6 MB5-cartridge baited Suna trap with 2-butanone 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap 	 3.3 MB5-cartridge 3.3 MB5-cartridge 8.1 MB5-cartridge baited 9.4 MB5-cartridge baited 9.4 MB5-cartridge baited 9.4 MB5-cartridge baited 9.4 MB5-cartridge baited 9.6 MB5-cartridge baited Suna trap supplemented with CO2 9.6 MB5-cartridge baited Suna trap supplemented with CO2 9.6 MB5-cartridge baited Suna trap with2-butanone 9.7 MB5-cartridge baited Suna trap with2-butanone 9.6 MB5, cartridge baited Suna trap with2-butanone 9.7 MB5-cartridge baited Suna trap with2-butanone 9.7 MB5-cartridge baited Suna trap with2-butanone 9.8 MB5, no CO2, only suction fan 9.8 MB5, no CO2, only suction fan 9.8 MB5, acve wrap + MB5-cartridge & with untreated cave wrap + unbaited Suna trap 9.8 Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment 		Suna trap with 2-butanone Sur	a trap with CO ₂		
 3.4 MB5-cartridge baited 3.4 MB5-cartridge baited 3.5 MB5-cartridge baited 3.5 MB5-cartridge baited Suna trap supplemented with CO₂ unbaited Suna trap (no MB5) supplemented with CO₂ 3.5 MB5-cartridge baited Suna trap supplemented with CO₂ unbaited Suna trap (no MB5, no CO₂, only suction fan) yes independent 3.6 MB5-cartridge baited Suna trap with2-butanone 3.6 MB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap (no MB5, no	 3.4 MB5-cartridge baited 3.5 MB5-cartridge baited Suna trap (no MB5) supplemented with CO₂ only suction fan) 3.5 MB5-cartridge baited Suna trap supplemented with CO₂ unbaited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment 	3.3	MB5-cartridge baited und	uited Suna trap (no MB5, no CO ₂ , only suction fan)	00	competing
Suma trap with CO2Suma trap with CO23.5MB5-cartridge baited Suma trap supplemented with CO2unbaited Suma trap (no MB5, no CO2, only suction fan)yesindependent3.6MB5-cartridge baited Suma trap with2-butanoneunbaited Suma trap (no MB5, no CO2, only suction fan)yesindependent3.6MB5-cartridge baited Suma trap with2-butanoneunbaited Suma trap (no MB5, no CO2, only suction fan)yesindependent4.1Experiment 44.1transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suma trapCO2	 Suma trap with CO2 3.5 MB5-cartridge baited Suna trap supplemented with CO2 unbaited Suna trap (no MB5, no CO2, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suna trap (no MB5, no CO2, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suna trap (no MB5, no CO2, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suna trap (no MB5, no CO2, only suction fan) 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap CO2 *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment 	3.4	MB5-cartridge baited unl	uited Suna trap (no MB5) supplemented withCO2 only	no	competing
 3.5 MB5-cartridge baited Suna trap supplemented with CO₂ unbaited Suna trap (no MB5, no CO₂, only suction fan) yes independent 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suna trap (no MB5, no CO₂, only suction fan) yes independent 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suna trap (no MB5, no CO₂, only suction fan) yes independent 4.1 Experiment 4 4.1 CO₂ 	 3.5 MB5-cartridge baited Suna trap supplemented with CO₂ unbaited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suna trap (no MB5, no CO₂, only suction fan) 3.6 MB5-cartridge baited Suna trap with2-butanone 4.1 Experiment 4 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap CO₂ *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment 		Suna trap with CO ₂)
 3.6 MB5-cartridge bailed Suna trap with2-butanone unbailed Suna trap (no MB5, no CO2, only suction fan) yes independent Experiment 4 What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbailed Suna trap yes independent CO₂ 	 3.6 MB5-cartridge baited Suna trap with2-butanone unbaited Suna trap (no MB5, no CO₂, only suction fan) Experiment 4 What is the impact of a complete push-pull set-up on the An. arabiensis biting rate? transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap CO₂ *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment 	3.5	MB5-cartridge baited Suna trap supplemented with CO2 unl	uited Suna trap (no MB5, no CO2, only suction fan)	yes	independent
 Experiment 4 What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap CO2 	 Experiment 4 What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? 4.1 transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap CO₂ *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment 	3.6	MB5-cartridge baited Suna trap with2-butanone unl	uited Suna trap (no MB5, no CO ₂ , only suction fan)	yes	independent
• What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? 4.1 transfluthrin 2.5g/m ² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap yes independent CO_2	 What is the impact of a complete push-pull set-up on the <i>An. arabiensis</i> biting rate? transfluthrin 2.5g/m² eave wrap + MB5-cartridge & with untreated eave wrap + unbaited Suna trap CO₂ *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatment 	Expe	riment 4			
CO ₂	CO ₂ *Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatmer	4.1	What is the impact of a complete push-pull set-up on the An arabiensis transfluthrin $2.5g/m^2$ eave wrap + MB5-cartridge & with unt	iting rate? cated eave wrap + unbaited Suna trap	yes	independent
	*Two semi-field systems were used for testing test and control treatments independently but concurrently. The treatmen		CO ₂			

to the two systems. Competing tests were set in the same semi-field system.

Chapter 2

Few experiments (Table 1) were done in smaller-sized semi-field systems measuring 11 m in length and 7 m in width, with 2.5 m at the highest point (Figure 1). The walls of these were screened with fibreglass netting of the same mesh size as the large systems while the roofs were made of translucent polycarbonate (Okal *et al.*, 2015). Ambient temperatures inside the semi-field systems ranged between a minimum of 18°C at night and maximum of 50°C during the day as monitored with data loggers suspended in the middle of the semi-field systems. During the nightly experiments between 19.00 and 23.00 h the average temperature ranged between 21°C and 24°C. The natural floor in all four semi-field systems was covered with a layer of around 20 cm of sand and was watered daily for two hours, prior to the experiments with free-flying mosquitoes, to maintain a relative humidity in the systems of around 70%. A summary of all experiments is found in Table 1. All experiments with human landing volunteers included were implemented in the large semi-field systems.

Mosquitoes

All experiments were implemented with host-seeking females of An. arabiensis Mbita strain, aged between 3 to 5 days post-emergence. Mosquitoes were reared under ambient conditions at *icipe*-TOC following standard operating procedures (Mbare et al., 2013). Nulliparous mosquitoes that had not taken a blood-meal previously were activated to host-seek by placing a human hand near the outside of the mosquito cage and only those that responded to human odours were aspirated and used in experiments. In experiments including ahumanvolunteer, 160 females were released in each semi-field system per experimental night. In experiments including traps only, 200 females were released in each system per night. The mosquitoes were transferred from rearing cages into release cups using mouth aspirators. In the release cups they were starved from water and glucose for a minimum of three and a maximum of five hours prior to release. Anticipating that the orientation of a female mosquito in the system will be affected by the direction of air movement, obstructions like the hut and outside light sources, mosquitoes were released from cups in all four corners of a semi-field system to account for such factors. In experiments including a human, each of the four release cups contained 40An. arabiensis females (total = 160 females). Mosquitoes in each release cup were dusted using a distinct colour of fluorescent dye to distinguish them according to the four corners of release (Verhulst et al., 2013). In choice experiments with traps only, the traps were rotated through the corners of the semi-field system and the mosquitoes released from one release cup in the centre of the screen house.

Repellent-treated fabrics (push component)

A passive release mechanism for spatial repellents was favoured in this project in order to reduce the operational complexity that would come with an electricity-powered active dispenser. Hence, the two test compounds, Citriodiol® and transfluthrin, were both presented on fabrics which can be easily attached to open eave gaps on houses(Menger *et al.*, 2016).

Citriodiol® (Citrefine International Ltd)was microencapsulated by Devan Chemicals, Portugal and applied to fabric by Utexbel, Belgium using the solvent evaporation technique with poly lactic acid as a shell material as previously described (Liu *et al.*, 2005; Menger *et al.*, 2015). The fabric was shipped to Kenya and stored in a cold and dark room prior to use. Two fabric weights with two loads of Citriodiol® were tested. The first was a100% cotton fabric(65g/m²)with1g/m²Citriodiol®and the second had a fabric weight of 550g/m² with a Citriodiol® load of 11g/m²(microcapsules for both were 15 micro-m with 17% wt. of the active ingredient of para menthane-3, 8-diol; PMD).

Transfluthrin(Bayer Global, Leverkusen, Germany) was obtained as emulsified concentrate(EC) of 0.2g/ml and applied on hessian fabric (obtained as burlap material from local markets in Kenya) to achieve two final loads on the fabric; namely $1.25g/m^2$ and $2.5g/m^2$ (Ogoma *et al.*, 2017).The impregnation of the hessian fabric was done in the laboratory at *icipe*-TOC where the respective amount of transfluthrin EC was added into to water that was sufficient for wetting the entire length of fabric without any water remaining. The fabric was soaked well and dried in the shade overnight then wrapped up in aluminium foil and stored in a cold (4°C) and dark room prior to use.

The treated fabrics were cut into strips measuring a length of 21m, corresponding to the perimeter of the eave gaps of the experimental huts and a width of 0.05m, corresponding to half of the width of the eave gap. Correspondingly, untreated fabric strips were prepared in the same dimensions and used for the control experiments. The fabric strips were fixed half an hour prior to mosquito release with flexible aluminium wires in such a way that they were covering only part of the eave gap leaving a similar space above and below(2.5cm each) to allow for movement of air. They represented an incomplete, easy to fix fabric strip along the gaps, not an eave screen. The fabric strips were removed in the morning and stored in the cold room till the next experimental night. Fabric strips were used continuously for a maximum of eight experimental nights. Experiments were done for 16 nights; hence two strips were used per experiment.

Suna trap and odour lure (pull component)

Odour-baited Suna traps were used throughout as pull devices. The trap's development, appearance and operation is described in detail elsewhere(Hiscox *et al.*, 2014). The principle odour bait re-evaluated in experiments was a synthetic chemical blend aiming to mimic human host odours and has previously been published under the name 'Mbita Blend 5' or MB5 (Mukabana *et al.*, 2012a; Mweresa *et al.*, 2014). The MB5 comprised of ammonia (2.5% in water), L-(+)-lactic acid (85% in water), tetradecanoic acid (0.00025g/l in ethanol), 3-methyl-1-butanol (0.000001% in water), and butan-1-amine, prepared at a concentration of 0.001% in paraffin oil(Menger *et al.*, 2014)and was recently associated with significant reductions in *An. funestus* populations during a mass-trapping vector control trial (Homan *et al.*, 2016). Two dispensing substrates of MB5 were compared. As in previously published work, MB5 was presented on nylon strips(Okumu *et al.*, 2010a; Mweresa *et al.*, 2015)where each strip was treated with one chemical of the blend and consequently five strips inserted in the trap. This was compared to a novel, slow-release cartridge developed by Biogents (Biogents Cartridge Lure (Mosquito Attractant) - LI-MR-43, Regensburg, Germany) containing the same five chemicals.

Carbon dioxide has been repeatedly reported as essential in combination with an odour blend for attracting host-seeking malaria vectors (Takken *et al.*, 1989; Healy *et al.*, 1995; Gillies, 2009) and remains one of the most challenging obstacles to area-wide operational use of odour-baited traps. Carbon dioxide gas released from cylinders is not manageable under field conditions; hence a previously developed method of producing CO_2 from fermenting sugar or molasses solution using yeast is now widely used (Smallegange *et al.*, 2010; Harwood *et al.*, 2014; Sukumaran *et al.*, 2016). However, the amount of sugar or molasses needed for every trap night is still prohibitive for operational vector control. The chemical 2-butanone has been proposed as a CO_2 mimic for mosquitoes, but the literature is controversial (Turner *et al.*, 2011; Mburu *et al.*, 2017). Here, CO_2 from fermentation was compared with 2-butanone treated (0.1ml) nylon strips (Mburu *et al.*, 2017) to gain a better understanding of its effectiveness as a supplement of the odour-bait in a Suna trap for reducing *An. arabiensis* bites.

In experiments including a human volunteer, a single odour-baited Suna trap was positioned 5 m from the experimental hut; with the volunteer seated mid-way between the hut and the trap in a straight transect (Figure 1C). The trap was suspended above the ground using a tripod (Hiscox *et al.*, 2014; Mburu *et al.*, 2017)with the main odour-release point, which is the bottom of the funnel, approximately 0.3 m off the ground. In experiments without a human volunteer, two traps were positioned at diagonally opposite corners of the small semi-field system approximately 13m from each other and less than 1m from the

walls of the system (Figure 1D).

Estimation of vector landing rates

Human landing catches (HLC) were carried out as the primary outcome measurement and were conducted on a randomly rotating basis by four adult men (aged between 18 and 50 years) seated 2.5m away from the experimental hut to mimic outdoor biting in a natural setting where people would spend time outside the house during the evening hours. Two volunteers were required per night. In preparation of the experiments, they cleaned their feet and lower legs with odourless soap and took position on a chair as shown in Figure 1. Collections were done for four hours from 19.00h to 23.00h, with volunteers mouthaspirating host-seeking *An. arabiensis* females as soon as they landed on their lower legs(Kenea *et al.*, 2017). The mosquitoes were transferred to collection cups, separated hourly. Protective jackets and shoes were worn to protect heads, arms and feet against bites and torches were used for visualization of mosquitoes when aspirating. Volunteers were randomized to the semi-field system and to the experiment to avoid any bias due to differences in collection efficacy and individual attractiveness to mosquitoes.

Experimental procedures

All experiments and their guiding research questions are detailed in Table 1. Those including human landing catches were conducted as set in two semi-field systems concurrently. All experiments were replicated over 16 nights. A baseline experiment was conducted to understand the mosquito response rate to human volunteers in the two semi-field systems in the absence of any behaviour modulating chemicals. This provided a reference for other experimental sets and helped gauge any differences in attractiveness and catching efficiency of the volunteers or between the two semi-field systems. This experiment also helped to understand the response rates that can be expected from receptive host-seeking mosquitoes in the system. Following this, a threshold was established where, if the response rate in the presence of a human volunteer was below 50% in the control treatment, results were discarded, and the replicate repeated. Spatial repellent treatments were rotated weekly given the need to air between treatments to avoid cross-contamination. Experiments were done for four consecutive nights then all test devices and chemical odours were withdrawn from the semi-field systems. Trap only experiments were conducted through the night from 19.00h to 07.00h the next morning.

Simulation-based power analysis

A simulation-based power analysis (Johnson *et al.*, 2015)was implemented for a 2x2 Latin square experiment with two treatments each tested by four volunteers in two semi-field systems. The aim was to be able to measure a 50% reduction in human landing rate; hence a recapture rate of 60% in the control and 30% in the push-pull experiment was used for the estimation. Assuming160 mosquitoes released in each semi-field system, and assuming10% dispersion due to variability between the semi-field systems, 10% variations between mosquito releases, and 50% variability between the HLC volunteers, 1000 simulations resulted in an estimated power of 0.94 (95% CI 0.87 - 0.98) to detect a 50% reduction in human landing rate for 16 replications.

Air sampling and detection of volatile chemicals released by the push-pull components

Air was sampled in one of the large semi-field systems in the presence of a fully set push-pull system, consisting of 2.5g/m²transfluthrinfabric strips on eave gaps and a Suna trap baited with MB5nylon strips and CO₂ generated through fermentation of molasses. Air was pumped through adsorbent Texan traps (30 mg; GERSTEL-Twister Desorption glass liners from GERSTEL, Muelheim an der Ruhr, Germany, glass wool from Supelco, Bellefonte, PA, USA and 25 mg of Tenax® TA polymer 60–80 mesh from Supelco, Bellefonte, PA, USA).Micro-diaphragm gas pumps were used at the rate of 400ml/min resulting in a total of 120 litres of air passing through each trap over a five hour sampling period (18.00-23.00 h),chosen to align with the time period when human landing catches were implemented under experimental conditions. The air sampling was carried out in the absence of a human to focus on the chemicals released by the push-pull components. All chemicals collected were reported as concentrations averaged over the time-period of trapping and calculated as nanograms per litre of air sampled; subsequently referred to only as 'concentration' in ng/l.

Twelve locations were sampled in a transect between the transfluthrin-treated fabric at the experimental hut and the odour-baited Suna trap placed at a distance of 5 m away from the hut (Figure 2).

Sampling was done every 1 m between the fabric (house wall) and the trap, at four distances. At every distance, sampling was done at three heights: 0.5 m, 1.0 m and 1.5 m (Figure 2).Sampling was replicated over 5 non-consecutive days, with each set up using freshly treated eave fabric and new nylon strips for the odour-blend to ensure consistency in the initial concentrations. At the conclusion of each sampling event, adsorbent filters were stored at -80°C degrees until chemical analysis. For quantification, trapped volatiles were eluted using dichloromethane (DCM; CAS 75-09-2, Merck, Massachusetts, USA) and

Chapter 2



Figure 2. Pictorial presentation of air sampling set up. Air-entrainment pumps were positioned at 1 m (A), 2 m (B), 3 m (C) and 4 m (D) distance from the experimental hut where a transfluthrin-treated fabric was positioned at the eave gap. The baited Suna trap was 5 m away from the hut. Sampling was done at every position at 3 heights: 0.5 m, 1.0 m, and 1.5 m above the ground.

analysed using gas chromatography (GC) with flame-ionization detection (GC-FID; Agilent 7890B, Agilent Technologies, California, USA; (Buse *et al.*, 2019)).

The lowest detection temperature was set at35 °C and the highest was set at 280°C. A Solgewax (SGE, Australia) column, 30m long and 0.25mm in diameter with an internal diameter of 0.2µm, was used.

To obtain calibration lines for quantification of air concentrations for all the push-pull constituent chemicals (except L-lactic acid and ammonia solution) concentration gradients were obtained by preparing dilutions of the chemicals from the stock solutions ranging from 0.5 to 10 ng/µl resulting in the preparation of the following concentrations: $0.5 \mu \text{g/µl}$, $1.0 \mu \text{g/µl}$, $2.0 \mu \text{g/µl}$, $4.0 \mu \text{g/µl}$, $6 \mu \text{g/µl}$, $8 \mu \text{g/µl}$ and $10 \mu \text{g/µl}$. Each concentration was injected separately into the GC-FID, then the area under the curve determined and plotted against the concentration. A linear equation was obtained by plotting all the concentrations against all the areas of each chemical where y represented the area under the curve while x represented the concentration in nanograms per litre of air. All linear equations met the minimum qualification of R² value of 0.98. Subsequent determination of concentration was determined by obtaining the area under the curve directly from the GC-FID and solving for x in the linear equation of each chemical(Oliveira *et al.*, 2010; Guo *et al.*, 2016; Feng *et al.*, 2018).

To determine direction and strength of air movements as well as temperature during collections, a longrange wireless wind logger (Navis WL 11X, NAVIS Elektronika, Kamnik, Slovenia) was set up at 2 m height next to the Suna trap during the air sampling period. Logging of parameters was done in fiveminute intervals.

Data analysis

Analyses of experimental data were done using R version 3.5.1(R Core Team, 2018). Data were descriptively explored and presented by generating box plots, where the boundary of the box closest to zero indicates the 25thpercentile, the black line within the box marks the median, and the boundary of the box farthest from zero indicates the 75thpercentile. Whiskers above and below the box indicate the 10th and 90thpercentiles. All mosquito catches were analysed as proportions(number of mosquitoes attempting to bite either out of the total number released in the system or out of the total number recollected with HLCs and/or traps) using generalized linear mixed models with the experimental night and HLC volunteer(where applicable) included as random factors. All proportions were modelled using binomial probability distributions with logit link functions fitted. Treatment group was included as the fixed factor in the models with the control group as reference. The semi-field system ID was also included as factor and retained in the final model only if significantly associated with the outcome. Where applicable,

interactions were explored. All analyses of volatile chemicals in air samples were done by calculating the means and the standard error for measurements made at every position across the five sampling days. Analysis of variance to determine differences between sampling positions A-D and sampling heights 0.5-1.5m were done for each chemical detected.

Results

Experiment 1: Establishing landing rates of *An. arabiensis* in semi-field systems in the absence of treatments

On average, 67% (95% CI 62-72%) of the released mosquitoes were recaptured in semi-field system A and 62% (95% CI 56-67%) in system B within four hours of human landing collections (19.00-23.00 h) with volunteers seated 2.5 m away from the hut. Adjusting for time of collection and volunteer, the semi-field system (A or B) was not associated with the odds of recapturing a mosquito (OR 0.96, p=0.286; Table 2).

Table 2: Association between outdoor *An. arabiensis* landing and time of collection, semi-field system, and volunteer.

Explanatory variables in	Odds Ratio (OR)	Confidence Interval (CI)		p-value	
multivariable analysis		Lower CI	Higher CI	1	
Collection time*					
19.00-20.00 h	1				
20.00-21.00 h	0.47	0.426	0.511	< 0.001	
21.00-22.00 h	0.28	0.250	0.307	< 0.001	
Semi-field system ID		•			
А	1				
В	0.96	0.868	1.043	0.286	
HLC Volunteer ID					
no. 1	1				
no. 2	1.25	1.098	1.425	< 0.001	
no. 3	1.26	1.099	1.439	< 0.001	
no. 4	1.17	1.018	1.338	0.027	

*no mosquitoes were captured between 22.00-23.00 h, hence this category not included in analysis

Collection time, however, was significantly associated with the outcome (Figure 3). The largest proportion of host-seeking mosquitoes was recaptured in the first hour, whilst none were captured in the 4th collection hour (22.00-23.00 h). The odds of collecting a landing mosquito decreased over time(Table 2, Figure 3).



Figure 3. Hourly *Anopheles arabiensis* landing on human volunteer in two semi-field systems in the **absence of any test.** Based on human landing collections by four volunteers that were randomly rotated between the systems; out of all mosquitoes released (n=160/experimental night).

There was some variability in the collection efficiency between volunteers either due to attractiveness or skills, with one of the volunteers collecting fewer mosquitoes than the others (Table 2). Based on this, in consecutive analyses, the volunteer IDs were included as a random factor in the model. Host-seeking females were recaptured in similar proportions from all four release corners. There was no significant association between the human landing rate and temperature or relative humidity in the semi-field system during experimental nights.

These results, obtained in the absence of any treatment, confirmed that the insectary-reared *An. arabiensis* were highly responsive to a human blood host and that both semi-field systems supported reproducible

results in the presence of an HLC volunteer. Subsequently, this set of experiments served as a reference for all following experiments with various treatments included.

Experiment 2: Investigating potential push components for a push-pull vector control strategy

Microencapsulated Citriodiol® fabric strips on open eave gaps

Neither the eave fabric encapsulated with 1 g/m²Citriodiol®(p=0.488), nor the heavier fabric with $11g/m^2$ Citriodiol ® (p=0.633) were associated with a reduction in the proportion of mosquitoes landing on a volunteer when compared to untreated controls(Figure4A). In all experimental treatments, catches were similar and consistent, ranging between a median of 66-71% of all released mosquitoes recovered through HLC.



Figure 4 Anopheles arabiensis host-seeking while exposed to putative spatial repellents in semi-field systems as estimated with HLC. (A) Two formulations of microencapsulated Citriodiol($1g/m^2$ and $11g/m^2$) were tested in comparison to untreated control fabric. (B) Two impregnation concentrations of transfluthrin were tested on hessian fabric; $1.25g/m^2$ and $2.5g/m^2$, in comparison to untreated control fabric. The proportions are based on the total number of mosquitoes released (n=160/experimental night).

Transfluthrin EC impregnated hessian fabric strips on open eave gaps

Transfluthrin-treated fabric strips, at both treatment loads, were significantly associated with reduced human landing at a distance of 2.5 m away from the hut (Figure4B, Table 3). The odds of a mosquito landing on the volunteer in the presence of the 1.25 g/m^2 transfluthrin fabric was 2.5 times less (OR 0.39; Table 3) than the odds in the presence of the untreated control fabric. This was consistent over time, even when the fabric used in the experiment had been treated over eight days prior. The higher load of 2.5 g/m² resulted in a significantly higher protection with the odds of a mosquito landing 16 times less (OR 0.06; Table 3) than the odds of landing in the control. However, this protection reduced when the age of the treated fabric increased. When the fabric treatment had been done more than a week prior to testing, the odds of receiving a bite increased nearly 3-fold as compared to the fresh treatment (Table 3) but was still superior to the lower load. The median percentage of 63-70% of released mosquitoes landing on the HLC volunteer in the experiments with untreated fabric related well to the reference experiment without any treatments included and confirmed the reproducibility of the test system.

Table 3:	Association	between	outdoor	4n. (arabiensis	landing	and	transfluthrin	-treated	fabrics	(1.25
and 2.5 g	g/m ²) around	l eave gap	DS								

Explanatory variables in multivariable	Odds	Confidence	p-value					
analysis [#]	Ratio (OR)	Lower CI	Higher CI					
Transfluthrin concentration on fabric strip placed on open eave gap of hut								
Untreated	1							
1.25g/m ²	0.39	0.34	0.45	< 0.001				
2.5g/m ²	0.06	0.05	0.08	< 0.001				
Time post-treatment								
< 8 days	1							
>8 days	1.22	0.82	1.81	0.325				
Interaction between transfluthrin concentration * time post-treatment								
1.25g/m ² * <8 days	1							
2.5 g/m ² * <8 days	1							
1.25g/m ² * >8days	1.07	0.81	1.40	0.641				
2.5 g/m ² * >8days	2.78	2.02	3.82	< 0.001				

[#]Human landing collections were done nightly for 4 hours (19.00-23.00 h).
Experiment 3: Investigating pull components for a push-pull vector control strategy

Comparing the attractiveness of two odour dispensing substrates for use in Suna traps

Previously, the MB5 blend was prepared experimentally by manually treating nylon strips with the five chemicals (Okumu *et al.*, 2010a; Menger *et al.*, 2014; Mweresa *et al.*, 2015). For operational large-scale use, this would not be a feasible method, hence a commercial cartridge was developed (Biogents, Germany) which would be easy to use and replace by lay personnel. The competitiveness of the cartridge in attracting mosquitoes to the trap was tested by comparing it to a trap with treated nylon strips in the same small, semi-field system in the absence of a human volunteer.



Figure 5. Exploring the impact of a novel MB5-release cartridge and 2-butanone instead of CO_2 on the *An. arabiensis* trapping efficiency of Suna traps under semi-field conditions. (A) Comparing the attractiveness of Suna traps baited either with MB5 treated nylon strips or MB5 containing cartridges (Biogents, Germany); both were supplemented with CO_2 . The attractiveness of a human is shown as reference. (B) Evaluation of the effectiveness of 2-butanone as a CO_2 replacement for supplementation of MB5-baited Suna traps for attraction of *An. arabiensis*. (C) Box plot comparing the proportion of mosquitoes recaptured with 2-butanone supplemented traps when tested in choice tests. (Note the different scales of the Y-axes in the figures). A total of 200 host-seeking mosquitoes were released per experimental night. The traps were run over night for 12 hours.

In addition to the chemical blend, CO_2 was released in both traps during experiments using the fermentation method(Patrascu *et al.*, 2009; Mweresa *et al.*, 2014).

The CO₂-supplemented Suna traps were equally efficient in recapturing host-seeking *An. arabiensis* females released in semi-field systems, irrespective of the presentation of the chemical blend on nylon strips or enclosed in a slow-release cartridge. The two traps together recaptured 61% (95% CI 55-67%) of the released mosquitoes, with a balanced 1:1 distribution(approximately 30% in a single trap) between the two types of blend dispensers (Figure 5A). Of all trapped females, 49% (95% CI 41-58%) were collected with the cartridge-baited trap. Since there was no advantage of using treated nylon strips, the cartridge was used for all further experiments.

Exploring the effectiveness of replacing CO₂ with 2-butanone as supplement in MB5-baited Suna traps for the attraction of host-seeking *An. arabiensis*

Experiments were implemented in the absence of a human volunteer with traps set up in competition. There was a strong association between the proportion of mosquitoes recaptured and the test (CO₂, 2butanone, or nothing). The odds of catching a mosquito in an MB5-baited trap supplemented with 2butanone was over 30-fold lower than the odds of catching a mosquito if the trap was baited with CO₂ from fermentation (Table4A). The CO₂ supplemented trap recaptured around 36% (95% CI 32-39%) of the released *An. arabiensis* females whilst the 2-butanone supplemented trap and the unbaited trap without any supplement recaptured well below 0.5% of the released mosquitoes (Figure5B).The low catching efficiency of an MB5-baited Suna trap supplemented with 2-butanone was similar in choice tests where the 2-butanone trap was tested in presence of a CO₂ trap and where the 2-butanone trap was tested in presence of a completely unbaited trap (p=0.337; Figure5C).

Notably, the attraction of an odour-baited Suna-trap appears to be largely due to the inclusion of only fermentation-basedCO₂. The chemical blend (MB5) appeared to add very little(CO₂ only vs. reference of MB5 plus CO₂: OR 0.73; 95% CI 0.64-0.84; Table 4)to the attraction of host-seeking *An. arabiensis*.

Testing the effectiveness of MB5-baited Suna traps as pull devices for trapping *An. arabiensis* in close vicinity of a human blood host.

Neither of the MB5-baited traps, either supplemented with CO₂or with 2-butanone, performed well in the presence of a human blood host (Figure6). Whilst in the absence of a human, the MB5-baited Suna trap

supplemented with CO₂ recaptured at least half of what was recaptured by a human volunteer (Figure5A), hardly any host-seeking *An. arabiensis* were trapped in the presence of a human blood host (Figure6).

Table 4: Model outputs for experiments aiming to (A) investigate the performance of 2-butanone and CO₂ in Suna traps; and (B) evaluate the Suna trap in presence of a human blood host.

Explanatory variables	Odds Ratio	Confidence	Interval (CI)	p-value		
	(OR)	Lower CI	Higher CI			
A. Exploring the association between 2-butanone or CO ₂ su	oplement to	Suna traps	and the pro	portion		
of released An. arabiensis trapped						
MB5-cartridge baited Suna trap supplemented with CO2	1					
MB5-cartridge baited Suna trap supplemented with 2-butanone	0.03	0.03	0.04	< 0.001		
Unbaited Suna trap (no MB5, no CO ₂ , only suction fan)	0.01	0.00	0.01	< 0.001		
Suna trap baited with CO₂ only (no MB5)	0.73	0.64	0.84	< 0.001		
B. Exploring the association between the proportion of released An. arabiensis recaptured by human						
landing volunteers and the presence of a pull device						
HLC in presence of MB5-baited Suna trap and CO ₂	1					
HLC in presence of MB5-baited Suna trap and 2-butanone	0.95	0.86	1.06	0.374		
HLC in presence of an unbaited Suna trap unbaited fan	1.06	0.98	1.15	0.170		

A total of 6901 host-seeking mosquitoes were collected by HLC and traps over all experimental nights, out of which only 1.5% (n=103) were trapped in the Suna traps, whilst the remaining 98.5% were attracted to the human landing volunteers. Consequently, the proportion of host-seeking mosquitoes landing on the volunteer was not affected by the presence of a trap in the system (Figure6A). The proportions landing were equally high in systems where there was only an unbaited trap, in systems where the trap was baited with MB5 and 2-butanone as well as in systems where the trap was baited with MB5 and CO₂ from sugar fermentation (Table 4B).

Experiment 4: Investigating the impact of a complete push-pull set-up

The push system consisting of the $2.5g/m^2$ transfluthrin-impregnatedhessian fabric placed around the eave gaps of the experimental hut was combined with the MB5-baited Suna trap supplemented with CO₂. This was the only pull treatment that was effective in attracting mosquitoes in the absence of a human. The combination was tested since it was considered plausible that the spatial repellent might mask the human

odour and hence the trap serving as pull might be more effective than when tested in the presence of a human without the push component.



Figure 6.Exploring the attractiveness of MB5-baited Suna traps as pull devices for trapping *An. arabiensis* in close vicinity of a human blood host.(A)Comparing the attraction of mosquitoes to human landing volunteers in the presence of Suna traps either baited with MB5 & CO_2 , MB5 & 2-butanone or unbaited with only the fan running. (B)Comparing the attraction of mosquitoes to the different traps in the presence of the human blood host. A total of 160 host-seeking mosquitoes were released in the semi-field system per experimental night. HLC was done for four hours (19.00-23.00 h) and the traps ran over night for 12 hours.

Comparing the functional push-pull set up with the set up containing all components but without chemicals (untreated), the odds of receiving a mosquito landing to bite was 3.4 times lower in the

presence of the push-pull system (OR 0.29 (95% CI 0.25-0.34), p<0.001; Figure 7). However, this result needs to be interpreted with caution since the reference set up here (all components without chemicals) already presents an intervention which was associated with increased outdoor biting as compared to the control where all components were absent. The presence of untreated and unbaited components was associated with a higher odds of a mosquito landing on a volunteer (OR 2.22 (95% CI 1.96-2.52), p < 0.001) compared to the setting where all components were completely absent. It is unclear if this might be due to the fabric strips preventing mosquitoes from entering the hut, hence keeping them closer to the human, or if the observation might be due to other unaccounted conditions given that the two control experiments were implemented at different time points (Figure 7).A more conservative approach in estimating the impact is therefore to compare the odds of a mosquito trying to bite a human volunteer between the push-pull system and the control without any intervention. In this case, the odds was 1.7 times lower in the presence of the repellent-treated and odour-baited push-pull system than in the control (OR 0.59 (95% CI 0.52-0.66), p<0.001). Notably, the estimated reduction in the proportion of An. arabiensis landing on the human host in this combined push-pull experiment was much lower than it was in experiment 2 when the push was tested alone (2.5 g/m²transfluthrin fabric compared to untreated fabric OR 0.17 after 1 week post-treatment).



Figure 7.Exploring the impact of a complete push-pull set-up on *An. arabiensis* **landing on a human volunteer or being attracted to a trap.(A)** Proportion of released mosquitoes landing on the human volunteer outside the hut in the presence of a Suna trap. **(B)** Proportion of mosquitoes collected while landing on the human volunteer out of all mosquitoes recollected total of HLC and trap catches). **(C)**Proportion of mosquitoes trapped in unbaited or baited Suna traps (in presence of HLC) out of all mosquitoes released. A total of 160 host-seeking mosquitoes were released in the semi-field system per experimental night. HLC were done for four hours (19.00-23.00 h) and the traps ran over night for 12 hours.

Temperature and relative humidity variations during experiments

During all experimental set ups, the temperature and relative humidity were logged to determine possible variations between experiments and impact on experimental output. The mean temperature during the experimental hours for the baseline experiment(open eaves, no trap) was 23.6° C (95% CI 23.4-23.7), while it was for the pull-only set ups 24.7°C (95% CI 24.6-24.8). The mean temperature during the final push-pull experiment was with a mean of 22.2°C (95% CI 22.1-22.3) nearly one degree lower that during the push-only experiment with transfluthrin 23°C(95% CI 22.8-23.1). The relative humidity was maintained at >70% during the experimental hours throughout the different experimental set ups.

Detection of volatile chemicals released by the push-pull components

Air movement and temperature variations during chemical quantification

Air samples for chemical analysis were taken in August during the dry season and the average temperatures during the sampling hours were 20.7° C (95% CI 20.4-21.0°C). The air-samplings were done in relatively still air. Air movement was recorded every 10 minutes during the sampling times and most of the recordings (75%) indicated 'no movement'. During the remaining times a low air speed (0.6-1.7 m/s) was measured, consistently from a North East to East North East direction (45°-66°). This meant as indicated in Figure 2, that at the sampling location the air moved from the direction of the hut towards the sampling points.

Detection and estimated concentration of transfluthrin

Transfluthrin was detected at all sampling points. The concentrations decreased greatly with distance from the release point (positions A, B, C and D; Figure 12) with the highest concentration detected nearest to the point of release (position A), and the lowest concentration being detected farthest away

from the point of release (position D). Variations in concentrations at different heights were seen across all the sampling positions (p=0.03)with a general trend for higher transfluthrin concentrations being found at lower sampling points of ≤ 1 m from the ground (Figure8). The averaged concentrations of transfluthrin at position A (nearest to experimental hut) and position D (nearest to Suna trap) were significantly different (p=0.02; Figure 12) as were the concentrations at positions B and D (p=0.002; Figure8). The highest concentration of transfluthrin detected was 26.3 ng/l (95% CI 21.6-31.0 ng/l) at 1 m from the release point (position A) at 0.5 m from the ground. At 1.5m of the same sampling position, the transfluthrin concentration was 5.7ng/l (95% CI 3.1-8.2ng/l). The lowest concentration was 1.7 ng/l (95% CI 1.2-2.3 ng/l), detected 4 m away from the release point (position D) 1.5m above the ground.

Detection and estimated concentration of MB5 constituents

Two of the five MB5 constituents namely L-(+)-lactic acid and ammonia solution were not detectable under the analytical conditions since the stationary phase of the column used was for the detection of non-polar compounds while the two compounds are polar in nature (Short *et al.*, 1983; Hasegawa *et al.*, 2008). The remaining compounds, namely 3-methyl-1-butanol, butan-1-amine and tetradecanoic acid were detected at low concentrations in some, but not all samples.

Of the 60 samples collected, 3-methyl-1-butanol was quantified in only 15 (25%), with the rest falling below the detection limit. The highest concentration of 0.4 ng/l (95% CI 0.07-0.75 ng/l) was detected closest to the release point at position D, 1 m from the Suna trap. Contrasting to all other chemicals in the push-pull system, 3-methyl-1-butanol was found at the highest average concentration at 1.5m above the ground. At the same position, the average concentration was0.13 ng/l (95% CI 0-0.38 ng/l) at 1.0 m above ground and 0.04 ng/l 95% CI 0-0.13ng/l)at 0.5 m.





Figure 8. Hourly mean air temperature in the semi-field systems during the human landing collections (19.00-23.00 h) of different experiments. The data variability is shown with 95% confidence intervals.

Chapter 2



Figure 9. Transfluthrin concentrations estimated from air-sampling at different distances and heights. (A) Median concentrations in nanograms per litre of transfluthrin across the four sampling positions from A (1 m from release point) to position D (4 m from release point) as well as across sampling heights from 0.5m above the ground to 1.5 m. (B) Mean transfluthrin concentration (Standard Error bars) in nanograms per litre of air sampled for every sampling point.

Chapter 2



Figure 10. Average concentrations in nanograms per litre (standard error bars) of 1-butylamine and tetradecanoic acid across all air-sampling positions.

Out of the 60 air samples, 1-butylamine was detected in 31 samples (52%), while tetradecanoic acid was detected in 42 samples (70% of samples). There was no strong association with distance and height for these chemicals, though some trends can be observed from Figure9. The chemical 1-butylamine was consistently detected at higher concentrations closer to the ground (≤ 1 m). This also applied to tetradecanoic acid, at least within 2 m from the release point (positions D and C). Generally, both chemicals were detected at significantly higher concentrations within 2 m of the releasing trap than further away.

Discussion

The results of this study provide essential insight into the behaviour of host-seeking *An. arabiensis* in response to tools aimed at manipulating their odour-orientation and consequently at reducing the number of potentially infectious bites in the peri-domestic area.

Transfluthrin-treated hessian fabric strips loosely fixed around eave gaps prevented, depending on the experimental conditions, between 40-80% of the *An. arabiensis* bites a human volunteer would have received in the absence of the treatment. On the contrary, the microencapsulated Citriodiol® did not show any spatial repellent properties as concluded from the unaffected human landing rates.

For An. arabiensis under the test conditions the components tested for pulling vectors in an attract and kill approach did not perform to expectations based on previous work (Takken et al., 1997; Takken et al., 1999; Okumu et al., 2010b; Hiscox et al., 2014; Mweresa et al., 2014; Mweresa et al., 2015). The Suna trap baited with the MB5 odour-blend and supplemented with CO₂ from molasses fermentation attracted only a third of the released host-seeking females when no human host was in the vicinity, confirming similar studies (Hiscox et al., 2014; Mburu et al., 2017), however, when a human volunteer was in the same system, the trap caught less than 1% of all released mosquitoes whilst a human consistently attracted over two thirds of all mosquitoes released. This was the case in the absence ('pull' only) and in the presence of a spatial repellent ('push-pull' set up). In its current configuration the tested 'pull' did not provide a valuable addition to a spatial repellent in a push-pull system for prevention of An. arabiensis bites. At closer scrutiny, in the absence of a human volunteer, it was found that the MB5 odour blend did not add much additional attraction to the trap, and it might be sufficient to only provide CO_2 from molasses fermentation as bait. This confirms data shown graphically from field indoor trap collections of An. arabiensis(Mburu et al., 2017)in western Kenya though the authors did not discuss this observation. Our experiments also clearly indicated that 2-butanone is not a suitable CO₂ replacement for the collection of An. arabiensis confirming previous observations under similar experimental conditions (Mburu et al., 2017).

Contrary to the majority of recent experimental studies with malaria vectors and odour-baited traps, here we included a human being in the system, since ultimately; a push-pull intervention aims to directly protect people from bites and hence would require traps to compete well with the human host odour when placed in close vicinity. However, human beings remain more attractive to host-seeking malaria vectors despite all the efforts employed in identifying host-seeking cues and to produce synthetic lures due to the high complexity in mosquito host-seeking behaviour (Okumu *et al.*, 2010a; Carde, 2015; van Breugel *et al.*, 2015; Hawkes *et al.*, 2017). Our results support the findings of Okumu et al. who observed that the

chemical odour blend attracted host-seeking *An. arabiensis* in representative numbers in the absence of a human, but when presented with the two odour sources side by side within the same hut in field settings, the mosquitoes retained their preferences for humans(Okumu *et al.*, 2010b). The authors suggested that preferences are dependent upon whether the stimuli are in direct short-range competition or whether they are far apart with completely separated odour plumes, which might be the best strategy to exploit for mass-trapping interventions (Kline, 2007; Hiscox *et al.*, 2016; Homan *et al.*, 2016).

Due to the increasing insecticide resistance levels in malaria vectors, this study aimed to explore Citriodiol® with its active ingredient PMD, as a spatial repellent since it belongs to a different class of chemicals than those currently used in public health. It is a well-known topical repellent (Barnard et al., 2002; Carroll et al., 2006; Colucci et al., 2018; Lee et al., 2018; Carroll et al., 2019; Afify et al., 2020) and has been suggested as having spatial repellent properties in a previous study (Menger et al., 2014). However, the previous evaluation was done using electricity-powered active emanators to dispense the chemical. Furthermore, the product was not microencapsulated but applied as Citriodiol® oil at high concentrations on a nylon strip and several emanators used in a semi-field system less than a quarter of the size of those used here(Menger et al., 2014). Such effort is neither operationally feasible, nor costeffective. Importantly, the previous study did not include a human blood host in the test system but used an MB5-baited trap as a substitute (Menger et al., 2014). We opted for microencapsulation to secure the Citriodiol® into the fabric with the aim to allow passive slow-release of the repellent for possible longterm usage when fixed on eave gaps(Onder et al., 2008; Misni et al., 2017). However, neither of the two test concentrations resulted in any protection against mosquito bites, not even when the material was fixed very closely to the human volunteer on the chair (data not shown). Optimal formulation and presentation of a repellent, whether spatial or topical, is key for effectiveness (Carroll et al., 2019) and further work might be warranted. For now, it remains unclear, if the concentrations of chemicals released from the fabric were just too low, or whether PMD does in fact have no spatial repellent properties.

Transfluthrin is a pyrethroid insecticide which is not only known for its killing effect but also its moderate volatility which makes it an effective spatial repellent (Ogoma *et al.*, 2012b; Wagman *et al.*, 2015a; Nentwig *et al.*, 2016; Mmbando *et al.*, 2018). Transfluthrin has been incorporated into commercial products for mosquito control with encouraging outcomes (Ramesh *et al.*, 2001; Pates *et al.*, 2002; Hill *et al.*, 2014; Jeyalakshmi *et al.*, 2014; Ogoma *et al.*, 2014a; Ogoma *et al.*, 2014b; Maia *et al.*, 2016). Applied on hessian material for passive emanation, it has been proposed to protect from 70-90% of bites from Afrotropical malaria vectors in a range of experimental laboratory and field studies implemented in coastal and inland Tanzania(Ogoma *et al.*, 2012b; Ogoma *et al.*, 2017; Mmbando *et al.*, 2018; Mmbando *et al.*, 2019; Mwanga *et al.*, 2019). Our results confirm the potential of transfluthrin for use as a spatial

repellent vector control tool. However, the protective efficacy under most of our test conditions, was much more moderate than in the Tanzanian studies. One reason for this might be the differences in average temperatures during experiments in the different regions (Russell et al., 2010; Ogoma et al., 2012a; Andres et al., 2015; Ogoma et al., 2017; Bibbs et al., 2018). During the implementation of our final push-pull experiment, only around 40% of the bites that would have been received without protection were averted. This was only half the protection we found for the 2.5 g/m²transfluthrin treatment during the push-only experiment. Given that the pull-only experiments did not provide any evidence that the presence of the odour-baited trap might increase the proportion of mosquitoes attempting to bite the human volunteer, other factors are likely responsible for the lower protection from the spatial repellent at the time. Notably, during the push-pull experiment, evening temperatures were an average of 22 °C, around 1 °C lower than during the push-only experiment. Increases in temperature increase the effective vapour pressure of a chemical and therefore the volatilization rate(Jensen et al., 2007). Cooler temperature conditions lead to lower transfluthrin evaporation rates and it has been previously suggested that the protective efficacy of passively emanated transfluthrin from hessian fabric reduces when temperatures are below 23°C (Ogoma et al., 2017). Conversely, increasing temperatures were associated with increasing airborne transfluthrin concentrations in closed test systems (Pettebone, 2014) and an increase of mosquito mortality with an increase of airborne transfluthrin(Martin et al., 2020).

Our samples for quantification of transfluthrin in the air within 5 m from the release point were taken during the cold season with temperatures during sampling of around 21 °C. Nevertheless, the chemical was consistently detected with concentrations decreasing by an order of magnitude from over 20 ng/l to 1.7 ng/l over five metres from the release point. These concentrations are significantly higher than those reported by Ogoma et al. (Ogoma *et al.*, 2017) who reported 0.13 ng/l from samples collected indoors from a non-ventilated 30 m³ room, however, the treatment load of the hessian test material was also three times lower (0.8 g/m²) than in our study. Our estimated concentrations are, nonetheless, well below the maximum acceptable exposure concentration for long-term inhalation exposure of human beings of 500 ng/l, as defined by the regulatory authorities of the European Union (Weinholt, 2016) Our findings relate well with a more recent study using a similar approach to ours on malaria vectors in Vietnam, where airborne transfluthrin concentrations were estimated at 1.32 ng/l at 4 metres from the release point(Martin *et al.*, 2020) and were below the detection limit further away. This study also showed higher knock-down and mortality rates for caged mosquitoes at ground level than above 1 metre from the ground (Martin *et al.*, 2020) supporting our observation of highest concentrations at 1.0 m and below. This might limit the 'protective bubble' especially in the outdoor environment around the house.

The inconsistent detection of the constituents of the putative attractant MB5 in the air, might suggest that the odours were not sufficiently released and dispersed, specifically 3-methyl-1-butanol was rarely picked up by the adsorbent filters. Tetradecanoic acid and 1-butylamine were detected more frequently, though not consistently and at very low concentrations in close vicinity to the trap. Whether the chemical release rates and hence performance of the MB5-baited Suna trap might also be affected by night-time temperatures during trapping needs further investigations. Mechanisms to increase the released concentrations and improve the dispersion might be explored in future by modifying the Suna traps(Verhulst *et al.*, 2015)or surveying alternative traps and baits (Menger *et al.*, 2015; Batista *et al.*, 2017; Hawkes *et al.*, 2017; Batista *et al.*, 2018; van de Straat *et al.*, 2019). A recent study for example suggested the combination of transfluthrin treated fabric on eave gaps with BG Malaria traps (Biogents, Germany) and suggested a larger distance of the pull trap from the human host, nevertheless the authors also found only a very marginal addition of protection from the traps in preventing *An. arabiensis* bites(Mmbando *et al.*, 2019). Strategies combining mass-trapping of mosquitoes with spatial repellents at independent locations rather than combining on household levels should be explored in future.

Finding a pull component that is efficient enough to attract mosquitoes even when a human is close by, yet easy to set and maintain, remains desirable to remove adult vectors from the transmission setting. The idea to develop a push-pull system with the MB5-baited Suna trap was inspired by the successful mass-trapping field trial in western Kenya with the pull component only, which was associated with significant reductions in vector densities (Hiscox *et al.*, 2016; Homan *et al.*, 2016). However, when analysed by vector species, only *An. funestus* densities were reduced in the study site whilst *An. arabiensis*, which accounted for around a quarter of the vector population, were not affected, resulting in only a moderate reduction in malaria parasite prevalence in the study area (Hiscox *et al.*, 2016; Homan *et al.*, 2016). This suggests a species-specific attraction to the MB5-baited trap. Similar observations were made by Mburu et al. (Mburu *et al.*, 2017)when investigating the use of 2-butanone as a CO_2 replacement in a rice irrigation area in western Kenya, where *An. funestus* is the predominant vector species.

Conclusions

This detailed step-by-step evaluation of the selected putative repellent 'push' and attractive 'pull' components has led to a better understanding of their prospect to affect the host-seeking behaviour of the malaria vector *An. arabiensis* in the peri-domestic space and helps to gauge the impact such tools would have when used in the field for vector monitoring or control.

This study has highlighted the need for testing odour-based interventions in the presence of a human host to gain accurate estimates. A trap cannot substitute a human being when changes in attraction and human landing rates are the outcome measure. Additionally, the importance of working with different vector systems has been elucidated. There is urgent need to further study potential differences in odour-orientation between the two major vector species complexes: *An. gambiae* s.l. and *An. funestus* group. Overall, it will be desirable to develop odour-baited traps that target all major vectors at the same time for use in varied eco-epidemiological systems.

This study further confirmed that, at least under standardised experimental conditions, that passive emanation of transfluthrin from treated hessian fabric strips around eave gaps can provide protection from mosquito bites in the peri-domestic space. Comparisons across published work have, however, also highlighted that the expected impact might be quite variable from location to location, depending on climate conditions and vector species. Data generated under standardized field conditions in a single location needs to be interpreted in the local context and should be replicated under different conditions to ensure recommendations can be generalised or can be tailored to local contexts. Mathematical modelling can support decision making by integrating data from different settings in prediction models to understand the geographical range where such tool might be useful and the impact to be expected under a varying environmental and epidemiological conditions. Field evaluations are required to investigate how results from semi-field experiments correlate to findings from field trials. For example, air movement was minimal in the semi-field systems and the repellent transfluthrin was detected within 5 m from the experimental hut. However, it must be assumed that this is quite different from natural conditions, especially during rainy seasons when vector densities and malaria transmission peak. Rainstorms characterising the tropical evenings might well interfere with the odour plumes and protective bubble around the house. One might therefore plausibly assume that the protective efficacy in the peri-domestic space in western Kenya field sites would be lower than the largely moderate effects observed in the current experiments.

Acknowledgements

We thank Dan Masiga for facilitation of this research, Baldwyn Torto and Xavier Cheseto for providing facilities and support with the GC analyses, Elisha Obudho and David Alila for providing mosquitoes for this study. We are grateful for the tireless work of the human landing volunteers Leonard Wanga (deceased), Shem Matthews Otieno, Dan Simiyu, and Philip Owigo. We also acknowledge the technical support of Mgeni Mohamed Tambwe, Bruno Otieno and Ferdinand Ong'wen. We thank Devan

Micropolis S.A. through Ana Carreira and Roberto Teixeira and Utexbel N.V. through Jean-Luc Derycke for their technical support.

Declarations

Ethics approval and consent to participate

This study was approved by the Kenya Medical Research Institutes Scientific and Ethics Review committee (KEMRI-SERU), protocol number NON-KEMRI 546.

Consent for publication

Not applicable.

Availability of data and materials

The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

Competing interests

The authors declare that they have no competing interests.

Funding

We gratefully acknowledge the research funding from the Innovative Vector Control Consortium (IVCC) and support provided through *icipe* core funding by the UK's Foreign, Commonwealth& Development Office (FCDO); the Swedish International Development Cooperation Agency (Sida); the Swiss Agency for Development and Cooperation (SDC); the Federal Democratic Republic of Ethiopia; and the Government of the Republic of Kenya. The views expressed herein do not necessarily reflect the official opinion of the donors. The funders had neither a role in the design of the study, nor in collection, analysis, interpretation of data, nor in writing the manuscript.

Authors' contributions

AH, AS, SM and WT conceived the initial idea for this research and secured the funding. AH, AS, SM, UF and MMN jointly developed the experimental designs for the experiments. MMN, UF and AH

developed experimental standard operating procedures. MMN implemented the experiments. MMN and UF analysed and interpreted the data and drafted the manuscript with support from AH, WT and JvL. All authors contributed to the final draft. All authors read and approved the final manuscript.

Sight versus smell: Dynamics of malaria mosquito attraction response in the presence of spatial repellents

Author list:

Margaret M. Njoroge, Alexandra Hiscox, Joop J.A. van Loon, Willem Takken, Ulrike Fillinger.

To be submitted in a modified form

Abstract

Mosquito behaviour is influenced by perception of and response to cues that provide information on host location and quality. Olfaction plays a central role in this process, with volatile attractants and repellents affecting mosquito behaviour. Strategies to manipulate the odour orientation of mosquitoes are currently under development to support vector control efforts for diseases vectored by mosquitoes such as malaria. Spatial repellents appear to be a promising new tool for vector control, but in order to prevent diversion of bites to unprotected persons, it would be beneficial to combine the spatial repellent with an attractantbased trapping method. Here we explore if and how the spatial repellent transfluthrin affects the trapping efficacy of Anopheles arabiensis using attractant-baited Suna and CDC UV light traps, both supplemented with CO2, under semi-field conditions and compare with that of An. gambiae s.s. Dual choice set ups were conducted in screen houses containing experimental huts with eave fabrics treated with spatial repellent or untreated. In the absence of a repellent, dual choice set-ups showed a relative 1:1 distribution of all trapped mosquitoes between the two trap types, although the proportion of trapped mosquitoes was less than 1/3 of all released. Trapping efficiency of an odour-baited Suna trap for An. arabiensis was reduced 2.3 times in the presence of a transfluthrin-treated eave fabric compared to when the odour-baited Suna trap was used in the presence of untreated eave fabric. CDC UV light traps trapping efficiency calculated as recaptured proportion of released mosquitoes remained relatively constant both in the presence and in the absence of the spatial repellent. Trapping efficiency for An. gambiae S.S. was not significantly different to An. arabiensis though fewer An. gambiae s.s. were trapped in odour-baited Suna trap (p=0.767) and more caught in the CDC UV light trap (p=0.178) in the presence of the repellent. The results showed that in very close proximity to a repellent, a CDC UV light trap would be preferred instead of an odour-baited Suna trap as a suitable inclusion in a combination set up for the control of malaria vectors though a more suitable alternative pull component should be considered.

Key words:

Anopheles arabiensis, Spatial repellent, CDC UV light trap, Olfaction, Eave-fabric, Dual-choice

Introduction

Understanding host-seeking responses of mosquitoes to chemical and physical cues is key to the development and improvement of tools aimed at vector surveillance and the management of diseases mosquitoes transmit (Mbogo *et al.*, 1993; Magbity *et al.*, 2002; Kline, 2007). Odour-baited traps targeting host seeking or egg-laying vectors were developed by studying mosquito biology and identifying the cues that mosquitoes utilize for locating a blood host or oviposition site (Takken *et al.*, 1997a; Spitzen *et al.*, 2008; Turner *et al.*, 2011; Dugassa *et al.*, 2013; Revay *et al.*, 2013; Wong *et al.*, 2013; Hiscox *et al.*, 2014; Wagman *et al.*, 2014). The long-term success of control tools reliant on insecticides has recently been questioned due to increased insecticide resistance among malaria vectors to all major groups of insecticides (Agossa *et al.*, 2015; Liu, 2015; Sande *et al.*, 2015; Hancock *et al.*, 2020). This has contributed to the stalling in decline of malaria cases and, in some areas, resurgence of malaria, emphasising the need for development of supplementary tools for enhanced control (WHO, 2019).

Recent developments in vector control have explored the use of insecticides used at sub-lethal concentrations as spatial repellents to divert host-seeking mosquitoes away from the potential human blood host and hence aim for a reduction in infective bites (Hill *et al.*, 2007; Achee *et al.*, 2012; Revay *et al.*, 2013; Maia *et al.*, 2016; LeClair *et al.*, 2017). Use of spatial repellents, such as transfluthrin for conferring outdoor protection, holds a lot of promise though there is the potential risk of diverting mosquitoes to unprotected populations (Achee *et al.*, 2009; Achee *et al.*, 2012; Maia *et al.*, 2016; Ogoma *et al.*, 2017; Maia *et al.*, 2018). To counter this, the use of spatial repellents in combination with traps that attract and kill the vector has been proposed (Takken, 2010; Wagman *et al.*, 2015).

Host-seeking behavioural responses in *Anopheles* mosquitoes have been largely attributed to the perception of carbon dioxide (CO₂) gradients and a combination of human odours, moisture and warmth that the mosquitoes use to orient themselves to a blood host (Healy *et al.*, 1995; Takken *et al.*, 1997a; Takken *et al.*, 1997b; Gibson *et al.*, 1999; Okumu *et al.*, 2010c; Turner *et al.*, 2011; Ray, 2015). Identification of specific host-odours associated with host-seeking has led to the development and utilization of synthetic lures used in traps (Okumu *et al.*, 2010b; Mukabana *et al.*, 2012a; Hiscox *et al.*, 2014; Hiscox *et al.*, 2016; Homan *et al.*, 2016). Using an odour-baited trap in combination with a spatial repellent for the control of outdoor-biting mosquitoes could potentially trap repelled mosquitoes and remove them from the environment instead of redirecting them to unprotected populations (Menger *et al.*, 2014; Menger *et al.*, 2016). However, studies with *Aedes albopictus* have shown that exposure to spatial repellents (geraniol, citral, eugenol and anisaldehyde) leads to a reduction in host-seeking behaviour

(Haoet al., 2008). Others found that repellent insecticides induce an irritant reaction leading to reduced feeding in *Anopheles* mosquitoes due to avoidance behaviour (Chareonviriyaphap *et al.*, 2004; Hodson *et al.*, 2016). Overall, there is evidence that repellents affect the odorant receptors of dipterans by affecting the ability of the insect to perceive attractive chemicals (Bohbot *et al.*, 2010; Hodson *et al.*, 2016; Afify *et al.*, 2020) which would be counter-productive in a combined push-pull intervention, particularly where the pull device is in proximity to the push device. Indeed, DEET has been shown to reduce the perception of human odours by *An. gambiae* olfactory receptor neurons (Ditzen *et al.*, 2008). The attractiveness of a lure in an odour-baited trap within the action range of a spatial repellent may be masked similarly to the odours from the natural blood host which should be protected from bites. Such an effect would render the trap less effective than intended (Achee *et al.*, 2012). Consequently, it is necessary to investigate how receptive mosquitoes exposed to repellents are to simultaneous exposure to potential chemical lures and alternative non-chemical cues.

Mosquito perception of light has been shown to aid in most of their biological activities including visual perception of blood hosts (Allan *et al.*, 1987; Lehane, 2005; McMeniman *et al.*, 2014). While mosquitoes are nocturnal in nature and usually feed when dark (Wamae *et al.*, 2015), light plays a major role in modulating their behaviour (Rowland, 1989). Exposure to short bursts of bright white light for example, prior to the start of the dark phase of the daily cycle, greatly reduced their host-seeking ability (Sheppard *et al.*, 2017).

Mosquitoes generally respond to light intensity levels by initiating flight activity followed by orientation based on visual perception towards an intended target (Allan *et al.*, 1987). To some extent, moonlight affects mosquito behaviour by triggering increase in flight in *An. taeniorhynchus* (Bidlingmayer, 1964). Host-seeking *Aedes* spp. and *Mansonia perturbans* female mosquitoes are drawn more to low intensity light such as blue, red and black than to high intensity ones such as yellow and white (Browne *et al.*, 1981). Like most vertebrates and invertebrates, mosquitoes have UV-sensitive rhodopsins, with *An. gambiae* photoreceptors specifically those that possess Agop8 rhodopsins, being shown to be particularly important in the detection of UV light (Hu *et al.*, 2014). As all mosquitoes require sugar meals for energy, it has been shown that some flowers have UV-reflecting guides (Silberglied, 1979) which mosquitoes, many of whose UV receptors are sensitive in the 340-360nm range, follow to obtain nectar (Stark *et al.*, 1982). The role of light in mosquitoes which are in the host-seeking phase of their gonotrophic cycle works in tandem with olfactory perception. This was seen in a study that showed how *Ae. aegypti* detection of CO₂ activated a significant attraction to visual features which prompted the mosquito to move closer to the potential host in order to detect other shorter-range cues such as host volatiles and heat (van Breugel *et al.*, 2015). Light traps utilize this perception and attraction to light of different

wavelengths with the combination of odour stimuli and air currents for monitoring and surveillance of mosquito populations (Joshi *et al.*, 1975; Fornadel *et al.*, 2010; González *et al.*, 2016; Ponlawat *et al.*, 2017b). The performance of light traps in attracting and killing mosquitoes has been shown, in many cases, to be species-specific (Wong *et al.*, 2013; Wagman *et al.*, 2014; Mburu *et al.*, 2019). The majority of commercial light traps employ the use of 4-6 W incandescent bulbs that produce a broad visual spectrum of light that goes beyond the visible light range (Evans *et al.*, 2009). Changes in specification of the light wavelength to more specific wavelengths have improved trapping efficiency for different mosquito species. UV light has been shown to be superior to incandescent light in trapping mosquitoes (Moore *et al.*, 2001; Li *et al.*, 2015b).

In the present study the trapping efficiency of CDC UV light traps was compared with the efficiency of MB5-baited Suna traps supplemented with CO_2 in the presence and absence of a spatial repellent under semi-field conditions. Mosquitoes released in the semi-field system were either momentarily or continuously exposed to transfluthrin as a spatial repellent. The sibling species *An. gambiae s.s.* and *An. arabiensis* were compared to investigate any species-specific responses.

Materials and methods

Experimental site

Small semi-field systems located at the International Centre of Insect Physiology and Ecology – Thomas Odhiambo Campus, Mbita, western Kenya (-0.430356, 34.206416) were used for all the experiments. The systems measured 11.4 m by 7.1 m and were screened with fibreglass netting. The roofs were made of translucent polycarbonate and the soil floors had a top layer of 50 cm of sand (Okal *et al.*, 2015). The floors were watered regularly to maintain a relative humidity between 60% and 70%, an optimal environmental condition for behavioural assays on host-seeking mosquitoes (Benedict, 1997). In each semi-field system, a traditional East African mud hut 3.1 m by 3 m and with a height of 2.2 m, with corrugated iron sheets on the roof, was located at the centre. The hut had one window and one door which were both screened and remained closed throughout the experiments, while the 10 cm wide eave gaps between the roof and wall were open all round (Figure 1). A 5 cm wide hessian fabric (burlap, which was procured from local markets) impregnated with transfluthrin was wrapped around the eave gaps of the hut leaving approximately 2.5 cm gap above and below the fabric.

Mosquitoes

Two hundred female insectary-reared *An. arabiensis* Mwea strain were used for each experimental replicate. In two experiments, comparison of behavioural responses was done using *An. gambiae s.s.* Mbita strain, also reared in the same conditions as *An. arabiensis*. Nulliparous females aged 3-6 d post-



Figure 1. Semi-field system $(11 \times 7 \text{ m})$ at *icipe*-TOC containing a traditional hut $(3 \times 3.1 \text{ m})$ made of mud walls and iron sheets for the roof.

emergence were selected from rearing cages. Mosquitoes were reared under ambient environmental conditions with the adults being fed on 6% glucose solution through wicks made from absorbent tissue paper from the time of emergence (Benedict, 1997).

The mosquitoes were selected from the cages into paper cups using hand-held mouth aspirators, by choosing the females that were activated by the presence of a hand close to the side of the cage. They were then starved (no access to water or glucose solution) for 5 h prior to experiments in a holding room that had similar conditions as the insectary.

Chapter 3



Figure 2. Schematic presentation of experimental design. Traps were placed diagonally opposite each other and were rotated through all corners (A-D) of the semi-field system to remove side bias over the 16 experimental nights. Mosquitoes were released from two sides of an experimental hut from release cups holding 50 mosquitoes each.

Odour-baited Suna trap

The combination of CO₂ and a synthetic odour-blend referred to as the Mbita blend 5 or MB5 (Mukabana *et al.*, 2012b; Mweresa *et al.*, 2014) was used in Suna traps (Biogents, Germany). Their development, structure and *modus operandi* are described elsewhere (Hiscox *et al.*, 2014). The MB5 comprised of five constituent chemicals namely; ammonia (2.5% in water), lactic acid (85% in water), tetradecanoic acid (0.00025 g/l in ethanol), 3-methyl-1-butanol (0.000001% in water), and butan-1-amine (0.001% in paraffin oil) (Mukabana *et al.*, 2012b; Menger *et al.*, 2014; Mweresa *et al.*, 2014). All chemical compounds were purchased from Sigma Aldrich Chemicals GmbH (Germany). Each component of MB5 was dispensed from an individual strip of nylon (measuring 26.5 cm × 1.0 cm). The nylon strips (15 denier microfibers, 90% polyamide, 10% spandex) were purchased from Bata Shoe Company Ltd, Kenya. Baits were prepared and used for a maximum of one experimental run (16 experimental nights) before fresh ones would be prepared. Every evening, the baits would be removed from an *-*20 °C freezer and hung in the traps for the entire night before careful removal in the morning and placing in a clean

aluminium foil and then folded into the foil (without bending the strips) and put back into the freezer. Carbon dioxide was produced by mixing 250 g of molasses (Mumias Sugar Company Ltd, Kenya), 17.5 g dry yeast (Angel[®] Yeast Company Ltd, China) and 21 of water (Saitoh *et al.*, 2004; Oli *et al.*, 2005; Mweresa *et al.*, 2014) with a fresh mixture used for each experimental night.

CDC-UV light trap

CDC-UV light traps from Bioquip Products (California, USA) were used for all the experiments. A blueblack light tube and transistorized inverter ballast provided radiation in the near UV range (4 W, ca. 320-420 nm) and was purchased with the trap used in the experiments (Bioquip Inc). The CDC-UV light trap was supplemented with CO_2 (produced from molasses) in the same way as the odour-baited Suna trap to elicit long-range activation of mosquitoes. For both traps, the CO_2 was released from a tube connected near the fan of the trap below the trap cover.

Spatial repellent

Emulsified concentrate (EC) of transfluthrin from Bayer Global (Leverkusen, Germany) at a concentration of 0.2 g/ml was used as spatial repellent. Hessian fabric measuring 21 m by 5 cm (Ogoma *et al.*, 2012) was impregnated by soaking it in transfluthrin EC solution to achieve a final load of 2.5 g/m². Preparation of the EC solution involved measurement of water required to sufficiently wet one roll of eave fabric (approximately 1300 ml of water per fabric) and adding 26.25 ml to attain 2.5 g/m² as the area of the fabric was 2.1 m²(Ogoma *et al.*, 2012; Ogoma *et al.*, 2017). This was carried out separately for each strip. Fabric was dried in the shade prior to use. The treated fabric was used continuously for up to 16 nights with the fabric being up 24 h daily (complete single experiment run). In experiments without exposure to transfluthrin (control experiments) or in which mosquitoes were exposed to the spatial repellent briefly before being taken to the semi-field system, untreated hessian fabric of similar dimensions (21 m by 5 cm) were prepared. Treated and untreated hessian wraps were hung in the same way around the eave gaps of the huts.

Experimental set up

All experiments were dual choice tests, where host-seeking *An. arabiensis*(in all experiments) and *An. gambiae s.s.* (in two experiments only) were released in the presence of two traps, either an odour-baited

Suna trap and a CDC-UV light trap or two specimens of the same trap types. The traps were both supplemented with CO_2 at any time. Each trap was mounted on a tripod stand with the mosquito entry point of the trap at 30 cm from the ground (Hiscox *et al.*, 2014). They were placed diagonally opposite to each other. Their positioning was changed nightly following a clockwise rotation around the semi-field system for 16 nights to remove location or side bias (Figure 2). Mosquitoes were released at two sides of the experimental hut from a total of 4 release cups with 50 mosquitoes in each cup (2 cups per side). In the experiments with both mosquito species, 25 specimens of each species were released from a cup with the mosquitoes colour-coded with two different fluorescent dyes; green for *An. gambiae s.s.* and red for *An. arabiensis*. The dusting was done by placing approximately 1 g of fluorescent dye (Fingerprint powder, BVDA International B.V., the Netherlands) in a 10 ml syringe (without a needle) then blasting this into a release cup containing the mosquitoes (Verhulst *et al.*, 2013). Traps were switched on at 18:30 h and mosquitoes obtained from either the holding room or wind tunnel as described below were released at 19:00 h. The traps were removed after 12 collection hours and all mosquitoes in the catch bags from either trap were counted.

Preliminary experiments

Dual choice tests aimed at identifying a preference for one trap type over the other. If both traps are equally attractive to host-seeking mosquitoes (or equally efficient in collecting them), as might be assumed by providing two identical traps, a 1:1 distribution of captured mosquitoes would be expected over the course of the 16 experimental nights. If one trap is more efficient than the other, we expect a significant deviation from the reference 1:1 distribution. In order to understand the dynamics in the semi-field system in terms of trapping efficacy in dual choice tests without any treatments in place, a set of preliminary experiments were implemented (Table 1, experiment 1-3; preliminary experiments). The first set of experiments (experiments 1 and 2) provided two equal traps (2 Suna traps and 2 CDC UV light traps respectively), whilst experiment 3 provided the two different trap types. The second set of experiments was implemented with either untreated or transfluthrin treated hessian fabric wrapped around the eave gaps of the experimental hut in the semi-field system (Table 1, experiments 4-6; effect of repellent on trapping).

Pre-exposure experiment

Mosquitoes were pre-exposed to a repellent immediately prior to release in the semi-field system to determine whether any changes in behaviour may result from a short exposure lasting 5 min. Specimens that had been starved for 5 h were exposed to transfluthrin by placing them in single-use release cups in a wind tunnel for 5 min. in the presence of a 10 cm x 30 cm hessian fabric impregnated with 0.075 g of transfluthrin (load on fabric = 2.5 g/m^2) which was positioned 30 cm from the release cups. The wind tunnel was switched off during the exposure; hence there was no directional airflow. Preliminary experiments confirmed that there was no mortality induced by this exposure for a 12 h post-exposure observation period. After exposure, the treated fabric was removed, wrapped in aluminium foil and stored in a cold room at 2°C for reuse for up to 16 experimental days.

Table 1: Overview of all two-choice semi-field experiments (no. 1-6) to compare the efficacy of trapping released insectary-reared *Anopheles* mosquitoes with odour-baited Suna and CDC UV light traps in the absence and presence of the spatial repellent transfluthrin. All traps were supplemented with CO₂.

Experiments	Trap 1	Trap 2	Test species			
Preliminary experiments in absence of fabric around eave gaps						
(1) Reference 1 – 2 equal trap choices (odour vs. odour)	odour-baited	odour-baited	AA			
(2) Reference 2 – 2 equal trap choices (UV light vs. UV light)	UV light	UV light	AA			
(3) Odour versus light - 2 trap choice	odour-baited	UV light	AA			
Experiments in presence of fabric around eave gaps investigating the effect of transfluthrin on						
trapping						
(4) Odour versus light - 2 trap choice, untreated fabric	odour-baited	UV light	AA, AG			
(5) Transfluthrin pre-exposed - 2 trap choice, untreated fabric	odour-baited	UV light	AA			
(6) Odour versus light - 2 trap choice, transfluthrin treated fabric	odour-baited	UV light	AA, AG			

AA: Anopheles arabiensis. AG: Anopheles gambiae

The wind tunnel was then wiped down with 70% ethanol and 2-propanol (undiluted laboratory grade) and allowed to dry to prevent accumulation of transfluthrin. Immediately following exposure, mosquitoes were taken for release in the semi-field system which featured an untreated fabric on the eave gaps of the experimental huts (Table 1, experiment 5; pre-exposure). In comparison, unexposed mosquitoes were starved as described before and exposed in a second screen house in parallel to the first (Table 1, experiment 4; unexposed).

Constant exposure experiment. To investigate host-seeking events in the presence of a repellent, transfluthrin-treated hessian fabric was positioned on the eave gap of the experimental hut inside the semi-field system and secured using aluminium wires, ensuring 2.5 cm gaps above and below the fabric. The second semi-field system served as control with an untreated eave fabric positioned in the same way. Test and control were alternated after every 4 nights with a 2 night break in between to ensure that there was no residual effect of the previous treatment. The experiment was replicated over 4 blocks of 4 nights (16 experimental nights, Table 1; experiment 6; exposed).

Data analysis

Data analyses were done using R version 3.5.1 (R Core Team, 2018). All mosquito catches were analysed as proportions, either relative to all mosquitoes released in the semi-field system or relative to all mosquitoes recaptured with the two traps. The total number of mosquitoes caught in both traps was defined as the 'responding' mosquitoes whilst those released but not recollected in either trap were classified as having 'not responded' either because they were not in a host-seeking stage, were killed by predators (e.g. spiders, geckos) before responding or were not attracted/trapped by any of the traps. The dual-choice assays were analysed using generalized linear models with a quasibinomial distribution fitted to account for over-dispersion. The proportions of mosquitoes collected in trap 1 or trap 2 were compared across experiments (dual choice tests implemented in different semi-field systems) in the absence of any treatments for the preliminary experiments and across experiments in the absence or presence of transfluthrin exposure in the experimental set up (Table 1). It was hypothesized that host-seeking females presented with an identical choice would be trapped in approximately equal proportions in both traps. The statistical analysis aimed to reveal if the test treatment of interest (trap) received an increased or decreased proportion of the total number recaptured in an experiment compared to the proportion recaptured in the reference group. Therefore, the experiment ID (Table 1; 1-6) was included in the model as fixed factor to analyse the impact of the experimental conditions on the outcome. The mean proportion of mosquitoes trapped in trap 2 under different experimental conditions and their 95% CIs were estimated based on the model by transforming the log odds (logit) of the outcome to the odds scale and from the odds scale to the probability scale. The data are graphically presented as box and whisker plots showing proportional distribution of trapped mosquitoes in reference traps with the middle line representing the median proportion with the whiskers indicating the maximum and minimum proportional catches.

Ethical considerations

This study was approved by the Kenya Medical Research Institutes Scientific and Ethics Review committee (KEMRI-SERU), protocol number NON-KEMRI 546.

Results

Preliminary choice experiments (Experiments 1-3, No eave wrap)

Dual choice experiments testing two identical traps were implemented to confirm the performance of the bioassay system; one with two MB5-baited Suna traps and another with two CDC UV light traps in one screen house; all traps were supplemented with CO₂. The third experiment provided a choice between the two different trap types. The proportion of mosquitoes trapped in trap 1 out of the total trapped was not associated with the experiment (Table 2, Figure 3A). Tests in which two identical traps were presented had, as expected, an approximate ratio of 1:1 of mosquitoes trapped by the two traps. A similar 1:1 ratio was observed when two different traps were provided as choice.

The proportional outcome was less variable around the median when the equal choices were odour-baited Suna traps (green box and whisker plot in Figure 3A), as opposed to a much higher variability in the nightly response when two UV light traps, or one UV light trap and one odour-baited Suna trap were used (blue and yellow box and whisker plot in Figure 3A).

Whilst the ratios indicate how the two traps compared, they did not provide insight as to how efficient any of them was in trapping a given mosquito population under semi-field conditions. It was hence important to additionally explore the recapture rate out of all mosquitoes released (as opposed to all trapped) in the relatively small semi-field system. This provides an indicator for attraction and/or trapping efficiency under the different experimental conditions.

utputs was done separately for preliminary choice experiments in the absence of an eave	pre-exposed mosquitoes in experiments with eave fabrics. For both, modelling was done	up 1 out of all trapped and secondly for overall efficiency of both traps out of all mosquitoes	
tical model outputs was done separately for pr	transfluthrin pre-exposed mosquitoes in expen-	efficacy of trap 1 out of all trapped and second	
Table 2: Summary of statist	fabric and for the impact of	firstly for relative trapping e	released.

			Mosquitoes recapt	ured in trap 1 out		Mosquitoes recaptured	with both traps or	Ħ
			of <u>all mosquit</u>	oes trapped		of <u>all mosqu</u>	uitoes released	
			Estimated * mean			Estimated [*] mean		
			percentage (%)	Odds Ratio	Å	percentage (%)	Odds Ratio	4
	Trap 1	Trap 2	in trap 1	(95% CI)	value	trapped	(95% CI)	value
Preliminary experiments in	absence c	of eave fal	bric					
2 equal trap choices	Odour	Odour						
(experiment 1)	-baited	-baited	51.4 (95% CI 45.9-56.9)	-1		53.7 (95% CI 46.1-61.3)	-1	
2 equal trap choices								
(experiment 2)	UV light	UV light	54.4 (95% CI 45.7-62.8)	1.13 (0.75-1.68)	0.564	21.9 (95% CI 16.2-29.0)	0.24 (0.15-0.39)	<0.001
2 trap types choice	Odour							
(experiment 3)	-baited	UV light	45.8 (95% CI 37.6-54.3)	0.80 (0.54-1.18)	0.265	23.0 (95% CI 17.1-30.0)	0.26 (0.16-0.41)	<0.001
Pre-exposure experiment in	n presence	e of eave f	abric					
2 trap choices, unexposed	Odour		_	_				
mosquitoes (experiment 4)	-baited	UV light	60.6 (95% CI 48.1-71.8)	1		34.1(95% CI 24.3-45.5)	1	
2 trap choices, transfluthrin								
pre-exposed mosquitoes	Odour							
(experiment 5)	baited	UV light	57.0 (95% CI 47.6-66.0)	0.86 (0.45-1.64)	0.651	33.1(95% CI 25.0-41.9)	0.96 (0.53-1.74)	0.883
*based on statistical model								

Chapter 3



Figure 3A-C. Data exploration across the 6 different experiments (choice assays): The box plots illustrate the proportion of released *An. arabiensis* trapped in the reference trap no. 1 **out of all mosquitoes trapped**. In all but experiment 2, trap no. 1 was an odour-baited Suna trap. Experiments 1-3 were reference experiments providing two traps as choice in the absence of any treatment. Experiments 4-6 contained either untreated (experiments 4 and 5) or transfluthrin-treated fabric around the eave gaps of the hut; mosquitoes were either unexposed to transfluthrin before release (experiments 4 and 6) or pre-exposed (experiment 5). Different colours indicate different choice tests and experimental conditions (see Table 1).

Combined trap catches in the preliminary choice experiments were highest when two odour-baited Suna traps supplemented with CO₂ were set in the semi-field system as they recaptured around half of all mosquitoes released (Table 2; Figure 4A). When two CDC UV light traps, also supplemented with CO₂, were set up, the odds of trapping a released mosquito with two CDC UV light traps was 4 times lower than the odds when two Suna traps were used, with the collective recapture being only around a fifth of the mosquitoes released (Table 2; Figure 4A). A similar collective recapture rate was found when one Suna odour-baited trap and one CDC UV light trap were presented in the choice system (Table 2; Figure 4A).



absence of any treatment. Experiments 4-6 contained either untreated (experiments 4 and 5) or transfluthrin-treated fabric around the eave gaps of the hut and mosquitoes were either unexposed to transfluthrin before release (experiments 4 and 6) or pre-exposed (experiment 5). arabiensis trapped out of all mosquitoes released. (A) Proportion trapped with both traps. (B) Proportion trapped with odour-baited Suna trap. (C) Proportion trapped with CDC UV light trap. Experiments 1-3 were reference experiments providing two traps as choice in the Figure 4. Data exploration across the 6 different experiments (choice assays): The box plots illustrate the proportion of released An. Different colours indicate different choice tests and experimental conditions (see Table 1).

Table 3: Summary of statistical model outputs seeking to investigate an association between the proportions of mosquitoes recaptured with traps and constant exposure of mosquitoes to transfluthrin.

Constant exposure experiment in presen	ce of eave fabric			
	Estimated* mean percentage (%) of An. gambiae s.s. & An. arabiensis	Odds Ratio (95% CI)	p- value	
Mosquitoes recaptured in <i>odour-baited Suna</i> trap out of all mosquitoes trapped				
Experiment				
Untreated eave fabric (experiment 4)	59.0 (95% CI 51.9-65.7)	1		
transfluthrin treated fabric (experiment 6)	29.7 (95% CI 20.9-40.4)	0.45 (0.22-0.92)	0.032	
Species				
An. arabiensis		1		
An. gambiae		0.85 (0.49-1.49)	0.581	
Experiment*Species interaction				
transfluthrin treated fabric * An. arabiensis		1		
transfluthrin treated fabric * An. gambiae		0.39 (0.13-1.20)	0.105	
Mosquitoes recaptured with <i>odour-baited Suna</i> trap out of all mosquitoes <u>released</u>				
Experiment	*	·		
Untreated eave fabric (experiment 4)	21.6 (95% CI 16.1-28.1)	1		
transfluthrin treated fabric (experiment 6)	4.94 (95% CI 2.56-9.32)	0.29 (0.12-0.73)	0.011	
Species				
An. arabiensis		1		
An. gambiae		1.10 (0.57-2.17)	0.767	
Experiment*Species interaction			1	
transfluthrin treated fabric * An. arabiensis	d fabric * An. arabiensis			
transfluthrin treated fabric * An. gambiae	hrin treated fabric * <i>An. gambiae</i> 0.36 (0.08-1.70)		0.201	
Mosquitoes recaptured with <i>UV light</i> trap out of all mosquitoes <u>released</u>				
Experiment				
Untreated eave fabric (experiment 4)	15.2 (95% CI 12.2-18.9)	1		
transfluthrin treated fabric (experiment 6)	11.4 (95% CI 8.77-14.8)	0.72 (0.49-1.06)	0.098	
Species				
An. arabiensis		1		
An. gambiae		1.30 (0.89-1.91)	0.178	
Experiment*Species interaction				
transfluthrin treated fabric * An. gambiae		was not retained in final model	0.995	
*based on statistical model				

The percentages of mosquitoes trapped with individual CDC UV light traps out of all released mosquitoes were similar (p=0.342) when set in the presence of another light trap (10%; 95% CI 7-14%) and when set in the presence of an odour-baited Suna trap (12.5%; 95% CI 9.1-16.8%; Figure 4C). Individual odour-baited traps caught around 3 times more mosquitoes (OR 3.2 (95% CI 1.8-5.7), p < 0.001) when set up in the presence of another odour-baited trap (27.7% (95% CI 21.6-34.6%) than when set up in presence of a CDC UV light trap (10.5% (95% CI 6.8-15.9%); Figure 4B).

Investigating the effect of transfluthrin on trapping (experiments 4-6; presence of eave wrap)

Transfluthrin pre-exposed An. arabiensis

Pre-exposure was neither associated with the ratio with which the mosquitoes were collected with the odour-baited Suna trap (Figure 3, comparing experiments 4 and 5; Table 2) nor with the overall recapture rate out of all released mosquitoes (Figure 4, comparing experiments 4 and 5; Table 2).

Continuous transfluthrin exposure of An. arabiensis and An. gambiae s.s.

Continuous exposure to transfluthrin-treated eave fabric in experiment 6 was associated with a reduction in proportional catches of host-seeking *Anopheles* in the Suna trap (out of all trapped; Figure 3, Table 2). The analysis did not suggest a statistically significant main effect of a species-specific trapping efficacy, but the interaction term between the treatment and mosquito species implies that *An. gambiae s.s.* might be more affected by the transfluthrin than *An. arabiensis* and consequently less likely to be trapped by the odour-baited trap in the presence of transfluthrin (OR 0.39; Figure 3 and 4, Table 3) even though this did not reach statistical significance.

A reduction in proportional catches of host-seeking *Anopheles* in the Suna trap was also observed when the efficiency in recapturing mosquitoes out of all released was examined. Analysing the outcome for the two traps separately revealed that the odds of trapping a mosquito in the Suna trap (out of all released) when constantly exposed to transfluthrin was over 3 times lower (OR 0.29) than when unexposed (Table 3). Again, a species-specific interaction suggests that the trapping efficiency for *An. gambiae s.s.* was more affected than for *An. arabiensis* (OR 0.36; Table 3). The trapping efficiency of the CDC UV light trap was more or less consistent under most experimental conditions (Figure 4) with a trend of slightly reduced trapping (OR 0.79) with the CDC UV light trap in presence of a Suna trap and the spatial repellent. This did not reach statistical significance (p = 0.098) and the minimal impact left this trap still superior to the odour-baited trap. The analysis suggests that *An. gambiae s.s.* might be slightly more responsive to UV light traps than *An. arabiensis* (OR 1.30) in general, irrespective of the presence or absence of transfluthrin; however, again this did not reach statistical significance.

Discussion

In the present study, the trapping efficiency of an odour-baited Suna trap was greatly impacted by the presence of transfluthrin in the semi-field system. In the absence of the repellent, one odour-baited trap caught more than a quarter of all the mosquitoes released. This reduced to just below 5% of the total released when the repellent was present. Momentary exposure of experimental mosquitoes to the repellent prior to release did not affect the trapping efficiency of the odour-baited trap. Interestingly, the trapping efficiency of an odour-baited trap was also influenced by the presence of either a second odour-baited trap in the system or the presence of a light trap. When two odour-baited traps were in a system, their combined trapping efficiency was highest in this particular experiment as they were able to recapture more than half of the total released mosquitoes (Hiscox *et al.*, 2014), while when set up in combination with a UV light trap, the trapping efficiency of an odour-baited in the system. In comparison, earlier experiments showed that a human being was able to trap more than 70% of the mosquitoes released in the semi field system as seen in previous work (unpublished data – Chapter 2).

The exact reasons why there was a disparity between recapture rate of the odour-baited trap in the presence of a second odour-baited trap compared to when in the presence of a light trap are unknown. One reason may be that the increased synthetic lure concentration in the semi-field system, working with CO_2 may have increased flight activation and attraction of mosquitoes to the traps, similar to host-seeking events (Takken *et al.*, 1997a; Dekker *et al.*, 2001; Dekker *et al.*, 2005; Dekker *et al.*, 2011). With this assumption, it would also be expected that odour-baited traps would have captured more mosquitoes than the light traps due to the presence of the synthetic lure in addition to CO_2 (Qiu *et al.*, 2007; Cardé *et al.*, 2010b; Cardé *et al.*, 2010a) and in the absence of any external interference. In the present study, the trapping efficiency of the odour-baited trap was greater than that of the CDC UV light trap except in the preliminary experiment 3; the calibration experiment done in the absence of any eave fabric, when the CDC UV light catches were slightly higher. This exception was unexpected and remains unexplained.

Trapping efficiency of the CDC UV light trap was interestingly not affected by the presence of the repellent. In fact, the proportion of recaptured mosquitoes remained relatively constant (10-12%) in the light trap; both in the absence and in the presence of the repellent. However, the proportion of mosquitoes trapped by the light traps was low relative to the total number of mosquitoes released. The reason for this low recapture rate could have been that light, even when supplemented with CO_2 , may not provide clear directional and specific cues for host-seeking mosquitoes that make it a suitable attractive cue compared to the combination of CO_2 and odour (Platt *et al.*, 1957; Stark *et al.*, 1982; Takken *et al.*, 2016; Hawkes *et al.*, 2016; Costa-Neta *et al.*, 2017; Ponlawat *et al.*, 2017a; Sheppard *et al.*, 2017) exemplified by how
exposure to white light or moonlight significantly affects the host-seeking behaviour of malaria mosquitoes (Bidlingmayer, 1964; Sheppard *et al.*, 2017).

Use of LED lamps that emitted blue light of 470 nm wavelength or green light of 520 nm wavelength was shown to catch high numbers of *Anopheles* mosquitoes in field assays compared to incandescent light (Costa-Neta *et al.*, 2017). Preference for UV (including near ultraviolet UV-A black light of 315-400 nm) and LED light over the standard CDC incandescent light by *Anopheles* mosquitoes was seen in a different study and on this basis UV light was selected over incandescent light in this study (Li *et al.*, 2015a; Mwanga *et al.*, 2019).

Species-specific behavioural responses to olfactory cues have been shown in previous studies and were attributed to differences in behaviour possibly due to divergent chemosensory receptor where one species may detect a repellent (e.g. Aedes and Culex to DEET) while another does not (Anopheles to DEET)(Ghaninia et al., 2019; Afify et al., 2020). During the development of the MB5 synthetic lure, attractive chemicals were assessed based on response to An. gambiae s.s. and it was only in later studies that responses by other malaria vectors to the synthetic blend were determined (Mukabana et al., 2002; Olanga et al., 2010; Verhulst et al., 2011; Mukabana et al., 2012a; Hiscox et al., 2014; Menger et al., 2014; Menger et al., 2015; Homan et al., 2016; Menger et al., 2016; Mburu et al., 2017; van de Straat et al., 2019). The present study was able to show both similar and dissimilar responses of the two sibling species; An. arabiensis and An. gambiae s.s. Proportional catches of the sibling species in the odourbaited trap in the absence of the repellent were comparable. The repellent, however, seemed to adversely affect An. gambiae s.s. as a lower proportion was caught in the odour-baited trap when transfluthrin was present compared to An. arabiensis under the same experimental conditions. On the other hand, An. gambiae s.s. seemed more responsive to the UV light trap in general compared to An. arabiensis. The sibling species studied have previously been shown to not only have different host-preferences but to also respond differently to various cues, partially explained by the structural and genetic differences in their odorant-binding proteins (van den Broek et al., 1999; Li et al., 2004; Li et al., 2005; Lorenz et al., 2013; Ghaninia et al., 2019).

The ideal mode of action for spatial repellents in mosquito-borne disease control has been highlighted as the ability to mask human host odours or at least induce an avoidance behaviour or feeding inhibition, therefore limiting the host-vector interaction (Schreck *et al.*, 1970; Achee *et al.*, 2009; Achee *et al.*, 2012; Ray, 2015). The repellent effect in general points to some influence in the perception and location of potential hosts through interference in odour-detection (Hao *et al.*, 2008; Turner *et al.*, 2011).

In our study, the reduction in proportional catches of mosquitoes in odour-baited traps in the presence of the repellent may be explained by possible interference with the perception of the lure. As all mosquitoes

were released less than 1.5 m from the point of application of repellent and at least 3 m from either trap, the assumption that the repellent plume influenced the mosquito perception of odours is plausible. The use of an odour-baited trap, at least in the relatively close vicinity of the repellent, is therefore unsuitable in the push-pull approach and further placement of these traps could be considered outside the area of protection of the repellent. In Chapter 2, the positioning of odour traps at 5 m was not protective against outdoor-biting *An. arabiensis* (Chapter 2, this thesis). Testing varied distances such as 7.5 m, 10 m and even 15 m could provide information on which would be the most beneficial positioning. While exploration of distance from the repellent plume could be one consideration, positioning of the trap in relation to other references such as breeding habitats (Okumu *et al.*, 2010a).

Previous studies have shown that repellents such as geraniol, citral, eugenol and anisaldehyde affect *Ae. albopictus* mosquitoes after prolonged exposure (48 h) such that their host-seeking ability is greatly affected leading to reduced movement towards hosts and probing (Hao *et al.*, 2008). In natural conditions, mosquitoes would normally fly out of a repellent plume to better reorient with host cues before attempting another approach and may spend just enough time in the plume to affect their immediate but not their subsequent host-seeking behaviour (Schreck *et al.*, 1970; Barnard, 2005; Klun *et al.*, 2006; Grieco *et al.*, 2007; Moore *et al.*, 2007; Hao *et al.*, 2008; Bohbot *et al.*, 2010).

While the trapping efficiency of the UV light trap was unaffected by the repellent and was able to perform trap better than an odour baited trap in the present study, it would be highly unlikely to replicate such artificial conditions in the field especially with regards to mosquito proximity to the treated house prior to picking up the host cues and how the airflow would influence the availability of the plume and host cues (Achee *et al.*, 2012; Kawada *et al.*, 2014; Menger *et al.*, 2015). It would be important to note that the CDC light trap was originally intended for indoor use (Garrett-Jones *et al.*, 1975), though recent studies have explored its utilization outdoors (Sriwichai *et al.*, 2015; Degefa *et al.*, 2017; Degefa *et al.*, 2020). Future studies could substantiate the differences seen in natural conditions and provide evidence on the suitability or otherwise of CDC UV light trap as a pull component. As a way forward, a more attractive pull trap, preferably an odour-baited one supplemented with additional host cues such as heat and humidity (Hawkes *et al.*, 2017) could be explored preferably in natural conditions. The reason for this is that the trapping efficiency of the odour-trap remains superior to that of the light trap in the absence of a repellent (Hiscox *et al.*, 2014; Mburu *et al.*, 2019). By adjusting the position of an improved odour-trap trap proximal to the repellent plume such that the mosquito encounters the trap prior to the repellent, the push-pull set up could be further improved (Okumu *et al.*, 2010a).

Conclusion

This study showed that MB5-baited traps are unsuitable in close vicinity to a spatial repellent for a push pull set up and that UV light traps might be more efficient and potentially easier to manage. However, since both traps types were not highly efficient even in the absence of a human, there is urgent need to develop more attractive traps by increasing the attractive cues (combining odours, warmth and humidity) that can be combined with spatial repellents in a push pull situation than currently available.

Acknowledgements

We thank Bruno Otieno and Mgeni Mohamed Tambwe for providing technical support in all the experiments. We also acknowledge Elisha Obudho and David Alila for providing all the mosquitoes needed for this study.

Less is more: Repellent-treated fabric strips as a substitute for full screening of open eave gaps for indoor and outdoor protection from malaria mosquitoes in Kenya

Margaret M. Njoroge, Alexandra Hiscox, Joop J.A. van Loon, Willem Takken and Ulrike Fillinger

To be submitted in a modified form

Abstract

Providing protection from malaria mosquitoes, both indoors and outdoors, is crucial to curbing malaria parasite transmission. Screening of house entry points, especially with incorporated insecticides, confers significant protection but remains a costly and labour-intensive application. Use of spatial repellents has shown promise in creating areas of protection against mosquito bites in the peri-domestic area of a household. This study aimed at comparing the protection provided by transfluthrin-treated and untreated eave screens and at determining if an incomplete transfluthrin-treated eave strip would be a suitable replacement for a full screen. Human landing catches were implemented indoors and outdoors for separate experiments, with insectary-reared *Anopheles arabiensis* mosquitoes under controlled semi-field conditions.

The odds of a female mosquito finding a human volunteer indoors and attempting to bite was similar whether the eaves were completely open, or when there was an untreated fabric strip fixed around the eaves. The application of a transfluthrin-treated fabric strip achieved the same protection indoors (OR 0.07, 95% CI 0.04-0.13) as the treated full screen and was 4 times lower than the odds of receiving a bite in the presence of an untreated full screen (OR 0.30, 95% CI 0.20-0.47). The impact of transfluthrin treatment on outdoor-biting was correlated with evening temperatures during the experiments. At comparatively low evening temperatures, a transfluthrin-treated full screen provided moderate and variable protection from bites (OR 0.62, 95% CI 0.37-1.03), whilst at higher evening temperatures the odds of receiving a bite outdoors was over 4 times lower in the presence of transfluthrin, either on a full screen (OR 0.22 95% 0.12-0.38) or on a fabric strip (OR 0.25, 95% 0.15-0.42) than when no treatment was present (open eave as reference).

The findings suggest that transfluthrin-treated fabric strips can provide a substitute for complete eave screens. They are a simple, easy to handle tool for protecting people from malaria mosquito bites indoors and potentially around the house in climatic areas where evening and night-time temperatures are relatively high.

Keywords:

Transfluthrin, eave screens, eave strips, vector control, spatial repellent, human landing catches

Introduction

Many malaria endemic areas in sub-Saharan Africa are rural, where housing structure pre-disposes people to mosquito bites; either because of open gaps along the eaves, uneven walls and unscreened windows and doors. House structures often necessitate evening activities, cooking for example, to be conducted outdoors or in structures near the main house that do not offer any protection from mosquito bites (Ghebreyesus *et al.*, 2000; Huho *et al.*, 2013; Wanzirah *et al.*, 2015; Kaindoa *et al.*, 2018).

Eave gaps provide ventilation indoors and sealing them off, while slightly protective against mosquitoes, can lead to reduced air flow and potentially increased indoor temperatures unless proper house adjustments are done to remedy this (Lindsay *et al.*, 2003; Jatta *et al.*, 2018). Improved housing structure has been shown to significantly reduce house-entry of adult mosquitoes and malaria infections (Lindsay *et al.*, 2002; Lindsay *et al.*, 2003; Kirby *et al.*, 2009; Tusting *et al.*, 2017; Corrêa *et al.*, 2019). Screening of open eave gaps in combination with screening of doors and windows is recommended for effective reduction of mosquito house entry(Kirby *et al.*, 2009).Despite the advantages of eave screening, a major challenge in this approach is the availability of low-cost materials, as well as the difficulty in properly fixing barriers in existing structures which may have uneven surfaces (Menger *et al.*, 2016; Ng'ang'a *et al.*, 2019). Furthermore, screening the house may reduce adherence to the use of long-lasting insecticide-treated bed nets (LLINs), due to perceived reductions in biting pressure (Ng'ang'a *et al.*, 2019) which could lessen the protection received by occupants.

In situations where house-improvement is not immediately achievable for varied reasons, and where LLINs provide incomplete protection against malaria infection, alternate means of vector control should be developed to provide protection to persons when they are in the periphery of their homes (outdoors), or indoors but not yet under protective bed nets (Mathania *et al.*, 2016).Evening and outdoor mosquito biting have been cited as contributors to the rising importance of residual malaria transmission that remains unaddressed (Durnez *et al.*, 2013; Killeen, 2014; Wamae *et al.*, 2015; Sherrard-Smith *et al.*, 2019).

Spatial repellents have received increasing attention in recent years as complementary control tools against adult malaria mosquitoes (Hao *et al.*, 2008; Achee *et al.*, 2012a; Ogoma *et al.*, 2012; Revay *et al.*, 2013; Andres *et al.*, 2015; Maia *et al.*, 2016; Norris *et al.*, 2017). Unlike toxicants that result in mortality of mosquitoes or irritants that result in agitation, spatial repellents drive mosquitoes away from a treated space, thereby being proposed as ideal candidates for outdoor application(Lynch *et al.*, 2016; Norris *et al.*, 2017). In addition, pyrethroid spatial repellents have been shown to induce deterrence, even in

mosquitoes which are resistant to this group of insecticides (Agossa *et al.*, 2015; Lynch *et al.*, 2016). As a deterrent mode of action may not lead to complete toxic effects, the selection pressure exerted by spatial repellents might be lower than that exerted by toxicants which end up selecting resistant mosquitoes in the population by killing of the more susceptible ones(Birget *et al.*, 2015; Lynch *et al.*, 2016). There is, however, the unfortunate possibility of selection of multiple resistance genes which could occur gradually with constant use of deterrent insecticides leading to increasing low-level resistant mechanisms (Gressel, 2011; Guedes *et al.*, 2017). An advantage of spatial repellents is the possibility to exert repellency on a wide-range of mosquito species, therefore potentially disrupting not just the transmission of malaria, but other mosquito-borne diseases such as dengue, Chikungunya and Zika among others (Achee *et al.*, 2012b; Sathantriphop *et al.*, 2014). Additionally, spatial repellents can be highly effective against biting by nuisance mosquitoes.

One such candidate spatial repellent is transfluthrin, which is a fast-acting, low persistence, fifteen-carbon pyrethroid insecticide. It has been included in mosquito repellent and killing products such as mosquito coils and vaporizers(Morton *et al.*, 1947; Schreck *et al.*, 1970; Ramesh *et al.*, 2001; Nazimek *et al.*, 2011; Ogoma *et al.*, 2014b; Andres *et al.*, 2015).Release of transfluthrin on commercial products relies on an external source of energy or combustion and has many limitations: coils produce smoke and need to be actively managed for optimal protection(Hill *et al.*, 2014), electrical emanators are not affordable for most rural low-income communities and require a power socket in the area to be protected(Pates *et al.*, 2002). Non-powered, low maintenance passive release options would make this intervention more accessible for large-scale use (Pates *et al.*, 2002; Ogoma *et al.*, 2012; Masalu *et al.*, 2017).

Inclusion of the repellent on eave screening fabric for passive release could increase the protection conferred indoors by existing tools and could potentially allow peri-domestic release of the repellent thereby also providing the much-needed bite protection outdoors, even against pyrethroid-resistant malaria vectors (Kawada *et al.*, 2014; Andres *et al.*, 2015; Govella *et al.*, 2015; Bowman *et al.*, 2018; Deletre *et al.*, 2019). Recent work done in Tanzania has explored the use of insecticide-treated strips that are secured on traditional houses to achieve a protective effect against malaria mosquitoes instead of the use of full eave screening (Ogoma *et al.*, 2012; Govella *et al.*, 2015). The idea of using a repellent-treated fabric that does not fully cover the eave gaps offers possible cost and logistical advantages as there would probably be less material used and less precise fitting necessary as compared to the treated or untreated-eave screen (Kirby *et al.*, 2009; Ng'ang'a *et al.*, 2019).

The present study aimed at comparing the protection provided by both treated and untreated full screening, and incomplete screening with treated and untreated eave strips against bites from insectary-reared *Anopheles arabiensis*(Mbita strain, Kenya) under controlled semi-field conditions. Human landing rates, both indoors and outdoors, were measured as a basis for determining if a transfluthrin-treated eave strip would be a suitable replacement for a full eave screen (treated or untreated).

Materials and methods

All experiments were conducted under semi-field conditions at the International Centre of Insect Physiology and Ecology at the Thomas Odhiambo Campus (icipe-TOC) in Mbita, on the shores of Lake Victoria in Homabay County, western Kenya (0°26'06.19"S, 34°12'53.13"E; altitude 1,137 m).

Semi-field systems

Two mesh-screened greenhouses (Amiran Kenya Ltd, Nairobi, Kenya) measuring 27 m long, 11 m wide and 4.3 m at the highest point were used for experiments. These systems were made of steel-structured frames with a SolarigTM covered roofs and 17-mesh netting (17 apertures per every linear inch of mesh) on all sides to ensure adequate ventilation (Figure 1). The floor of the semi-field systems was covered with up to 30 inches of sand and was kept clear of any vegetation. The sandy floors were watered daily prior to experiments to ensure that the relative humidity remained above 70%. Temperatures inside the semi-field system varied during the months the experiments were implemented (September 2018 to March 2019) between a minimum of 18 °C at night and a maximum of 50 °C during the day. Data loggers (Tinytag View 2 Data Logger, Gemini data loggers, UK) were placed in both semi-field systems and temperature readings were taken every 30 minutes throughout. Each semi-field system contained a makeshift hut made of angle iron frames, ply wood walls and grass-thatched gable roofs with open eave gaps to mimic a traditional house in western Kenya. Each traditional house measured 6.5 m by 3.5 m and 2.5 m at the highest point (Figure 1). Between the roof and the top of the wall, all round the house was a 0.1 m wide eave gap that was representative of gaps left open in traditional western Kenyan houses. The doors and windows of the experimental huts were fully screened. Human landing catches (HLC) were conducted either outdoors 2.5 m away from the experimental hut or inside at the centre of the hut.



Figure 1. One of the semi-field systems where the bioassays took place. Inside each semi-field system, that measures 27 m by 11 m, there was an experimental house made of plywood with a grass thatched roof and the frames made of angle iron to mimic a traditional western Kenya house. The experimental huts measured 6.5 m by 3.5 m.

Mosquitoes

Host-seeking female *Anopheles arabiensis* Mbita strain were obtained from the icipe-TOC mosquito insectary and were used for all bioassays. Rearing of mosquitoes in the insectary followed a standard procedure (Mbare *et al.*, 2013) where they were reared under ambient conditions. Female mosquitoes, between 3 and 5 days old post-emergence that had never fed on blood were selected for use in these experiments. Selection was done by holding a hand close to the outside of the cage and picking responding females out of the cage with a mouth aspirator. In each experimental night, 160host-seeking female mosquitoes were used per semi-field system. Selected mosquitoes were starved of water and glucose for 3-5 h prior to release in the semi-field systems. In each semi-field system, 40 mosquitoes were released from cups placed in all 4 corners. Mosquitoes were dusted using a distinct fluorescent dye to distinguish them according to the four sites of release(Verhulst *et al.*, 2013).

Insecticide susceptibility

World Health Organization (WHO) insecticide susceptibility cone tests were conducted on the insectaryreared *An. arabiensis* females (WHO, 2013). Four cones were used to test susceptibility of mosquitoes to 0.05% deltamethrin by exposing mosquitoes to treated papers for one hour and then monitoring them for 24 h. Knock down and mortality were compared to a control group that was exposed to untreated paper in parallel. In all susceptibility tests, between 20 and 25 mosquitoes were used in each cone with four cones being used to expose mosquitoes to test paper and two cones to expose mosquitoes to the castor oiltreated control paper (WHO, 2013).

Preparation of eave screens and strips

For full eave screens, burlap material from local markets was cut into bands measuring 21m by 0.12m to ensure full coverage of the eave gaps when fixed. For incomplete screening, narrow fabric strips were cut at the same lengths but only at 0.05m wide (Figure 2).



Figure 2: Eave screen secured on the eave gap of the experimental hut using aluminium wires to ensure equal complete coverage of the eave gap by the eave fabric leaving no gap above or below.

Emulsified transfluthrin from Bayer Global (Leverkusen, Germany) supplied at a concentration of 0.2 g/ml was used as a spatial repellent. As transfluthrin is insoluble in water (WHO, 2006; Ogoma *et al.*, 2012), the emulsified concentrate simplified the preparation of a solution and the impregnation of fabric (Ogoma *et al.*, 2012).

For treatment, a transfluthrin solution was prepared with approximately 1300 ml of water for each hessian screen and 700 ml for each hessian strip to ensure the fabrics are fully wetted without any dripping of the solution. Emulsified transfluthrin was added to the water to achieve 2.5 g active ingredient per m^2 of fabric. Eave screens and strips were kneaded into the transfluthrin solution until visibly saturated (Ogoma *et al.*, 2012). All fabric was dried in a non-experimental screened hut overnight immediately after impregnation, then rolled up carefully and stored, wrapped in aluminium foil, in a dark room at room temperature. Untreated screens or strips were soaked in plain water and then dried, and used in the same way as the treated fabric but stored at a different location.

During experiments, fabric was retrieved from the cold room and put up on the experimental hut on day one in the evening then removed the next morning before being stored again in the cold room. The treated fabric was unwrapped and put along the eaves of the experimental hut every evening at 17.30 h. Care was taken to avoid contact with the walls of the hut when placing the fabric. At the end of every experimental night, the fabric was removed in the morning, rolled up, covered in aluminium and stored in a dark room at room temperature till evening when it was placed again. The same fabric was used for 16 experimental nights – the duration of one full experiment with a fresh set of fabric being used for every subsequent experiment. Untreated fabric was handled with a clean set of gloves to avoid cross contamination. For the application of eave screens, care was taken to ensure that the entire eave gap was covered each evening by using wires to secure the fabric over it (Figure 2). For the incomplete screen, the fabric strips were positioned at the centre of the eave gap, ensuring an equal amount of space left open both above and below the fabric (Figure 3).

Experimental procedure

Five experimental treatments consisting of: (i) no eave fabric (open eaves), (ii) untreated full eave screen, (iii) untreated strip, (iv) transfluthrin-treated full eave screen, and (v) transfluthrin-treated strip, were tested in six blocks of experiments as outlined in Table 1. Due to availability of two screen-house systems, two experiments were run at the same time (experimental block).Human landing catches (HLC) were conducted for four hours between 19:00 h and 23:00 h in all experiments for 16 nights.



Figure 3. Eave strip secured on the eave gap of the experimental hut using aluminium wires to ensure equal gap (2.5cm) left above and below the eave fabric.

Treatments were crossed-over between semi-field systems after every four days and a resting period of three days in-between to allow for aeration of experimental residues (Ogoma *et al.*, 2014a).

Four male volunteers (aged between 18 and 50 yr) were rotated between the semi-field systems and experimental nights to correct for human bias. The impact of the treatments was estimated separately for outdoor and indoor biting. For the simulation of outdoor biting, the volunteers sat on a chair situated 2.5 m away from the experimental hut, approximately in the middle of the semi-field system (semi-field system as shown in Chapter 2). For estimates of indoor-biting, the volunteers sat in the middle of the experimental hut. The volunteers were screened weekly for malaria parasite infections to prevent any circulation of infected insects in the semi-field systems and were exempted from participation till a negative result was confirmed. In preparation for the experiments, each volunteer cleaned their feet with odourless soap up to the knee and took position on the chair. Host-seeking *An. arabiensis* females were manually aspirated as soon as they landed on their lower legs of the volunteers. The mosquitoes were transferred to collection cups labelled with the hour of collection and semi-field system identification letter. Protective jackets to cover heads and arms were provided while torches were used for visualization

when aspirating. Volunteers conducted HLC while shoeless with landings on feet included in the experimental outcome.

Table 1: Summary of all experimental procedures implemented as blocks of two test treatments in parallel in two semi-field systems. Indoor and outdoor human landing catches were done in independent experiments.

Experimental blocks	Location of human landing catches	Treatment in semi-field system A*	Treatment in semi-field system B*
1	Indoors	Open eave	Untreated incomplete eave strip
2	Indoors	Untreated full eave screen	Transfluthrin-treated full eave screen
3	Indoors	Transfluthrin-treated full eave screen	Untreated incomplete eave strip
4	Outdoors	Open eave	Untreated incomplete eave strip
5	Outdoors	Untreated full eave screen	Transfluthrin-treated full eave screen
6	Outdoors	Transfluthrin-treated full eave screen	Untreated incomplete eave strip

*All treatments were switched between semi-field systems every four days ensuring that in each experimental block, the treatments were equally applied in system A and B.

Data analysis

All analyses were carried out using R Studio statistical software from R core group version i386 3.5.1 (R Core Team, 2018). Associations between the proportion of mosquitoes landing on volunteers (number of mosquitoes collected out of all mosquitoes released) and test treatments were analysed using generalized estimating equations (GEE) fitted with a binomial distribution with logit link function. An exchangeable correlation matrix was assumed. The unique ID of every experimental night was included in the model as repeated measure. The experimental test and the volunteer ID were included as the fixed factors in the models. The experiment without any eave fabric in the system (open eave) served as reference. The model generated odd ratios (OR) and their associated confidence intervals (CI) which are reported. Mean proportions and their 95% CIs were estimated based on the model by transforming the log odds (logit) of

the outcome to the odds scale and from the odds scale to the probability scale. The denominator in all experiments was the total number of females released per experimental night less the mortalities prior to release. Results from experimental nights where mortality in the release cups exceeded 10% were excluded from the analysis. The point of mosquito release had no significant association with the outcome and was removed from the final models. The possible correlation between the mean air temperature (in °C) during the four-hour mosquito collections duration of every experiment and the human biting rate was explored for transfluthrin containing experiments and for non-insecticidal experiments using the Pearson Correlation.

Ethical considerations

This study was approved by the Kenya Medical Research Institutes Scientific and Ethics Review committee (KEMRI-SERU), protocol number NON-KEMRI 546.

Results

Insecticide susceptibility test

In total, 88 *An. arabiensis* were exposed to 0.05% deltamethrin and 48 exposed to the untreated control (castor oil) in the WHO cone assay. Twenty-four hours after a 1 h exposure, a mortality of 93.2% was found, which was corrected to 91.15% according to WHO guidelines. This suggests that the insectary-reared mosquitoes used in consequent experiments were not fully susceptible to pyrethroid insecticides.

Indoor impact of treatments

In the absence of any fabric on the open eave gaps of the experimental huts, on average 45% (95% CI 38-52%) of all released mosquitoes were collected whilst seeking to bite the human volunteer indoors. This was similar when an untreated strip was placed at the eave gaps (Table 2, Figure 4). An untreated full screen had a highly significant impact on the proportion of mosquitoes seeking out the volunteer indoors (OR 0.3; Table 2); it prevented around half the bites a volunteer would have received in the absence of a screen (Table 2, Figure 4).

Treating the full screen with transfluthrin had a significant added benefit in reducing the proportion of mosquitoes seeking the host indoors. The odds of a mosquito landing on a human volunteer was 97% lower (OR 0.03) than it was when the screen was untreated based on the data from the experimental block no. 2 (Table 2, Figure 4).

Chapter 4

 Table 2: Association between proportion of mosquitoes landing on human volunteer and test treatments.

	Estimated* mean proportion of released <i>An. arabiensis</i> biting (95% confidence interval)	Odds Ratio (95% confidence interval)	p-value
	INDOORS		
Treatment			
open eave	0.45 (0.38-0.52)	1	
full untreated eave screen	0.20 (0.15-0.26)	0.30 (0.20-0.47)	< 0.001
untreated eave strip	0.43 (0.36-0.50)	0.90 (0.66-1.24)	0.520
treated full eave screen			< 0.001
(block 2)	0.03 (0.01-0.06)	0.03 (0.02-0.07)	-0.001
(block 3)	0.06 (0.04.0.10)	0.08 (0.04.0.16)	< 0.001
(DIOCK 5)	0.05(0.04-0.10)	0.03 (0.04 - 0.10) 0.07 (0.04 - 0.13)	< 0.001
Voluntoor	0.03 (0.03-0.07)	0.07 (0.04-0.13)	-0.001
volunteer		1	
no. 1	-	1 06 (0 73 1 55)	0.720
no. 2	-	1.00(0.73 - 1.53) 1.03(0.71 - 1.52)	0.720
no. 5	-	0.97(0.53-1.40)	0.020
110. 4		0.57 (0.55-1.40)	0.910
Transformed	OUTDOOKS		
I reatment	0.54 (0.45.0.(2))		
open eave	0.54 (0.45-0.63)	1	0.054
full untreated eave screen	0.50 (0.41-0.60)	0.85 (0.53-1.41)	0.354
Untreated eave strip	0.57 (0.48-0.65)	1.14 (0.86-1.52)	0.547
treated full eave screen	0.42 (0.22.0.51)	0.62(0.27,1.02)	0.064
treated full eave screen	0.42 (0.55-0.51)	0.02 (0.57-1.05)	0.004
(block 6)	0.19 (0.13-0.28)	0.22 (0.12-0.38)	< 0.001
treated eave strip	0.21 (0.14-0.30)	0.25 (0.15-0.42)	< 0.001
Volunteer			
no. 1	-	1	
no. 2	-	1.33 (0.83-2.13)	0.230
no. 3	-	0.90 (0.65-1.26)	0.592
no. 4		0.88 (0.68-1.13)	0.336

*based on statistical model





Treated full screens were also tested in experimental block no. 3 and the odds of receiving a bite, even though still highly protective, was slightly higher (OR 0.08, Table 2) than in block no. 2. One difference observed between these two experimental blocks was a drop in the mean air temperature by around 1 °C (Figure 4) during the duration of the experiments. When comparing the impact of treated eave screens with treated fabric strips in experimental block no. 3, both were equally effective in reducing the odds of a mosquito approaching the volunteer indoors (Table 2).

Outdoor impact of treatments

The proportion of host-seeking mosquitoes recovered outdoors through human landing catches was with an average of 54% (95% CI 45-63%) around 10% higher than what was recovered indoors in the absence of any treatments.

No fabric (open eaves), untreated full screen and untreated fabric strips at the eaves led to the collection of similar proportions of released mosquitoes outdoors when landing on the human volunteers (Table 2). More clearly than for the indoor environment, the outdoor protection from transfluthrin as a spatial repellent appeared to be temperature dependent. In the experimental block no. 5 where the experiment with untreated full screen was run in parallel to the experiment with treated full screen, only a slight, borderline significant, reduction in human landing was observed, from an average of 54% of released mosquitoes landing on a volunteer in the absence of transfluthrin (open eave) to an average of 42% in the presence of transfluthrin (p = 0.064; Table 2). In contrast, when the treated full screen experiment was repeated in experimental block no. 6, only around 19% of the released mosquitoes landed on the human volunteer, presenting a statistically highly significant reduction (p < 0.001) when compared to no intervention and when compared to untreated full screen (Table 2, Figure 4). Mean temperatures between the two experimental blocks differed by around 1 °C (Figure 4). Pooling all data for experiments with transfluthrin and for experiments without transfluthrin in the system and performing a descriptive analysis found a negligible and statistically non-significant positive correlation between biting rates and air temperatures during the experiments (r=0.186, p=0.099) in the absence of transfluthrin. In contrast, in the presence of transfluthrin either on screens or strips, there was a negative association between landing rate and evening temperature (r=-0.529, p<0.001) with higher temperatures resulting in lower landing rates hence higher protection (Figure 5).

The impact of a treated screen and a treated strip was, as observed indoors, similar outdoors in reducing the proportion of host-seeking mosquitoes landing on human volunteers at the test conditions (mean temperature of 23.7°C; Figure 4). It must however be expected that a treated strip would equally not

provide large reductions if evening/night-time temperatures are low as seen for treated full screen experimental in block 5.



Mean air temperature in °C during 4 hours experiment

Figure 5. Scatter plot and trendlines indicating average air temperatures during the experimental runs with trendlines indicating how proportional mosquito recapture relate to temperature both in the presence and absence of transfluthrin.

Discussion

A strip of eave fabric treated with transfluthrin as a spatial repellent provided similar protection against both indoor and outdoor host-seeking mosquitoes compared with the treated screen, showing the plausibility of using a treated eave strip as a substitute to the treated eave screen. Untreated full screening was protective indoors but not outdoors, likely due to the physical barrier presented that prevented houseentry of mosquitoes. Treating a full eave screen with a spatial repellent added significant additional protection indoors compared to the reference group (open eaves) as well as the untreated eave screen. Treatment of eave screening could be envisaged in a scenario where screens are fitted to a house that has

irregular walls and where it is not possible to seal all gaps and entry points. A repellent-treated full eave screen and a repellent-treated strip provided up to 60% protection to a human seated outdoors, at least within an area of around 3 m from the house outside, which we refer to as the peri domestic space.

Use of transfluthrin seemed to be greatly affected by the ambient temperature as lower temperatures during this experiment were associated with lower protection conferred outdoors by the treated full eave screen. It is assumed that the same effect would be observed for the treated strip, but during the experiments with the treated strip, mean temperatures were higher. Interestingly though, indoor protection was not affected by temperature, possibly due to close contact with house-entering mosquitoes, or already higher concentration of the repellent contained indoors that continued to confer protection even when temperatures reduced. This could also be attributed to higher temperatures indoors than those experienced outdoors - though it is important to note that the position of the repellent-treated eave screen was the same during indoor and outdoor HLCs. The impact of indoor temperature was however not tested as all temperature measurements were taken outdoors. Further experiments with treated full eave screening and treated strips, with temperature recordings taken indoors and outdoors across seasons could provide more information about the critical temperature threshold for transfluthrin used as a spatial repellent(Glunt et al., 2014; De Keyser et al., 2017; Ogoma et al., 2017). From the initial studies in the absence of any treatment, indoor-biting rates of An. arabiensis were lower than those outdoors. The reason for this could partly be that following release of mosquitoes outdoors, they had easier access to the persons outdoors than to volunteers indoors who were partially protected by virtue of being inside a house which was a physical barrier. The mosquitoes gained entry through the open eaves as both the doors and windows were screened. The preferential outdoor-biting behaviour of An. arabiensis may also have contributed to the higher outdoor catches compared to indoors. While the presence of screens; both treated and untreated, significantly reduced the number of mosquitoes indoors, there were still mosquitoes that managed to gain entry, possibly through insignificant gaps such as the grass-thatched roof. This substantiates the need to continue using current vector control tools even as supplementary tools are developed.

The role that house structure plays in either aiding or preventing mosquitoes in gaining entry has been elaborated in many studies (Ghebreyesus *et al.*, 2000; Lindsay *et al.*, 2002; Njie *et al.*, 2009; Wanzirah *et al.*, 2015). This has led to several suggestions of ways to improve house designs in order to reduce the vector-host contact towards malaria control (Tusting *et al.*, 2017). One of the major improvements that have been proposed is screening of the eave gap that is usually left open in rural houses for both ventilation and structural purposes (Ghebreyesus *et al.*, 2000; Lindsay *et al.*, 2002). While in the long run the improvement of house design would be desirable for all, it may not be immediately achievable due to

the financial handicap that plagues many people in malaria endemic areas (Tusting *et al.*, 2017). Development of tools such as the ones investigated in the current study that work on current house designs should be prioritized to protect people from receiving potentially infective bites. The application of treated eave strips could be a simple modification to houses, which does not require complex technical knowledge, or permanent changes to the house structure.

In line with previous studies, the current study was able to confirm that full eave screens are an effective way of creating a physical barrier against mosquitoes and keeping house entry rates low (Ogoma *et al.*, 2010; Menger *et al.*, 2016). Confirming the findings of Menger *et. al.* (2016), our results showed limited additional benefit of adding a spatial repellent to a full eave screen where indoor biting rates were the sole outcome measure. If eaves are screened fully, the physical barrier prevents house entry. However, one of the major practical and financial challenges of house screening is ensuring the availability of screens that cover open spaces adequately, particularly where variation in housing structures exists. For scale-up of full eave screening across whole areas, eave screens would need to be repeatedly adapted for each individual house to ensure that all gaps are sealed. This would be labour-intensive and add substantially to the cost of such an intervention.

To counter this inconvenience, the use of repellent-treated eave strips in place of eave screens was suggested. In our study, eave strips filled only half the width of the eave gap in our experimental huts. This meant that the physical barrier present with eave screens, whether treated or untreated, was absent in the eave strip. This lack of physical barrier with an eave strip and impact on house entry was indeed demonstrated in our results where the untreated eave strip did not offer protection indoors as entry of mosquitoes was not prevented.

Use of treated eave strips instead of treated full eave screens would be preferred for several reasons: less fabric would be needed, lower amount of insecticides would be used, a greater applicability to houses of different sizes of eave gaps would be possible and less expertise in ensuring sealing of eave gaps would be needed. The strips would be easier to install and to replace than a full eave screen. In our study, we used half the amount of fabric for eave strips as in eave screens, as well as half the amount of transfluthrin. For a wider eave gap, the reduction in amount of fabric required compared to full screening would be relatively greater, as we anticipate that the same narrow strip of fabric could be used. Variations of the width of eave wrap were not done in this study but could provide measurements of the extent of protection that can be conferred using a 0.05 m wide strip of fabric. We observed protection in outdoor landing rates at 2.5 m from the house, indicative of a substantial spatial repellent effect.

Previous studies have shown that anopheline mosquitoes enter the house predominately through open eaves (Snow, 1987). By sealing these eaves with physical barriers, even in the absence of any insecticide, house entry can be reduced considerably(Atieli *et al.*, 2009; Kampango *et al.*, 2013; Mburu *et al.*, 2018). During house entry, Spitzen *et. al.* 2016 showed that the majority of *An. gambiae* mosquitoes spent a considerable amount of time near the eave gap, proving that an intervention that exposes these mosquitoes to an insecticide or repellent at this point, even at sub-lethal concentrations, can have a significant impact on house entry (Spitzen *et al.*, 2016). If sub-lethal effects have an impact on host-seeking behaviour, this impact on house entry and outdoor biting, could extend to a wider-scale community-level effect. Further work, preferably at the field level, would be needed to demonstrate this conclusively.

In the present study, the use of transfluthrin on eave fabrics further confirmed the potential that spatial repellents have in diverting outdoor-biting mosquitoes that are currently not targeted by any other tool. It is however of particular concern that its protective effect is greatly affected by temperature conditions making future work on how to enable vaporisation of transfluthrin at lower temperatures critical (Glunt *et al.*, 2014; Ogoma *et al.*, 2017). It would be necessary to find ways to increase vaporisation without adapting costly technology with a reliance on electrical power (Masalu *et al.*, 2017). In spite of mortality rates of 91% to pyrethroids, that according to the WHO guidelines is indicative of presence of resistance, mosquitoes in the present study responded to transfluthrin showing promise of spatial repellent use even in times of rising pyrethroid-resistance(WHO, 2013). This advantage is, however, not a long-term guarantee as already studies have shown a reduction of repellent effect on resistant mosquitoes (Agossa *et al.*, 2015) and in the long run, alternative pyrethroid and non-pyrethroid spatial repellents should be considered.

From the present study, while availability of the hessian fabric meant ease of access to the quantity required, the relatively short lifespan of the impregnated repellent may be a challenge if rollout would be considered. In previous studies, repellent on hessian fabric was shown to be effective for up to six months (Ogoma *et al.*, 2012). In our studies, while the exact lifespan of the fabric was not measured, gradual reduction in efficacy was noticed in baseline studies (data not reported here), leading to a decision to use each fabric for a maximum of one experiment spanning sixteen experimental days. If the efficacy were found to decline over a span of more than a month, then the need to replace or re-treat the fabric could increase the cost of the tool making it less ideal for application in rural communities (Kirby *et al.*, 2008; Masalu *et al.*, 2017).

The impregnation process itself, while relatively easy, involved handling of 0.2g/ml concentration of transfluthrin. While this concentration was below the amount considered risky, handling would require

semi-skilled personnel for safety purposes(Tisch *et al.*, 2005). Treatment and handling of the eave fabric in future studies needs to be re-thought in terms of how to increase safety for the people involved in preparing them.

Looking ahead, the use of hessian fabric holds promise as the material is relatively durable, available and mostly affordable (Ogoma *et al.*, 2012; Govella *et al.*, 2015; Masalu *et al.*, 2017). Sourcing for alternative material that may be more available and could remain treated for even longer than hessian fabric, if not longer, may be some of the next steps to be considered. Currently, plastic shields impregnated with transfluthrin have already shown promise due to their relative longevity, easier to handle as they come pre-treated and their ease of application in spaces where protection needs to be achieved (McPhatter *et al.*, 2017).

Malaria transmission is not limited to indoor-biting mosquitoes and as such, tools that can target outdoorbiting mosquitoes are needed to further supplement the existing tools(Killeen, 2014; Sangoro *et al.*, 2014; Benelli *et al.*, 2017). Our results confirm similar experiments implemented in Tanzania in parallel to the here presented work (Mmbando *et al.*, 2018) and demonstrate that the inclusion of a spatial repellent on the screen or strip offers protection against *An. arabiensis* biting in the outdoor environment. Protection against other mosquito species that are inherently outdoor and indoor-biting can be explored to further improve the suitability of this tool for mosquito control.

Further studies could be done to establish the range of protection and potential community effect conferred to provide further information for scaling up of this promising new vector control tool while ensuring that issues arising such as those seen during incomplete coverage (Maia *et al.*, 2016) are adequately addressed, possibly by the addition of supplementary tools to work in complement such as attract and kill devices (Matowo *et al.*, 2013; Mafra-Neto *et al.*, 2019; Mbare *et al.*, 2019).

Conclusion

Transfluthrin-treated fabric strips can provide a simple, easy to handle tool for protecting people from malaria vector bites in the peri-domestic space both indoors and outdoors around the house in climatic areas where evening and night time temperatures are high. Field studies need to confirm these findings under more variable natural conditions and to determine the range and longevity of the spatial repellent effect.

Acknowledgements

We are grateful for the tireless work of the human landing volunteers Leonard Wanga (deceased), Shem Matthews Otieno, Dan Simiyu, and Philip Owigo. We are appreciative to Elisha Obudho and David Alila

for providing all the mosquitoes used in this study. We also acknowledge the technical support of Bruno Otieno and Ferdinand Ong'wen.

Pre-evaluation of push-pull mosquito control under natural conditions in western Kenya

Margaret M. Njoroge, Ulrike Fillinger, Willem Takken, Joop J. A. van Loon and Alexandra Hiscox

To be submitted in modified form

Abstract:

Residual malaria transmission is fuelled in part, by lack of vector control tools that target outdoor-biting mosquitoes. New vector control tools are needed to effectively protect susceptible human hosts from outdoor-biting malaria vectors. Transfluthrin, released passively from an eave wrap as the "push" component, and a Suna trap baited with the MB5 odour blend as a "pull" component were tested under natural conditions in western Kenya; both separately and combined as a "push-pull" tool on their effect on indoor and outdoor host-seeking mosquitoes. Human landing catches were conducted both indoors and outdoors with secondary data obtained from the Suna trap. In the push-pull treatment there were fewer primary malaria vectors *Anopheles gambiae* s.l. and *An. funestus* s.l. mosquitoes indoors than in the control by 4-fold and by 3-fold, respectively. The push only treatment had a similar effect on indoor mosquito numbers as the push-pull treatment. Outdoors, catches of host-seeking primary malaria vectors were not associated with any of the treatments applied. *Culex* spp. and *Mansonia* spp. catches outdoors were reduced both in the presence of push-pull and push only components. It is concluded that the push-pull system with transfluthrin and an odour-baited trap was effective in controlling indoor-biting malaria vectors and outdoor-biting non-malaria mosquitoes.

Keywords:

Residual malaria transmission, Suna trap, Mbita blend 5, Transfluthrin, Push-pull, Mosquito control

Introduction

Development of new and improved vector control tools remains a high priority to counter recent impediments in malaria control (WHO, 2019). Increased coverage of long-lasting insecticide treated bed nets (LLINs), as well as improved diagnostics and treatment contributed to the huge and significant reductions in malaria (Bhatt *et al.*, 2015), but since 2015, further declines have stalled prompting the need for the development of new tools (WHO, 2019). The increase in insecticide resistance and the growing importance of outdoor malaria transmission are concerns that should be addressed to enhance malaria control (Killeen, 2014; Mathania *et al.*, 2016; Meyers *et al.*, 2016; Benelli & Beier, 2017; Moshi *et al.*, 2017; Saavedra *et al.*, 2019; Sherrard-Smith *et al.*, 2019; WHO, 2019).

Development of the push-pull mosquito control tool was designed to target outdoor-biting malaria vectors and to reduce the number of house-entering mosquitoes (Moshi *et al.*, 2017; Saavedra *et al.*, 2019; Sherrard-Smith *et al.*, 2019). The etymology of the tool was derived from the actions exerted by the constituent components with the 'push' exerted by a potent spatial repellent and the 'pull' being applied by a trapping and killing mechanism that is baited with attractants. Extensive semi-field experiments were carried out that screened a diverse range of chemicals for each component of the 'push-pull' system

(Chapter 2, this thesis). In summary, para-menthane-3,8-diol (PMD) was not found to offer passive protection to susceptible humans outdoors against *Anopheles* mosquitoes possibly because of its characteristics as a contact repellent unless volatilized using electrical fans (Barasa *et al.*, 2002; Menger *et al.*, 2014; Murchie *et al.*, 2016). Transfluthrin, a pyrethroid with toxic and repellent effects on mosquitoes had previously been shown to confer a passive protective effect in reducing biting by *Anopheles arabiensis* mosquitoes on human hosts and had been included in some industrial repellent products (Pates *et al.*, 2002; Ogoma *et al.*, 2012; Ogoma *et al.*, 2014; Andres *et al.*, 2015; Govella *et al.*, 2015; Maia *et al.*, 2016; Masalu *et al.*, 2017; Ogoma *et al.*, 2017). Semi-field experiments determined that use of 2.5g/m²transfluthrin impregnated in a strip of fabric applied around the eave as the "push" was effective to reduce outdoor human landing catches by close to 60%. This was therefore decided as the 'push' component in our tool (Chapter 2, this thesis).

The Mbita-blend (MB5), as odour bait released from Suna traps, had been shown to be attractive to Afrotropical *Anopheles* mosquitoes, with field studies showing strong evidence of the impact of the traps on malaria vectors and malaria prevalence (Hiscox *et al.*, 2014; Hiscox *et al.*, 2016; Homan *et al.*, 2016; Menger *et al.*, 2016; Mburu *et al.*, 2019). Following extensive experiments, MB5 supplemented with 2butanone was chosen as our "pull" component even though supplementation with CO₂ increased the catches of the trap in the absence of a human (Chapter 2, this thesis). For the reported field trial, supplementation of MB5 with CO₂ in the Suna trap was done.

The combined use of a transfluthrin spatial repellent (the push) and an odour-baited trap (the pull) on the human landing rate had to our knowledge not been tested in a natural setting. The present study aimed to evaluate the effect of push-pull mosquito control on outdoor and indoor-biting wild mosquito populations with transfluthrin as the spatial repellent. The impact of the separate push-pull constituents was tested as well to determine individual effects on both outdoor and indoor-biting mosquito populations with particular emphasis on malaria vectors.

Materials and methods

Field study site

The study was carried out at Kigoche village located in the Kano flood plains, western Kenya ($00^{\circ}34^{\circ}S$, $034^{\circ}65^{\circ}$ E and 1158 m above sea level). In the flood plains, rice farming is undertaken as an agricultural activity and due to the standing irrigation water, mosquito breeding occurs all through the year. The study took place between June and August 2019. Annual rainfall in this area varies between 1000 and 1800 mm and follows a bimodal pattern with the long rainy season typically occurring between April and June while the short rains occur between October and November. Daily temperatures range between 17 - 31

°C while the average relative humidity is 65% (Mweresa *et al.*, 2015; Mburu *et al.*, 2017). The study was carried out immediately after the long rains prior to the start of the short rains with minimal rainfall recorded in the meteorology department.

Malaria transmission is perennial and caused primarily by *Plasmodium falciparum* which is transmitted by *An. arabiensis* and *An. funestus s.s.* (Mukabana *et al.*, 2012; Mweresa *et al.*, 2015).



Figure 1a. One of the study houses in Kigoche depicting iron sheets roofing, mud walls and wooden door and windows as well as a PVC tent under which the outdoor HLC were conducted. The solar panel on the roof was not linked to this particular study.



Figure 1b. Fitting of the hessian fabric strip in one of the study houses in Kigoche with the fastening of fabric done to ensure equal space above and below the fabric.

Study houses

Four houses each measuring 6 m by 4 m were recruited randomly from a cohort of houses used in a previous study (Menger *et al.*, 2015). All the houses were two-roomed, mud-walled with earthen floors and had corrugated iron-sheet roofs with open eaves and no ceiling (Figure 1a). During human landing catches (HLC) between 19.00 h and 23.00 h, the houses were occupied and householders continued their routine activities.

Transfluthrin eave wrap

Hessian (burlap) fabric procured from local markets was cut and sewn into strips measuring 21 m by 0.5 m and the edges secured with cotton fabric to prevent fraying. Emulsified transfluthrin from Bayer Global (Leverkusen, Germany) supplied at a concentration of 0.2 g/ml was used to impregnate the hessian fabric. Emulsification of transfluthrin eased the preparation of solution which is necessary for impregnation of fabric as transfluthrin is insoluble in water (Ogoma *et al.*, 2012).

The hessian strips were kneaded in a transfluthrin solution with 26.25 ml of EC added in approximately 1300 ml water to achieve a final impregnation concentration of 2.5 g/m² (Ogoma *et al.*, 2012). All fabric was dried under a shelter overnight immediately after impregnation and then rolled up, wrapped in aluminium foil and stored in a 4 °C cold room prior to transportation to the field site. Treated fabric was usually used within a minimum of two days and a maximum of one week after treatment. Fabric thus

treated was used consistently for up to one week (four experimental nights) before being removed and replaced three days later with a freshly treated fabric to allow for complete aeration of houses between treatments (Ogoma *et al.*, 2014).

Application of the eave strip was done by securing it along the eave gap of the houses using nails to fasten on the wooden wall pillars supporting the roof while ensuring an equal gap both above and below the material to allow air flow (Figure 1b). The eave gap was therefore not completely sealed using the transfluthrin-treated hessian fabric.

Odour-baited traps

Suna traps (Hiscox *et al.*, 2014) were used for all set ups and baited with MB5 nylon strips (Mukabana *et al.*, 2012; Mweresa *et al.*, 2015) supplemented with CO₂ which was produced nightly by the fermentation of molasses using yeast (Mweresa *et al.*, 2014). The traps were positioned directly next to the house near the bedroom window with the main trap entry positioned 30 cm from the ground as done in a previous study (Homan *et al.*, 2016). All traps were powered by 26 Ah/12V Chloride Exide batteries (Chloride Exide (K) Ltd, Kenya) which were charged once a week. The traps were run from 18:00 h to 06:00 h on experimental days.

Human landing catches

Human landing catches were conducted indoors and outdoors by four adult male volunteers living in the area and who had consented to participate in the study. Their ages ranged between 25 and 50 years. Weekly *Plasmodium* screening using SD Bio-line Rapid Diagnostic Test (Abbot Illinois, USA) was done on the volunteers and artemisinin combination therapy (ACT) administered in case of a positive test. Those who tested positive were excluded from the study for one week and replaced by one of two extra volunteers recruited as standby volunteers and also living in the same area. The volunteers washed their legs prior to the start of mosquito collection using non-odorous bar soap supplied by the investigators. The volunteers sat on chairs wearing appropriate protective clothing with only the legs exposed and, using a mouth aspirator, trapped mosquitoes landing on their exposed legs.

For indoor HLCs, a chair was positioned in the middle of the first room upon entering the front door designated 'sitting room' during which time normal household activities continued. Outdoor HLCs were conducted at 2.5 m from the house with a Polyvinyl Chloride (PVC) tent cover (Figure 1) provided as protection from rain. During outdoor HLC no CDC light trap was hung inside the house.

HLCs were conducted from 19.00 h to 23.00 h with up to 10 min break allowed every hour.

Mosquitoes caught were transferred to collecting cups labelled according to the collection hour (one collection cup per hour per volunteer) and stored in a cool dry place before being transported to the laboratory for identification and counting the next morning.

Mosquito identification

Morphological identification was done prior to enumeration of mosquitoes caught according to genus and species under the microscope (Gillies & Coetzee, 1987). All mosquitoes were thereafter stored in -20°C freezers for future studies. All collections were entered in an electronic data storage system.

Table 1: Treatment allocation to the four study houses during the 8-week study period following a Latin square design with HLC location indicated

	Week 1	Week 2	Week 3	Week 4	Week 1	Week 2	Week 3	Week 4
	(4 nights)	(4 nights)	(4 nights)	(4 nights)	(4 nights)	(4 nights)	(4 nights)	(4 nights)
	Outdoor	Outdoor	Outdoor	Outdoor HLC	Indoor HLC	Indoor HLC	Indoor HLC	Indoor HLC
	HLC	HLC	HLC					
House 1	Push-pull	Pull only	Push only	Control	Push-pull	Pull only	Push only	Control
House 2	Pull only	Push only	Control	Push-pull	Pull only	Push only	Control	Push-pull
House 3	Push only	Control	Push-pull	Pull only	Push only	Control	Push-pull	Pull only
House 4	Control	Push-pull	Pull only	Push only	Control	Push-pull	Pull only	Push only

Chapter 5

Study design

Each of the four houses was randomly allocated a number 1-4 and four treatments: push-pull, push only, pull only and control applied in each with a weekly variation according to Table 1.

The treatments in the houses were as follows:

- Push-pull house: Eave fabric, transfluthrin-treated; Suna trap, baited with MB5 + CO₂
- o Pull only house: Eave fabric, not treated; Suna trap, baited with MB5+ CO2
- o Push only house: Eave fabric, transfluthrin-treated; Suna trap, unbaited
- o Control house: Eave fabric, not treated; Suna trap, unbaited

After four nights all the fabrics and baits were removed and fresh ones were applied three days later to allow for sufficient aeration of the houses prior to the next treatment. Human landing catches were conducted for four weeks outdoors and then four weeks indoors in all the houses (i.e. all four out, then all four in). As Suna traps were part of the push-pull set up, catches in them were considered as secondary data when baited (pull only and push-pull) and when unbaited (push only and control).

Statistical analysis

All data analysis was done using R version 3.5.1 (R Core Team, 2018). Data preparation, visualization and statistical analyses were conducted using functions from the packages gee, geepack, effects and lme4. Generalized estimating equations (GEE) using the geeglm function were used to analyse the effect of different treatments; specifically the repellent only (push only), odour-trap only (pull only) and the combination (push-pull) against untreated control on mosquito catches(Ballinger, 2004). The GEE was fitted with a Poisson distribution and a log-link function with an exchangeable matrix.

Human landing catches (HLC) was the primary indicator to determine the effect of treatments on the number of host-seeking mosquitoes. Odour-baited Suna traps provided secondary indicators (pull only and push-pull).

Experimental days were included in the model as repeated measurements. Experimental houses and HLC volunteers were tested as factors and since they did not significantly affect the model, were subsequently excluded from the statistics. The treatment effect on both outdoor and indoor human landing rates (HLC) was calculated on the total number of female mosquitoes per species/genera per trapping night with a test on

Results

In total, 8267 mosquitoes were caught over the 8-week study in all the traps used: 1623 *Anopheles gambiae* s.l. among which, 1615 (99.5%) were female, 1159 *An. funestus* s.l. among which 1087 (93.8%) were female, 432 *An. coustani* s.l. among which 422 (97.7%) were female, 50 *An. pharoensis* among which 43 (86%) were female, 3893 *Culex* spp. among which 3883 (99.7%) were female and 1110 *Mansonia* among which 1089 (98.1%) were female.

All effects were calculated in relation to population of female mosquitoes.

House variations

The four houses had variations in mosquito catches though the differences were not statistically significant. The estimated mean catches of the various mosquito genera and species per house is presented in Table 2.

caught	using HLC during th	le 8-week study period	l with 95% CI included.	The mean catches we	re calculated from st	atistical models.
	An. gambiae s.l.	An. funestus s.l.	An. coustani	An. pharoensis	Culex spp.	Mansonia spp.
House	9.64 (95% CI	5.89 (95% CI	0.81 (95% CI 0.62-	0.13 (95% CI 0.07-	8.64 (95% CI	4.49 (95% CI 3.95-
1	8.80-10.55)	5.24-6.62)	1.07)	0.26)	7.95-9.39)	5.09)
House	6.82 (95% CI	3.82 (95% CI	0.70 (95% CI 0.52-	0.01 (95% CI	14.44 (95% CI	5.59 (95% CI 4.98-
2	6.13-7.59)	3.31-4.41)	0.95)	0.001-0.07)	13.47-15.48)	6.29)
House	9.47 (95% CI	4.98 (95% CI	2.30 (95% CI 1.92-	0.34 (95% CI 0.21-	16.65 (95% CI	3.51 (95% CI 3.02-
3	8.64-10.39)	4.39-5.65)	2.76)	0.57)	15.58-17.79)	4.07)
House	3.63 (95% CI	4.65 (95% CI	2.37 (95% CI 1.98-	0.04 (95% CI 0.01-	15.30 (95% CI	5.20 (95% CI 4.61-
4	3.13-4.20)	4.08-5.29)	2.84)	0.12)	14.30-16.37)	5.87)

Table 2: Estimated mean catches per experimental night of the major mosquito genera and species in each of the four study houses

Chapter 5





Figure 2. **Outdoor HLC** *Anopheles* mosquito catches in the presence of the four treatments applied. The graphs represent a. *An. gambiae* s.l. and b. *An. funestus* s.l. catches outdoors collected by HLC with the bold line in the box plots showing median catches and the box containing the upper and lower quartiles.



Figure 3. **Outdoor HLC** *Culex* spp. mosquito catches in the presence of the four treatments applied. The box plots represent the median catches (bold line) with the boxes representing both the upper and lower quartiles catches.
Volunteer variations

The four volunteers who participated over the eight weeks attracted variable numbers of mosquitoes per night though none were statistically different. Due to the rotation of treatments in their houses, this variation was adequately catered for.

Outdoor HLC catches

Outdoor HLC catches of malaria vectors, both *An. gambiae* s.l. and *An. funestus* s.l., were not different in the presence of the various treatments compared to the control (Figure 2; Table 3).

In the presence of the push-pull treatment *Culex* spp. mosquito catches were 2-fold lower than in the control (p<0.001; Figure 3; Table 3). Catches of *Culex* spp. in the presence of the push only treatment were similar to those obtained in the presence of the push-pull treatment (Table 3).

Table 3: Summary of statistical model outputs for analyses on catches of major malaria vectors, *Culex* spp. and *Mansonia* spp. mosquitoes by **outdoor HLC** in the presence of fabric and Suna trap that were untreated (control), untreated fabric and baited Suna (pull), treated fabric and unbaited Suna (push) and treated fabric and baited Suna (push-pull). All estimated means were calculated from statistical modelling while comparisons of catches in the different traps were made to the catches in the control set up. Figures highlighted in red denote statistical significance.

Female mosquitoes cau	ight outdoors	using HLC		
Mosquito species	Treatment	Estimated means	Odds Ratio (95% CI)	p-value
An. gambiae s.l.	Control	10.25(95% CI 8.77-11.91)	-	-
	Pull only	14.08(95% CI 12.36-16.04)	1.38 (0.84-2.26)	0.205
	Push only	12.77(95% CI 11.14-14.65)	1.25(0.74-2.09)	0.398
	Push-pull	13.15(95% CI 11.49-15.05)	1.29(0.60-2.75)	0.516
An. funestus s.l.	Control	2.12(95% CI 1.52-2.97)	-	-
	Pull only	2.43(95% CI 1.78-3.33)	1.15(0.64-2.07)	0.649
	Push only	1.87(95% CI 1.31-2.68)	0.88(0.41-1.89)	0.747
	Push-pull	2.37(95% CI 1.73-3.26)	1.12(0.66-1.91)	0.683
Culex spp.	Control	65.01(95% CI 61.18-69.08)	-	-
	Pull only	70.48(95% CI 66.49-74.72)	1.08(0.80-1.48)	0.609
	Push only	31.48(95% CI 28.85-34.35)	0.48(0.32-0.74)	< 0.001
	Push-pull	28.80(95% CI 26.29-31.55)	0.44(0.31-0.63)	< 0.001
Mansonia spp.	Control	13.56(95% CI 11.89-15-47)	-	-
	Pull only	13.15(95% CI 11.51-15.03)	0.97(0.64-1.48)	0.886
	Push only	9.39(95% CI 8.03-10.99)	0.69(0.47-1.01)	0.057
	Push-pull	10.16(95% CI 8.73-11.82)	0.75(0.52-1.08)	0.126





Figure 4. Indoor HLC *Anopheles* mosquito catches in the presence of the four treatments applied. The graphs represent a. *An. gambiae* s.l. and b. *An. funestus* s.l. catches indoors collected by HLC with the box plots showing median catches (bold line) as well as upper and lower quartiles.

Indoor HLC catches

In the presence of push-pull and push only treatments, *An. gambiae* s.l. HLC catches indoors were almost 4-fold lower than those in the control treatment (push-pull: p<0.001; Figure 4; Table 4). Catches of *An. funestus* s.l. in the presence of either push-pull or the push only treatments were almost 3-fold lower than those in the control treatment (push-pull: p<0.001; Figure 4; Table 4). Catches of *Culex* spp. in the push-pull treatment were half of those in the control treatment (p=0.136; Table 4).

Table 4: Summary of statistical model outputs for analyses on catches of major malaria vectors and *Culex* spp. mosquitoes by **indoor HLC** in the presence of the four treatments (control, pull, push and push-pull). All estimated means were calculated from statistical modelling while comparisons of catches in the different traps were made to the catches in the control set up. Figures highlighted in red denote statistical significance.

Female mosquitoes cau	ight indoors	using HLC		
Mosquito species	Treatment	Estimated means	Odds Ratio (95% CI)	p-value
An. gambiae s.l.	Control	5.31(95% CI 4.30-6.56)	-	-
	Pull only	6.71(95% CI 5.56-8.11)	1.26(0.50-3.20)	0.621
	Push only	1.28(95% CI 0.84-1.97)	0.24(0.08-0.72)	0.011
	Push-pull	1.40(95% CI 0.93-2.11)	0.26(0.13-0.52)	< 0.001
An. funestus s.l.	Control	9.15(95% CI 7,78-10.76)	-	-
	Pull only	7.30(95% CI 6.09-8.74)	0.80(0.55-1.16)	0.235
	Push only	3.34(95% CI 2.56-4.36)	0.37(0.13-1.02)	0.055
	Push-pull	2.84(95% CI 2.13-3.80)	0.31(0.17-0.56)	< 0.001
<i>Culex</i> spp.	Control	3.43(95% CI 2.65-4.44)	-	-
	Pull only	4.33(95% CI 3.43-5.46)	1.26(0.49-3.2)	0.624
	Push only	1.11(95% CI 0.72-1.71)	0.32(0.15-0.69)	0.003
	Push-pull	1.90(95% CI 1.36-2.66)	0.55(0.26-1.20)	0.136

Suna trap catches

During outdoor HLC, Suna trap catches of *An. gambiae* s.l. and *An. funestus* s.l. were lower in the push-pull treatment than in the pull-only treatment, with *An. funestus* s.l. showing a significant difference (p=0.017; Table 5). Similarly, *Culex* spp. mosquito catches were also lower in the push-pull treatment than in the pull only (p<0.001; Table 5).

During indoor HLC, Suna trap catches of *An. gambiae* s.l were comparable in the pull-only and the push-pull treatments (Table 5). Catches of all other mosquito species were slightly lower in the push-pull treatment than in the pull only though were not of statistical significance (Table 5).

tatistical model outputs for analyses on catches of major malaria vectors as well as Culex spp. and Mansonia	ted outdoors by Suna traps at the same time as when HLC was being conducted outdoors and indoors. All	alculated from statistical modelling. Figures highlighted in red denote statistical significance.
Fable 5: Summary of statistical model ou	spp. mosquitoes collected outdoors by S	estimated means were calculated from sta

	Female moso	quitoes caught using Suna tra	ıp when HLC is outdoors	S
Mosquito species	Treatment	Estimated means	Odds Ratio (95% CI)	p-value
An. gambiae s.l.	Pull only	3.87(95% CI 3.02-4.97)	ı	ı
	Push-pull	3.06(95% CI 2.31-4.05)	0.79(0.48-1.30)	0.351
An. funestus s.l.	Pull only	7.93(95% CI 6.67-9.44)	·	ı
	Push-pull	4.75(95% CI 3.79-5.94)	0.60(0.39 - 0.91)	0.017
Culex spp.	Pull only	6.87(95% CI 5.68- 8.29)	ı	ı
	Push-pull	2.69(95% CI 2.01-3.62)	0.39(0.26-0.59)	<0.001
Mansonia spp.	Pull only	3.61(95% CI 2.79-4.68)	ı	ı
	Push-pull	2.17(95% CI 1.56-3.02)	0.60(0.29-1.25)	0.170
	Female moso	quitoes caught using Suna tra	ıp when HLC is indoors	
An. gambiae s.l.	Pull only	9.11(95% CI 7.75-10.72)	ı	ı
	Push-pull	9.05(95% CI 7.69-10.65)	0.99(0.63-1.56)	0.976
An. funestus s.l.	Pull only	7.71(95% CI 6.47-9.20)	ı	ı
	Push-pull	5.41(95% CI 4.39-6.68)	0.70(0.44-1.12)	0.130
Culex spp.	Pull only	6.37(95% CI 5.24-7.75)	I	ı
	Push-pull	4.19(95% CI 3.30-5.32)	0.66(0.37-1.17)	0.150
Mansonia spp.	Pull only	3.87(95% CI 3.02-4.97)	I	ı
	Push-pull	2.97(95% CI 2.23-3.94)	0.77(0.43-1.35)	0.360

Discussion

In the present study, the push-pull set up was not associated with a reduction in catches of outdoor-biting malaria vectors. Numbers of other mosquito species, however, specifically those belonging to the genera *Culex* and *Mansonia*, were reduced in the presence of both the push-pull set up and the push only set up. Indoors, the push-pull set up was associated with reduction in both indoor-biting malaria vectors and other mosquito species, specifically *Culex* mosquitoes. It is possible that the spatial repellent, transfluthrin, in the eave fabrics affected house-entry of mosquitoes leading to lower mosquito catches indoors (Menger *et al.*, 2015; Mmbando *et al.*, 2018; Mmbando *et al.*, 2019; Mwanga *et al.*, 2019). This may be substantiated by the similar levels of indoor protection seen in the present study by both push only and push-pull treatments whose common component is the use of transfluthrin in a passive-release form from the eave fabrics applied on the eave gap. Interestingly though, the indoor reduction in mosquito population was very species-specific in magnitude with *An. gambiae* s.l. showing the largest reduction in absolute mean numbers. Reduction in indoor numbers of *An. funestus* s.l. shows promise for additional malaria control efforts as this is a major indoor-biting vector (Moiroux *et al.*, 2014; Zhou *et al.*, 2016; Limwagu *et al.*, 2019).

A very similar study had been done in the same area in some of the same houses using a different spatial repellent (delta-undecalactone) in combination with a similar odour-bait as in the present study (Menger *et al.*, 2015). The former study showed a significant reduction in malaria vector indoor catches in the presence of both push only and push-pull control with up to 50% difference in catches with simulations predicting up to 20-fold reduction if applied on large scale (Menger *et al.*, 2015). Unlike the present study, Menger *et al.*(2015) showed significant reduction in indoor malaria vector catches in the presence of pull only set up (Menger *et al.*, 2015).

Outdoor responses of mosquito species to the push-pull set up in the present study showed reduction in numbers of *Culex* spp. and *Mansonia* spp. but not with the malaria vectors. In terms of numbers, *Culex* mosquitoes were the most abundant and in reducing their numbers by more than half could be considered as a benefit against nuisance biting with lower numbers available for the transmission of lymphatic filarial worms and arboviruses (Verhulst *et al.*, 2015; Ogoma *et al.*, 2017; Nchoutpouen *et al.*, 2019).

One possible contributor to the low response among the malaria vectors in the present study could have been that the level of transfluthrin volatilization was insufficient for *Anopheles* mosquitoes, possibly due to lower night temperatures (Ogoma *et al.*, 2017). Transfluthrin requires a minimum ambient temperature to sufficiently volatilize hence the use of electrical volatilisers and burning coils in commercial products where it has been incorporated(Pates *et al.*, 2002; Jeyalakshmi *et al.*, 2014; Ogoma *et al.*, 2014; Bibbs *et*

al., 2018; Kwan *et al.*, 2018), as well as better protection seen on days which had higher temperatures (Ogoma *et al.*, 2017).

Very recent studies done in Tanzania that combined transfluthrin-treated eave fabric and BG Sentinel malaria traps showed a higher level of protection against malaria vectors outdoors than in the present study using all the components of push-pull; separately and combined (Mmbando *et al.*, 2017; Mmbando *et al.*, 2019). One main difference in the set ups included the use of 2 traps, 15 m away from the house that resulted in higher protection compared to one and four traps that were tested(Mmbando *et al.*, 2019). Several studies have explored the utilization of different passive emanators for releasing transfluthrin outdoors ranging from chairs, mats and decorations that could be explored instead of the eave fabric (Masalu *et al.*, 2017; Masalu *et al.*, 2018; Masalu *et al.*, 2020).

The contribution of the pull component in reducing catches in the present study was not significant neither for indoor nor outdoor catches. In the cases of *An. gambiae* s.l. and *Culex* mosquitoes, the presence of the pull-only component actually increased the number of mosquitoes caught by the human volunteers indoors. This was an interesting observation considering similar set ups with MB5 supplemented with either CO_2 or 2-butanone had shown significant reductions in house-entry numbers in previous studies (Menger *et al.*, 2014; Menger *et al.*, 2015; Homan *et al.*, 2016; Menger *et al.*, 2016). The explanation for this could be that once mosquitoes were proximal to the house due to the combined attraction of the host cues and artificial lure, the unprotected human provided a more attractive host for the mosquitoes than the trap, directing the mosquitoes to the human (Takken & Knols, 1999; Carde, 2015; Hawkes & Gibson, 2016).

The presence of the human conducting HLC outdoors led to a reduction by one-third of *An. gambiae* s.l. in the Suna trap compared to when the human was indoors. The effect of the repellent in the trap catches was seen in reduced trap catches in almost all species with push-pull in place. But while the trap catches reduced in the presence of the repellent, the human landing catches were similar both in the presence of the repellent alone or a combination of the trap and repellent indicating how much more attractive an unprotected human remains to the mosquito even in the presence of a spatial repellent. The implication of this on the provision of a peri-domestic safe area for unprotected humans would be a requirement for stronger manipulation of host-seeking mosquitoes to sufficiently divert them away from humans and into a trapping device.

In the present study we used four houses and attempted to provide insightful data in natural conditions that can inform larger studies which will in turn generate throughput data to guide the development of an outdoor malaria vector control tool. This study provided substantial information that will guide in the fine-tuning of the push-pull mosquito control tool. One is the consideration of an alternative, more

effective spatial repellent that could work effectively against malaria vectors outdoors and an improved pull component that could synergistically work with the repellent and produce a stronger protection than either the repellent or the pull component alone. Additionally, consideration on the production and lifespan of the push-pull control products may be made to make them more effective for longer duration and easier to prepare for large scale roll out.

Conclusion:

The transfluthrin/MB5 push-pull system, while not effective at reducing the number of outdoor-biting malaria mosquitoes, caused significant reductions of indoor-biting malaria mosquitoes and therefore provided additional indoor protection. Bites of other mosquito species were reduced both outdoors and indoors. Further research is needed to achieve protection against malaria vectors outdoors with the aim to reduce residual malaria transmission.

Acknowledgements

We are grateful for the community at Kirindo village for accommodating us for the entire study period, we acknowledge Charles Amara for his tireless technical work during the study with additional technical support from Fedinand Ong'wen. We would like to acknowledge the human landing volunteers, namely Paul Odiwuor, Moses Ochieng, Walter Ochieng and Maurice Otieno and the village elder Charles Anyona for assisting in their coordination when necessary. We are grateful for the Chief of the area who initially gave us permission to work in the area and introduced us to the local leaders through which we made our entry into the study area.

General Discussion

General Discussion

This thesis explored the changes in host-seeking behaviour of malaria vectors exposed to spatial repellents and attractants, either separately or in combination, to inform the development of a potential control measure targeting outdoor-biting malaria vectors. Malaria transmission has been shown to be affected by changes in mosquito behaviour, with, in many areas, more mosquitoes caught outdoors than indoors, giving an indication of an increasing role in outdoor-transmission. In western Kenya *Anopheles arabiensis* is the predominant malaria vector outdoors, while *An. funestus* remains the most abundant species indoors (Menger *et al.*, 2015; Degefa *et al.*, 2017; Ogola *et al.*, 2018).

In my study, application of transfluthrin as a spatial repellent on eave fabric significantly reduced the number of mosquito bites both outdoors and indoors. The efficacy of transfluthrin-treated eave fabric in providing significant protection to human hosts outdoors was similar when the eave fabric was presented as a screen or when presented as a strip (partial covering of eave gap). The transfluthrin concentrations in the air declined in a linear fashion from the source of release with higher concentrations detected closer to the ground than higher up. This research has also shown that screening eave gaps with untreated eave fabric confers protection indoors by preventing house entry of malaria vectors. The research in this thesis explored the utilization of a push-pull system; attractant-baited traps in combination with transfluthrin-treated eave strips. However, no additional benefit of the push-pull system over the push only, deployed as transfluthrin-treated eave strips, was seen neither for outdoor nor for indoor protection. The research showed that the odour-baited Suna trap has a better trapping efficiency when supplemented with carbon dioxide than with 2-butanone, a previously reported potential mimic for *An. funestus* s.s. Additionally, my study showed that an attractant-baited Suna trap was not able to divert host-seeking mosquitoes from unprotected humans in a pull-only set up.

The main findings and opportunities for future research arising from this thesis are discussed below.

Malaria control in Africa

Utilization of long-lasting insecticide-treated bed nets (LLINs) and Indoor residual spraying (IRS) in controlling adult malaria vectors has been the core of malaria control and elimination since 2000(WHO, 2019). Since the call to scale up these interventions, over 1 billion LLINs have been distributed in malaria endemic areas leading to a reported 54% coverage in sub-Saharan Africa by 2016(WHO, 2019). This massive effort led to a 62% reduction in *Plasmodium falciparum* malaria mortality in Africa between the years 2000 and 2015 (Bhatt *et al.*, 2015; WHO, 2019). Other intervention indicators such as intermittent

preventive treatment in pregnancy and prevalence of severe anaemia also pointed to a decrease in prevalence during this period (Hershey *et al.*, 2017; Kayentao *et al.*, 2018; Wangdi *et al.*, 2018).

Challenges in vector control

After 2015, stagnation occurred in the decrease of mortality rates and in a few worrying cases, increase in infection rates was reported, which adversely affected malaria control efforts(Kiware *et al.*, 2017; WHO, 2019).

Several reasons have been given for the recent lack of progress. Use of LLINs and IRS target indoorbiting mosquitoes and while they reduce malaria incidence, they contribute to growing levels of insecticide resistance and do not offer protection from outdoor transmission (Killeen, 2014; Meyers *et al.*, 2016; Benelli *et al.*, 2017). Physiological changes in mosquito host-seeking behaviour from more predominant late-night and indoor-biting mosquitoes to increasing occurrence of outdoor biting taking place earlier in the evenings or later in the morning (Moiroux *et al.*, 2014; Sougoufara *et al.*, 2016; Sherrard-Smith *et al.*, 2019) demonstrates the need for supplementary tools (Killeen, 2014; Meyers *et al.*, 2016; Sherrard-Smith *et al.*, 2019) which inspired this thesis.

The overall objective of malaria control efforts is to interrupt parasite transmission, as this is the most effective way to prevent parasites from being transferred to new hosts (Smith *et al.*, 2007). LLINs and IRS have contributed much to reach this goal, but I have outlined that insecticide resistance as well as increased outdoor-biting behaviour are growing obstacles to meet this objective. These developments provide challenges that need to be addressed.

Evaluating putative repellent 'push' and attractive 'pull' components for manipulating the odourguided orientation of malaria vectors in the peri-domestic space

The natural biting propensity of *An. arabiensis* in western Kenya was explored in Chapter 2 with the findings showing that there was a greater risk of being bitten outdoors than indoors by this particular malaria vector species. The area around the house, referred to as the peri-domestic area, was assumed to be where most of the early evening activities were conducted in rural western Kenya and therefore the area that warranted protection for susceptible humans prior to retreating indoors, where existing tools confer protection (Manin *et al.*, 2016; Finda *et al.*, 2019).

These findings further emphasised the need to develop tools that can be protective against outdoor malaria transmission and that can be supplemented to the indoor tools in use (Benelli*et al.*, 2017). Another major reason for the diminishing gains in mosquito control is the emergence of insecticide-resistant mosquitoes due to the selection pressure applied following constant use of insecticide-based interventions in both agriculture and in public health vector control (Nkya *et al.*, 2014; Benelli *et al.*, 2017; Awolola *et al.*, 2018; Rakotondranaivo *et al.*, 2018; WHO, 2019). Utilization of other insecticide groups has been limited by the high toxicity for mammals and other non-target organisms (Tingle *et al.*, 2003; King *et al.*, 2015; Eldakroory *et al.*, 2017; Han *et al.*, 2018) as well as evidence of pre-existing resistance in malaria vectors against all the major insecticide groups (Hemingway *et al.*, 2000; Cook *et al.*, 2018). Plant-based oils have in the past been used to either repel or kill disease vectors with the added advantage of being environmentally friendly due to being biodegradable, less toxic to humans and non-target organisms and less likely to induce resistance in insects(Diaz, 2016).

I investigated the potential of para-menthane-3,8-diol (PMD) to achieve protection for humans in the peridomestic area as a spatial repellent (Chapter 2). The results showed no apparent repellent effect of PMD on insectary-reared *An. arabiensis*, over a range of concentrations. The most likely reason for this was that PMD in my study did not volatilize to sufficiently high aerial concentrations. In comparison with the Menger *et al.* (2014) study, I relied on passive release of PMD from an eave fabric while the latter utilized multiple mechanical fans to avail PMD spatially(Menger *et al.*, 2014). It appears, therefore, that without this facilitation to release PMD with powered fans, it is more likely that PMD acted as a contact repellent as cited in several publications (Carroll *et al.*, 2006; Diaz, 2016; Colucci *et al.*, 2018; Tangena *et al.*, 2018). PMD is considered to volatilize less than essential oils, making it more likely to offer protection as a topical application to users over a longer period of time (Carroll *et al.*, 2006; Maia *et al.*, 2011). Additionally, as it is derived from plant extracts, PMD is perceived to be safer for use by consumers compared to synthetic repellents while providing protection to a level similar as potent repellents such as DEET (Carroll *et al.*, 2006).

Spatial repellents present more opportunities for application in Africa as they volatilize easily and reduce mosquito-human contact probability while offering protection to both users and non-users at relatively low costs and a low demand on technology (Achee *et al.*, 2012; Norris *et al.*, 2017; Maia *et al.*, 2018).Use of sub-lethal concentrations of insecticides creates opportunities of use as they can evoke a non-toxic but repellent effect on the mosquitoes, which reduces their vectorial capacity while utilizing lower amounts leading to a more cost effective application in the long run(Shaalan *et al.*, 2005; Guedes *et al.*, 2017; Sampaio *et al.*, 2017). Insecticides thus applied may repel mosquitoes or compromise their ability to locate potential human hosts thereby reducing the chances of transmitting diseases without necessarily

inducing an immediate toxic or lethal effect(Cohnstaedt *et al.*, 2011; Mbare *et al.*, 2013).On the downside, it has been shown that sub-lethal concentrations of insecticides could actually increase the fitness of insects such as mosquitoes and subsequently increase their vectorial capacity in transmitting diseases (Margus *et al.*, 2019). Another undesired effect, however, could be the gradual selection of multiple resistance genes in the population as opposed to the selection of one major single resistance gene leading to the undesirable increase of low level resistance mechanisms that may increase in frequency in the population(Guedes *et al.*, 2017). Thirdly, it is reported that sub-lethal concentrations could increase the risk of major resistance developing in the population eventually as opposed to if there would have been no selection pressure(Gressel, 2011).

Transfluthrin, a fast-acting pyrethroid, has been used in commercial products such as mosquito coils and electric vaporizers and has been shown to volatilize easily and act as both repellent and a killing agent (Nazimek *et al.*, 2011; Jeyalakshmi *et al.*, 2014; Ogoma *et al.*, 2014b). Due to unavailability of power sources to run electric vaporisers in rural areas and in order to reduce risks involved in the inhalation of toxic smoke when burning coils, the passive release of transfluthrin impregnated in locally available hessian fabric has been explored for both indoor and outdoor protection with marked success (Masalu *et al.*, 2018; Mmbando *et al.*, 2018; Mwanga *et al.*, 2019a; Masalu *et al.*, 2020).

In Chapter 2,I examined the utilization of a sub-lethal concentration of transfluthrin, applied on an eave fabric for passive release to achieve protection against *An. arabiensis* in the peri-domestic area. Frequency of mosquito bites outdoors in the peri-domestic area was significantly reduced in the presence of transfluthrin-treated eave fabric.

Due to previous studies that pointed to the risk of spatial repellents diverting mosquitoes to unprotected persons when coverage in a community was incomplete(Moore *et al.*, 2007; Maia *et al.*, 2016), the inclusion of a trap baited with an attractive lure in combination with the spatial repellent was explored in a "push-pull" set up. This was heavily inspired by the push-pull strategy developed for application in agriculture where the use of a non-preferred plant species intercropped with a target crop plant (e.g. maize) diverted stem borer moths away (push) and to an alternative 'trap' plant (pull) (Cook *et al.*, 2007; Khan *et al.*, 2016).

Host-seeking malaria mosquitoes have been known to locate their potential blood hosts based on the detection of carbon dioxide and a range of other physical and chemical cues released naturally by the host that the mosquito has evolved to detect when in need of a blood meal (Takken *et al.*, 1997; Takken *et al.*, 1999; Ray, 2015).Following identification of key chemicals involved in host-seeking behaviour, the

Mbita blend 5 (MB5) was developed and used in traps together with carbon dioxide for effective surveillance and control of malaria vectors (Mukabana *et al.*, 2012; Mweresa *et al.*, 2015).

In Chapter 2 it was shown that the MB5 lure in combination with carbon dioxide did not confer protection to susceptible humans outdoors. The attractiveness of the human in the presence of the synthetic lure was significantly higher and thus the odour-baited trap did not reduce the human landing catches in the system. In the absence of a human, the odour-baited trap, however, captured at least one-third of all released mosquitoes, which was consistent with previous studies (Hiscox *et al.*, 2014; Mburu *et al.*, 2017).

Supplementation of odour-baited traps with carbon dioxide has presented challenges in the field as the acquisition and proper storage of the gas in any form is logistically demanding. This necessitated the exploration of carbon dioxide mimics (Turner *et al.*, 2011; Mburu *et al.*, 2017). One mimic, 2-butanone, showed promising results in field studies (Turner *et al.*, 2011; Hiscox *et al.*, 2016; Homan *et al.*, 2016; Mburu *et al.*, 2017). Replacement of carbon dioxide with 2-butanone showed significantly lower *An. arabiensis* catches (Chapter 2),which was not comparable to catches obtained with carbon dioxide alone or when carbon dioxide with the MB5 lure and as such2-butanone was not considered an adequate mimic for carbon dioxide for the capture of *An. arabiensis*. Previous studies have implied that use of 2-butanone as a mimic of carbon dioxide with MB5 lure in Suna traps may be more attractive to *An. funestus* and was able to effectively reduce its population size following mass trapping(Homan *et al.*, 2016; Mburu *et al.*, 2017).

A combination of repellents and attractants has been proposed to potentially synergize the protective effect conferred by each alone and in turn to provide increased protection against both indoor and outdoor-biting mosquitoes (Takken, 2010; Menger *et al.*, 2014; Mmbando *et al.*, 2019).

The combination of transfluthrin-treated eave fabric and MB5-baited Suna trap led to a reduction in the proportion of outdoor-biting *An. arabiensis* but was not better than the repellent alone (Chapter 2). Therefore, I concluded that the combination of transfluthrin and an odour-baited trap did not result in additional protection against outdoor-biting *An. arabiensis*. In line with my results, recent studies done in Tanzania showed only a modest additional protection conferred by the push-pull set up against outdoor-biting mosquitoes compared to the push only set up (Mmbando *et al.*, 2019). The inclusion of an odour-baited trap may not have provided additional personal protection due to its proximity to both the repellent and the human. In Chapter 3, the influence of the spatial repellent on trapping effectiveness of the odour-baited trap was demonstrated when recapture rates dropped from 30% to under 5%. Based on this and observations made in other studies (Menger *et al.*, 2015; Mmbando *et al.*, 2019), the odour-baited traps

should be positioned further away from the house and monitored more on a community-level – users and non-users alike, as opposed to direct personal protection(Mmbando *et al.*, 2019). This was further shown from mathematical simulations developed to determine the value of push-pull with impact at community level being the main focus(Menger *et al.*, 2015).

Detection of mosquito control chemicals in the air provides additional data on the actual concentrations presented to mosquitoes as well as offer information on the levels of exposure as part of public health advise on safety (Martin *et al.*, 2013; Ogoma *et al.*, 2014a; Kwan *et al.*, 2018). In Chapter 2, quantification of chemicals in air samples obtained in the push-pull system showed an approximately linear reduction in transfluthrin concentration from the point of release outwards and upwards with lowest concentrations being detected at the furthermost point of sampling and at higher points of sampling. MbitaBlend-5 (MB5) chemicals were only sporadically detected. Availability of the chemicals in the air as confirmed in Chapter 2 further confirms the possibility of presenting the push-pull control to mosquitoes outdoors and influencing their approach to the human host. Future prospects are improvement of the push-pull mosquito control tool by ensuring that effective concentrations are reached separately and when combined preferably with synergistic effects to reduce host biting frequency. Utilization of active dispensers, especially for the spatial repellent, would further increase the release rate and increase aerial concentration though this would inadvertently increase the costs and technical requirements making this an inhibiting factor for large-scale roll out for malaria control.

Sight versus smell: Dynamics of malaria mosquito attraction response in the presence of spatial repellents

Host-seeking mosquitoes utilize both physical and olfactory cues detected by different sensory pathways in order to successfully locate a potential host for a blood meal (Cardé *et al.*, 2010; Takken, 2011; Hawkes *et al.*, 2017). Repellents have been shown to work by interfering with host-seeking mosquitoes' perception of human odours through either olfactory masking or undefined "irritation" of mosquitoes such that the frequency of contact with a potential human host is either reduced or is avoided altogether, leading to a reduced probability of a successful blood-meal(Bohbot *et al.*, 2010; Murphy *et al.*, 2013; Ray, 2015; Tzotzos *et al.*, 2018). Further, it has been shown that in many cases, compounds that at relatively low concentrations act as attractants, can function as repellents at higher concentrations indicating a concentration-dependent switch of olfactory perception resulting in attractants in a pushpull set up towards synergizing the protection conferred by each individual component against outdoor-

biting mosquitoes (Chapter 2) could inadvertently cause diminished perception of the attractive lure by the olfactory neurons(Bohbot *et al.*, 2010; Tsitoura *et al.*, 2015; Tzotzos *et al.*, 2018).

Utilization of a different attractive cue such as artificial light in the form of the CDC UV light trap was proposed in my study to be used as the pull component in the push-pull system due to its trapping efficiency seen in various studies(Sexton *et al.*, 1986; Rubio-Palis, 1996; Mwanga *et al.*, 2019b).As nocturnal mosquitoes can identify their hosts without artificial light, the selection of light as an attractive cue was an arbitrary one and future exploration of an alternative cue or a combination of attractive cues that could work best in the presence of a spatial repellent is warranted. It is also possible that the odour bait can be made more attractive by varying the composition or concentration of the chemical components.

In Chapter 3 it is demonstrated that mosquito catches in the odour-baited trap were reduced significantly in the presence of the transfluthrin repellent as compared with the absence of this repellent, while mosquito catches in the CDC UV light trap remained relatively similar both in the absence and presence of this repellent.MB5-baited traps may be unsuitable for use in close vicinity to a spatial repellent in a push-pull device as the spatial repellent may interfere with the mosquito's perception of synthetic lures. Positioning the traps further from the repellent source may improve the trapping efficiency in an area where mosquitoes can detect and respond to the synthetic lures dispensed by the trap. UV light traps might be more efficient and potentially easier to manage with constant trapping potential even in close proximity of a spatial repellent; though in my study the proportion of trapped mosquitoes was too low to consider this a potential pull component. Since both trap types were not very efficient in trapping mosquitoes in the presence of a spatial repellent, especially in the absence of a human, there is an urgent need to develop more attractive traps that can be combined with spatial repellents in a push-pull setup to improve efficacy.

Host-seeking mosquitoes constantly fly upwind towards a source of carbon dioxide and only perceive additional attractive odours and other cues once in proximity of the host (Spitzen *et al.*, 2013; Cardé, 2015). Entering a repellent plume may momentarily interrupt the tracking of odour cues but by flying randomly in and out of this plume, a mosquito could once again determine the source of the attractive odour (Cardé *et al.*, 2010; Achee *et al.*, 2012; Hawkes *et al.*, 2017). By positioning the trap further from the repellent plume, like 7.5 m or 10 m, it may improve the location of the source of the synthetic lure by the host-seeking mosquito and improve the overall impact of the push-pull system. This hypothesis may be corroborated by the fact that momentary exposure to the repellent plume did not have any effect on the proportion of mosquitoes caught in either the odour-baited trap or CDC UV light trap.

Mosquitoes have species-specific responses to chemical compounds, demonstrated in various studies (Afify *et al.*, 2020; Martin *et al.*, 2020; Nararak *et al.*, 2020). Findings in Chapter 3 support the differential response of sibling species to repellents and attractants and provide an indication of how highly specific chemosensory perception is among closely related mosquito species. This creates both obstacles and opportunities in vector control towards ensuring that tools generated are not only effective against one target species but could potentially work against vectors having different physiological traits to achieve a broad applicability.

Less is more: repellent-treated eave strip as a substitute for eave screen for indoor and outdoor protection of malaria mosquitoes

Open eaves have been shown to contribute significantly to the house-entry of malaria mosquitoes with subsequent increase in malaria transmission and by simply sealing eaves, the risk is eliminated (Njie *et al.*, 2009; Ogoma *et al.*, 2010; Mburu *et al.*, 2018). Improvement on house structure to ensure that entry ways used by mosquitoes to access houses have been either completely sealed or reduced substantially have been shown to reduce populations of malaria vectors indoors (Kaindoa *et al.*, 2018; Mburu *et al.*, 2018; Ngadjeu *et al.*, 2020). Further evidence of this was seen in Chapter 4 when untreated screens on eaves significantly reduced indoor biting populations of malaria vectors with no anticipated effect on outdoor biting mosquitoes. This was expected as the untreated screens provided a physical barrier that prevented mosquitoes from accessing the house confirming the importance of sealed eaves in malaria control (Menger *et al.*, 2016; Tusting *et al.*, 2017; Getawen *et al.*, 2018).

Passive release of spatial repellents enables utilization in endemic areas with minimal need for retreatment or regular compliance by users and no requirement for provision of electrical power that may not be available or affordable in some rural settings (Ogoma *et al.*, 2017; Moshi *et al.*, 2018). Impregnation of hessian cloth with transfluthrin and utilization as an eave fabric applies low technology using locally available natural fibre and provides effective protection for up to six months; both indoors and outdoors (Ogoma *et al.*, 2012; Govella *et al.*, 2015; Masalu *et al.*, 2017).

In Chapter 4 it was shown that passive release of transfluthrin from the eave fabric provided protection from indoor-biting mosquitoes in semi-field conditions. There was an indication that temperature played a role in protection conferred outdoors, with more protection being seen when ambient temperatures were higher and a reduction of protection being seen at lower temperatures. The effect that temperature had on protection conferred by transfluthrin was cited in a study by Ogoma *et al.*(2017) who stated that when

preceding daily temperatures fell below 23 °C, protection conferred by transfluthrin was greatly reduced(Ogoma *et al.*, 2017).

Presentation of the treated eave fabric as a strip (partial covering) as opposed to a screen (full coverage) significantly reduced both the number of mosquito landings indoors and outdoors. The interesting observation was that despite the remaining eave gaps, transfluthrin released from the eave strip was sufficient to disrupt host-vector interaction both indoors and outdoors, to a similar extent as treated eave screens. The advantage of using eave strips as opposed to eave screens would potentially be lower costs if scaled up since the amount of fabric used is less than half that used in the screening and since perfect fitting for each house is not a requirement, it results in ease of application without the need for customization per house and reduced skilled application(Lindsay *et al.*, 2003; Njie *et al.*, 2009; Ogoma *et al.*, 2010).

Pre-evaluation of push-pull mosquito control under natural conditions in western Kenya

Due to evidence obtained in the semi-field conditions as described in Chapter 2, a field experiment was conducted in western Kenya to test the performance of the transfluthrin/MB5 push pull system under natural conditions and to determine how the separate components of the tool perform in comparison to when combined as a way to gather research-based evidence for their recommendation and to direct future research and large-scale trials(Takken, 2010; Menger *et al.*, 2015; Mmbando *et al.*, 2019).As semi-field experiments were limited to only two sibling mosquito species (*An. arabiensis* and *An. gambiae s.s.*), the field set up was specifically designed to investigate how mosquitoes belonging to diverse mosquito genera and species would respond to the push-pull set up. Ultimately, products developed for vector control need to be applied in natural environments and so their performance in the field informs this application.

In Chapter 5 it was shown that significantly fewer malaria mosquitoes were caught indoors in the presence of both the push-pull and push only set ups indicating protection achieved by their application. Indoor transmission of malaria still accounts for the majority of malaria transmission (Bayoh *et al.*, 2014) and so additional protection from indoor-biting malaria vectors can contribute to the reduction in malaria prevalence. However, the increasing role of outdoor malaria transmission remains unaddressed by the transfluthrin/MB5 combination of the push-pull set up and would therefore warrant additional research to remedy this.

A previous study done in the same area showed a reduction in number of indoor-caught mosquitoes with all the applied treatments (repellent only – delta undecalactone, attractant only – MB5-baited MMX trap and the two treatments combined) (Menger *et al.*, 2015). Our study showed a similar trend with the

significant reduction in *Anopheles* species indoors in the presence of the repellent only and when combined with the attractant, but there was no effect with the attractant only(Jawara *et al.*, 2009; Menger *et al.*, 2015).

The odour-baited trap in our study had previously been used for mass-trapping mosquitoes in western Kenya that led to significant reductions in *An. funestus* populations and consequently, reduction in malaria cases (Homan *et al.*, 2016). The possibility of a species-specific response to the odour-baits is plausible as in our study a slight reduction in *An. funestus* populations was seen indoors in the presence of the MB5-baited Suna trap but not in any of the other mosquito species. If the study had continued for a longer period and with more households as was the case in the Homan *et al.* (2016) study, it may be possible that this would have led to significant reductions in *An. funestus* numbers (Homan *et al.*, 2016). As currently the most abundant malaria vector is *An. funestus*, reductions in its numbers would positively contribute to efforts in malaria control, especially in western Kenya where indoor transmission still accounts for most of malaria transmission (Degefa *et al.*, 2017; Ogola *et al.*, 2018).

Another study done in Central America evaluating combining indoor applied transfluthrin with an outdoor odour-baited CDC light trap demonstrated a species-specific reduction in indoor entry of *Anopheles* species, especially due to the repellent only (Wagman *et al.*, 2015). While the results of this study closely mirror ours, the diminished role that push-pull played in indoor entry of mosquitoes was more evident in the Wagman *et al.* (2015) study than it was in ours.

Positioning of the push-pull components appears to contribute positively to the outcome based on comparisons made with a recent field study done in Tanzania that showed a reduction in outdoor-biting by *An. arabiensis* when the repellent was actively emitted at least 5 m from the house and the pull component positioned at least 10 m from the repellent(Mmbando *et al.*, 2017).Preliminary results from Chapter 3 in this thesis indicate a possible interference in performance by odour-baited traps in the presence of a repellent, especially when in very close proximity of the repellent plume detected in Chapter 2. By positioning the trap further away from this plume, this may enhance its overall contribution to the push-pull strategy.

Another major difference between the Tanzania study and our study was the active release of transfluthrin using powered fans, which could have facilitated increased aerial distribution of transfluthrin that may have increased protection conferred(Mmbando *et al.*, 2017). While the use of active emanators might be a limiting factor due to unavailability of electric power in many malaria endemic areas in sub-Saharan Africa as well as high cost of the equipment especially for large scale application, it may be interesting to consider additional sources of cost-effective passively-emitted repellent for example from baskets and decorations placed inside or near the homes in addition to the eave fabric as an alternative to further improve the outcome(Masalu *et al.*, 2018).

Non-malaria vectors in our study were more responsive to the transfluthrin/MB5 set up outdoors as their proportional landing catches were significantly reduced both in the push-pull set up and in the presence of the repellent only. A number of studies have suggested that both *Culex* and *Mansonia* species are not very host-specific in their quest for blood meals given their non-specific host selection (Muturi *et al.*, 2008; Garcia-Rejon *et al.*, 2010; Telang *et al.*, 2019). These species could possibly be reduced by this tool to improve control of transmission of arboviruses and endophagic filariasis and to reduce nuisance-biting by mosquitoes. Although these mosquito species also readily enter houses and would probably equally have been affected by the transfluthrin, in the present study their densities were too low for a statistical evaluation.

Future perspectives

The findings in this thesis provide several opportunities to further develop and improve the control of malaria mosquitoes and present possible advancements in addressing hurdles in malaria control.

Manipulating odour-orientation of malaria vectors in the peri-domestic space using push-pull strategy

While the importance of outdoor transmission of malaria is not questioned(Degefa *et al.*, 2017; Sougoufara *et al.*, 2020), development of vector control tools targeting outdoor-biting mosquito populations has not yet resulted in inclusion in malaria control programmes(WHO, 2019). The use of transfluthrin as a spatial repellent passively emanating from eave fabric reduced the number of mosquito bites outdoors in semi-field conditions (Chapters 2 and 4) but not in natural conditions (Chapter 5). Likely reasons for this were the requirement of higher temperatures required to volatilize transfluthrin that are usually lacking during the hours when the vectors are active in various endemic areas(Ogoma *et al.*, 2017).

One remedy would be to use a spatial repellent that volatilizes more efficiently at lower temperatures occurring in the evenings and night in many endemic areas, and is preferably a non-pyrethroid compound to avoid the contribution to selection of multiple resistant genes when used over time(Guedes *et al.*, 2017). One possible spatial repellent would be delta-undecalactone that had shown promise when used as an eave fabric both in semi-field (Menger *et al.*, 2014; Menger *et al.*, 2016) and in field assays(Menger *et al.*, 2015). Other volatile organic compounds that present a similar repellent effect on mosquitoes as well as possible blends of non-pyrethroid chemicals could also be considered for optimized testing and application that could also affect pyrethroid resistant mosquitoes. In addition to their application on eave

fabric, the integration of additional points of emissions for the spatial repellents such as sisal baskets and functional hangings positioned further from the house could be explored to increase the space in which the repellent concentration is above the effective threshold in the peri-domestic area.

Another improvement on the present study would be the microencapsulation of the spatial repellent as opposed to impregnation (Miro Specos *et al.*, 2016). Transfluthrin in this thesis was impregnated into hessian strips (Ogoma *et al.*, 2012; Ogoma *et al.*, 2017)which, while it was not tested for association with experimental outcomes, could present an issue in the large scale roll out of this tool. By inclusion of the repellent as microcapsules in the eave fabric, the preparation of the repellent prior to use of fabric in the field could be commercialized for scaling up with uniformity of repellent load per fabric guaranteed(Menger *et al.*, 2015; Miro Specos *et al.*, 2016; Misni *et al.*, 2017). An unfortunate downside to microencapsulation is that, being a high-tech application, it may potentially increase both the short and long-term costs in large scale application unless a more affordable option is generated that would be just as effective and safe for use(Misni *et al.*, 2017).

Attractive pull component

The pull component of the set ups tested in this study did not confer protection of human hosts indoors or outdoors, in field or semi-field conditions. An attempt to replace the odour-baited trap with a light trap as described in Chapter 4 did not improve the trapping efficiency in the presence of the spatial repellent, transfluthrin.

As a way forward, an improved trap, possibly an odour-baited one with additional host cues such as warmth and humidity (Hawkes *et al.*, 2017; Cribellier *et al.*, 2020)may be considered while also exploring its placement further away from the repellent plume – possibly between 7.5 m and 20 m in the direction of breeding habitats for more efficient trapping of mosquitoes either separately or in combination with the spatial repellent (Okumu *et al.*, 2010). Additionally, the placement of the pull component could be explored to provide community protection as opposed to household protection (Homan *et al.*, 2016) to further improve protection and potentially lower costs.

Protective eave fabric

In this thesis, transfluthrin was used on eave fabric and presented either as an eave strip (partial covering) or eave screen (full covering). Results obtained in this study (Chapter 4) provide evidence of similar effectiveness when using eave strips to replace eave screens with comparable protection obtained both indoors and outdoors(Govella *et al.*, 2015; Mmbando *et al.*, 2018; Mwanga *et al.*, 2019a).

As a way forward, eave fabric could be applied as repellent-treated strips and offer the same level of protection as treated eave screens. Ultimately, improvement of housing would be ideal (Atieli *et al.*, 2009; Tusting *et al.*, 2017; Kaindoa *et al.*, 2018; Mburu *et al.*, 2018) though in low income areas where this may not be an immediate solution, the eave strips will provide adequate and easy to apply protection to both users and non-users (Mwanga *et al.*, 2019a).

Conclusions

In conclusion, this thesis has shown that i) repellent-treated eave strips provide protection from malaria vectors; both indoors and outdoors, with possibilities of significant cost-effectiveness in scale-up exercises compared to repellent-treated eave screens, ii) push-pull control provides an opportunity for outdoor control of nuisance-biting non-malaria mosquitoes and possibly reduce transmission of arboviruses and lymphatic filariasis, iii) the current push-pull strategy is not yet effective against malaria vectors outdoors under natural conditions and iv) a more effective attract and kill tool against host-seeking malaria vectors than the ones tested in my study would possibly improve the protection conferred by the push-pull mosquito control tool.

Future research should focus on a) identification and utilization of potent volatile compounds with high vapour pressure whose application is not restricted to climatic conditions specific for certain geographic regions, b) diversifying from odour-baited traps as attractants to more robust multi-cue traps to further optimize protection of susceptible human hosts outdoors and c) optimizing combination of the spatial repellent with a very attractive mosquito trap to achieve a maximum synergistic performance by investigating the placement geometry and d) possibilities for community effect push-pull set ups could be explored where design is done such that some households employ the repellent only while others strategically utilize attractant-baited traps for possible improvement of outdoor protection.

Summary

Summary

The urgency to develop new tools to supplement the currently existing ones; namely LLINs and IRS, has been fuelled by recent stagnation in reducing prevalence and incidence of malaria especially in sub-Saharan Africa. This stagnation has been attributed to increasing insecticide resistance in major malaria vectors and changes in mosquito populations; from predominately endophilic and anthropophilic to exophilic and opportunistic feeders. The rising importance of outdoor-biting malaria vectors in maintaining residual malaria transmission has warranted the development of control tools that target them while supplementing the existing tools. Due to the mainly indoor protection conferred by LLINs and IRS, development of new control tactics is required to target mosquito populations that bite and rest predominately outdoors. Tools previously developed to potentially target malaria mosquitoes when used singly were experimentally combined towards synergizing their protective efficacy outdoors. Spatial repellents have in recent years received renewed interest due to their ability to prevent host-vector contact and can be used to control outdoor-biting mosquitoes and in many cases, act on mosquitoes shown to be insecticide resistant. When used in combination with odour-baited traps, spatial repellents present opportunities for mosquito control by applying a proposed 'push-pull' strategy in controlling outdoor-biting mosquitoes that could potentially contribute to the control of residual transmission.

In order to understand the dynamics that a push-pull mosquito control strategy would involve, research questions were studied; 1) Does the inclusion of an optimally-performing spatial repellent in an eave fabric reduce outdoor-biting rates of malaria vectors? 2) Does the addition of an optimized odour-baited trap to the spatial repellent improve on any protection conferred? 3) Are the airborne concentrations of the chemical components of the spatial repellent and odour trap quantifiable towards determining effective range of possible protection?

In **Chapter 2**, I outlined systematic investigations of different concentrations of PMD and transfluthrin as potential spatial repellents applied on eave fabric and their protective efficacy on unprotected persons in the peri-domestic space outdoors quantified in semi field systems in western Kenya. Two concentrations of PMD tested did not offer any protection while two concentrations of transfluthrin were both protective, with the higher concentration offering more protection than the lower one, confirming the spatial repellency effect of transfluthrin on *Anopheles arabiensis*. Supplementation of transfluthrin with MB5-baited Suna trap did not have any effect on the protective efficacy seen. The MB5-baited Suna trap on its own did not divert mosquitoes from an unprotected human confirming that an unprotected human remains more attractive to host-seeking malaria vectors than to synthetic lures. Utilization of 2-butanone as a supplement to MB5-baited Suna was not comparable to addition of carbon dioxide indicating that 2-

Summary

butanone is not an adequate carbon dioxide mimic in attracting *An. arabiensis*. Quantification of the airborne concentrations of the push-pull chemical components showed that transfluthrin was detectable within a 5 m radius from the eave fabrics applied on experimental houses, with higher concentrations being detected nearest the house and closer to the ground with reducing concentrations further away and higher up in the treated space.

In **Chapter 3**, I explored the influence that transfluthrin as a spatial repellent has on the trapping efficiency of a MB5-baited Suna trap compared to a CDC UV light trap as a basis of considering alternative pull components in the push-pull system. In the absence of transfluthrin, the odour-baited Suna trap caught more *An. arabiensis* and *An. gambiae s.s.* than the CDC UV light trap indicative of the strong and specific attractive lure offered by the odour traps that is absent in the light trap. In the presence of transfluthrin, a significant reduction in the catches was seen in the odour trap which was indicative of the effect that transfluthrin had on the mosquito perception of synthetic lures from the odour trap. Trapping efficiency of the light trap remained fairly constant even in the presence of transfluthrin as mosquito perception of light remained unaffected in the presence of spatial repellents. The efficiency of the light trap was generally quite low and as such, use of a light trap as a potential pull component could not be considered.

In **Chapter 4**, I investigated the plausibility of presenting transfluthrin on an eave fabric that partially covered the eave gap as a strip instead of the traditional screen that requires complete coverage of the eave gap. Transfluthrin-treated eave strips provided similar protection to unprotected persons, both indoors and outdoors, as transfluthrin-treated eave screens providing a more cost-effective solution especially when scaled-up due to the utilization of less fabric and increasing the ease of application. Additionally, my investigations were able to reiterate the importance of eave screening as application of an untreated eave screen significantly reduced indoor-biting mosquitoes.

In **Chapter 5**, I carried out field evaluations of the push-pull strategy in a village in western Kenya to establish its performance in natural conditions. The application of the transfluthrin-treated eave fabric on its own significantly reduced the number of *An. gambiae* s.l. and *An. funestus* s.l. caught indoors. The inclusion of an odour-baited Suna trap did not have an additional effect on the protection conferred by transfluthrin indoors. Non-malaria vectors were significantly reduced outdoors, both in the presence of the repellent alone or repellent and odour-baited trap. None of the primary malaria vectors had a difference in their catches in the presence of any of the treatments applied.

The final chapter of this thesis provided a general discussion of the observations seen in the various chapters as well as conclusions obtained.

Summary

In summary, push-pull strategy presents a tool that can control outdoor-biting mosquitoes and may need to be optimized to effect a protective action against malaria vectors in various geographical and climatic conditions. Also, presentation of spatial repellents in an eave fabric that does not fully cover the eave gap is just as protective as a repellent-treated eave screen and could present opportunity for cost-effective control of malaria in areas where improvement of housing is not immediately possible. Generation of a more effective attract and kill tool that can supplement transfluthrin would improve the overall performance of the push-pull strategy and in the long run, identification of an even better spatial repellent that is effective against both malaria and non-malaria vectors in various geographical conditions would be ideal in ensuring effective disease control.

- Achee, N. L., Bangs, M. J., Farlow, R., Killeen, G. F., Lindsay, S., Logan, J. G., Moore, S. J., Rowland, M., Sweeney, K., Torr, S. J., Zwiebel, L. J. & Grieco, J. P. (2012) Spatial repellents: from discovery and development to evidence-based validation. *Malaria Journal*, 11, 164.
- Achee, N. L., Masuoka, P., Smith, P., Martin, N., Chareonviryiphap, T., Polsomboon, S., Hendarto, J. & Grieco, J. (2012) Identifying the effective concentration for spatial repellency of the dengue vector *Aedes aegypti. Parasites and Vectors*, 5, 300.
- Achee, N. L., Sardelis, M. R., Dusfour, I., Chauhan, K. R. & Grieco, J. P. (2009) Characterization of spatial repellent, contact irritant, and toxicant chemical actions of standard vector control compounds. *Journal of the American Mosquito Control Association*, 25, 156-167.
- Afify, A. & Potter, C. J. (2020) Insect repellents mediate species-specific olfactory behaviours in mosquitoes. *Malaria Journal*, 19, 127.
- Agossa, F. R., Gnanguenon, V., Anagonou, R., Azondekon, R., Aïzoun, N., Sovi, A., Oké-Agbo, F., Sèzonlin, M. & Akogbéto, M. C. (2015) Impact of Insecticide Resistance on the Effectiveness of Pyrethroid-Based Malaria Vectors Control Tools in Benin: Decreased Toxicity and Repellent Effect. *PLoS ONE*, **10**, e0145207.
- Allan, S. A., Day, J. F. & Edman, J. D. (1987) Visual ecology of biting flies. *Annu Rev Entomol*,**32**, 297-316.
- Andres, M., Lorenz, L. M., Mbeleya, E. & Moore, S. J. (2015) Modified mosquito landing boxes dispensing transfluthrin provide effective protection against *Anopheles arabiensis*mosquitoes under simulated outdoor conditions in a semi-field system. *Malaria Journal*, 14, 255.
- Antonio-Nkondjio, C., Kerah, C. H., Simard, F., Awono-Ambene, P., Chouaibou, M., Tchuinkam, T. & Fontenille, D. (2006) Complexity of the malaria vectorial system in Cameroon: contribution of secondary vectors to malaria transmission. *Journal of Medical Entomology*,43, 1215-1221.
- Asmare, Y., Hill, S. R., Hopkins, R. J., Tekie, H. & Ignell, R. (2017) The role of grass volatiles on oviposition site selection by *Anopheles arabiensis* and *Anopheles coluzzii*. *Malaria Journal*, 16, 65-65.
- Atieli, H., Menya, D., Githeko, A. & Scott, T. (2009) House design modifications reduce indoor resting malaria vector densities in rice irrigation scheme area in western Kenya. *Malaria Journal*, 8, 108.
- Awolola, T. S., Adeogun, A., Olakiigbe, A. K., Oyeniyi, T., Olukosi, Y. A., Okoh, H., Arowolo, T., Akila, J., Oduola, A. & Amajoh, C. N. (2018) Pyrethroids resistance intensity and resistance mechanisms in *Anopheles gambiae* from malaria vector surveillance sites in Nigeria. *PLoS ONE*, 13, e0205230.

- Ballinger, G. A. (2004) Using Generalized Estimating Equations for Longitudinal Data Analysis. *Organizational Research Methods*, 7, 127-150.
- Barasa, S. S., Ndiege, I. O., Lwande, W. & Hassanali, A. (2002) Repellent activities of stereoisomers of p-Menthan-3,8-diols against *Anopheles gambiae* (Diptera: Culicidae). *Journal of Medical Entomology*, 39, 736-741.
- Barnard, D. R. (2005) Biological assay methods for mosquito repellents. *Journal of the American Mosquito Control Association*, 21, 12-16.
- Barnard, D. R., Bernier, U. R., Posey, K. H. & Xue, R. D. (2002) Repellency of IR3535, KBR3023, paramenthane-3,8-diol, and Deet to Black Salt Marsh Mosquitoes (Diptera: Culicidae) in the Everglades National Park. *Journal of Medical Entomology*, **39**, 895-899.
- Batista, E. P. A., Ngowo, H. S., Opiyo, M., Shubis, G. K., Meza, F. C., Okumu, F. O. & Eiras, A. E. (2017) Semi-field assessment of the BG-Malaria trap for monitoring the African malaria vector, *Anopheles arabiensis. PLoS ONE*, **12**, e0186696.
- Batista, E. P. A., Ngowo, H., Opiyo, M., Shubis, G. K., Meza, F. C., Siria, D. J., Eiras, A. E. & Okumu, F. O. (2018) Field evaluation of the BG-Malaria trap for monitoring malaria vectors in rural Tanzanian villages. *PLoS ONE*, 13, e0205358.
- Bayoh, M. N., Walker, E. D., Kosgei, J., Ombok, M., Olang, G. B., Githeko, A. K., Killeen, G. F., Otieno, P., Desai, M., Lobo, N. F., Vulule, J. M., Hamel, M. J., Kariuki, S. & Gimnig, J. E. (2014) Persistently high estimates of late night, indoor exposure to malaria vectors despite high coverage of insecticide treated nets. *Parasites and Vectors*, 7, 380.
- Benedict, M. Q. (1997) Care and maintenance of anopheline mosquito colonies. In: Crampton J.M., Beard C.B., Louis C. (eds) *The Molecular Biology of Insect Disease Vectors. Springer, Dordrecht.*
- Benelli, G. & Beier, J. C. (2017) Current vector control challenges in the fight against malaria. *Acta Tropica*, **174**, 91-96.
- Bhatt, S., Weiss, D. J., Cameron, E., Bisanzio, D., Mappin, B., Dalrymple, U., Battle, K., Moyes, C. L., Henry, A., Eckhoff, P. A., Wenger, E. A., Briet, O., Penny, M. A., Smith, T. A., Bennett, A., Yukich, J., Eisele, T. P., Griffin, J. T., Fergus, C. A., Lynch, M., Lindgren, F., Cohen, J. M., Murray, C. L. J., Smith, D. L., Hay, S. I., Cibulskis, R. E. & Gething, P. W. (2015) The effect of malaria control on *Plasmodium falciparum* in Africa between 2000 and 2015. *Nature*, 526, 207-211.
- Bhatt, S., Weiss, D. J., Mappin, B., Dalrymple, U., Cameron, E., Bisanzio, D., Smith, D. L., Moyes, C. L., Tatem, A. J., Lynch, M., Fergus, C. A., Yukich, J., Bennett, A., Eisele, T. P., Kolaczinski, J., Cibulskis, R. E., Hay, S. I. & Gething, P. W. (2015) Coverage and system efficiencies of insecticide-treated nets in Africa from 2000 to 2017. *eLIFE*,4, e09672.

- Bibbs, C. S., Tsikolia, M., Bloomquist, J. R., Bernier, U. R., Xue, R. D. & Kaufman, P. E. (2018) Vapor toxicity of five volatile pyrethroids against *Aedes aegypti*, *Aedes albopictus*, *Culex quinquefasciatus*, and *Anopheles quadrimaculatus* (Diptera: Culicidae). *Pest Management Science*, 74, 2699-2706.
- Bidlingmayer, W. L. (1964) The effect of moonlight on the flight activity of mosquitoes. *Ecology*, 45, 87-94.
- Birget, P. L. G. & Koella, J. C. (2015) A genetic model of the effects of insecticide-treated bed nets on the evolution of insecticide-resistance. *Evolution, Medicine, and Public Health*, 2015, 205-215.
- Birhanu, A., Asale, A. & Yewhalaw, D. (2019) Bio-efficacy and physical integrity of piperonyl butoxide coated combination net (PermaNet((R)) 3.0) against pyrethroid resistant population of *Anopheles* gambiaes.l. and Culex quinquefasciatus mosquitoes in Ethiopia. Malaria Journal, 18, 224.
- Bohbot, J. D. & Dickens, J. C. (2010) Insect repellents: modulators of mosquito odorant receptor activity. *PLoS ONE*,**5**, e12138.
- Bowman, N. M., Akialis, K., Cave, G., Barrera, R., Apperson, C. S. & Meshnick, S. R. (2018) Pyrethroid insecticides maintain repellent effect on knock-down resistant populations of *Aedes aegypti* mosquitoes. *PLoS ONE*, 13, e0196410-e0196410.
- Browne, S. M. & Bennett, G. F. (1981) Response of mosquitoes (Diptera: Culicidae) to visual stimuli. *Journal of Medical Entomology*, **18**, 505-521.
- Buse, J., Robinson, J. L., Shyne, R., Chi, Q., Affleck, D., Duce, D. & Seiden-Long, I. (2019) Rising above helium: A hydrogen carrier gas chromatography flame ionization detection (GC-FID) method for the simultaneous quantification of toxic alcohols and ethylene glycol in human plasma specimens. *Clinical Biochemistry*, **73**, 98-104.
- Cardé, R. T. (2015) Multi-Cue Integration: How Female Mosquitoes Locate a Human Host. Current Biology, 25, R793-795.
- Cardé, R. T. & Gibson, G. (2010a) Host finding by female mosquitoes: mechanisms of orientation to host odours and other cues. In *Olfaction in Vector-Host Interactions* (ed. by W. Takken & B. G. J. Knols), pp. 115-141. Wageningen Academic Publishers, Wageningen.
- Cardé, R. T. & Gibson, G. (2010b) Long-distance orientation of mosquitoes to host odours and other host-related cues. *Ecology of Vector-Borne Diseases*, 2, 115–141.
- Carroll, S. P. & Loye, J. (2006) PMD, a registered botanical mosquito repellent with deet-like efficacy. *Journal of the American Mosquito Control Association*, **22**, 507-514.
- Carroll, S. P., Venturino, J. & Davies, J. H. (2019) A Milestone In Botanical Mosquito Repellents: Novel PMD-Based Formulation Protects More Than Twice As Long As High-Concentration Deet and Other Leading Products. *Journal of the American Mosquito Control Association*, **35**, 186-191.

- Chareonviriyaphap, T., Bangs, M. J., Suwonkerd, W., Kongmee, M., Corbel, V. & Ngoen-Klan, R. (2013) Review of insecticide resistance and behavioral avoidance of vectors of human diseases in Thailand. *Parasites and Vectors*, 6, 280.
- Chareonviriyaphap, T., Prabaripai, A. & Bangs, M. J. (2004) Excito-repellency of deltamethrin on the malaria vectors, Anopheles minimus, Anopheles dirus, Anopheles swadiwongporni and Anopheles maculatus, in Thailand. *Journal of the American Mosquito Control Association*, 20, 45-54.
- Charlwood, D. (2003) Did Herodotus describe the first airborne use of mosquito repellents? *Trends in Parasitology*, **19**.
- Churcher, T. S., Lissenden, N., Griffin, J. T., Worrall, E. & Ranson, H. (2016) The impact of pyrethroid resistance on the efficacy and effectiveness of bednets for malaria control in Africa. *eLIFE*,**5**.
- Cohnstaedt, L. W. & Allan, S. A. (2011) Effects of sublethal pyrethroid exposure on the host-seeking behavior of female mosquitoes. *Journal of Vector Ecology*, **36**, 395-403.
- Colucci, B. & Muller, P. (2018) Evaluation of Standard Field and Laboratory Methods to Compare Protection Times of the Topical Repellents PMD and DEET. *Scientific Reports*, **8**, 12578.
- Cook, J., Tomlinson, S., Kleinschmidt, I., Donnelly, M. J., Akogbeto, M., Adechoubou, A., Massougbodji, A., Okê-Sopoh, M., Corbel, V., Cornelie, S., Hounto, A., Etang, J., Awono-Ambene, H. P., Bigoga, J., Mandeng, S. E., Njeambosay, B., Tabue, R., Kouambeng, C., Fondjo, E., Raghavendra, K., Bhatt, R. M., Chourasia, M. K., Swain, D. K., Uragayala, S., Valecha, N., Mbogo, C., Bayoh, N., Kinyari, T., Njagi, K., Muthami, L., Kamau, L., Mathenge, E., Ochomo, E., Kafy, H. T., Bashir, A. I., Malik, E. M., Elmardi, K., Sulieman, J. E., Abdin, M., Subramaniam, K., Thomas, B., West, P., Bradley, J., Knox, T. B., Mnzava, A. P., Lines, J., Macdonald, M., Nkuni, Z. J. & Implications of Insecticide Resistance, C. (2018) Implications of insecticide resistance during a WHO-coordinated multi-country prospective study. *Parasites and Vectors*, 11, 550.
- Cook, S. M., Khan, Z. R. & Pickett, J. A. (2007) The use of push-pull strategies in integrated pest management *Annual Reviewof Entomology*, 52, 375-400.
- Corrêa, A. P. S. A., Galardo, A. K. R., Lima, L. A., Câmara, D. C. P., Müller, J. N., Barroso, J. F. S., Lapouble, O. M. M., Rodovalho, C. M., Ribeiro, K. A. N. & Lima, J. B. P. (2019) Efficacy of insecticides used in indoor residual spraying for malaria control: an experimental trial on various surfaces in a "test house". *Malaria Journal*, 18, 345.
- Costa-Neta, B. M., da Silva, A. A., Brito, J. M., Moraes, J. L. P., Rebelo, J. M. M. & Silvas, F. S. (2017) Light-emitting diode (LED) traps improve the light-trapping of anopheline mosquitoes. *Journal of Medical Entomology*,**54**, 1699-1703.

- Costantini, C., Gibson, G., Sagnon, N., Della Torre, A., Brady, J. & Coluzzi, M. (1996) Mosquito responses to carbon dioxide in a west African Sudan savanna village. *Medical and Veterinary Entomology, Jul*; **10**(3), 220-227.
- Cribellier, A., Spitzen, J., Fairbairn, H., van de Geer, C., van Leeuwen, J. L. & Muijres, F. T. (2020) Lure, retain, and catch malaria mosquitoes. How heat and humidity improve odour-baited trap performance. *Malaria Journal*, **19**, 357.
- De Keyser, R., Cassidy, C., Laban, S., Gopal, P., Pickett, J. A., Reddy, Y. K., Prasad, M., Prasad, G., Chirukandoth, S., Senthilven, K., Carpenter, S. & Logan, J. G. (2017) Insecticidal effects of deltamethrin in laboratory and field populations of Culicoides species: how effective are hostcontact reduction methods in India? *Parasites and Vectors*, 10, 54.
- Degefa, T., Yewhalaw, D., Zhou, G., Atieli, H., Githeko, A. K. & Yan, G. (2020) Evaluation of humanbaited double net trap and human-odour-baited CDC light trap for outdoor host-seeking malaria vector surveillance in Kenya and Ethiopia. *Malaria Journal*, **19**, 174.
- Degefa, T., Yewhalaw, D., Zhou, G., Lee, M. C., Atieli, H., Githeko, A. K. & Yan, G. (2017) Indoor and outdoor malaria vector surveillance in western Kenya: implications for better understanding of residual transmission. *Malaria Journal*, 16, 443.
- Dekker, T. & Carde, R. T. (2011) Moment-to-moment flight manoeuvres of the female yellow fever mosquito (Aedes aegypti L.) in response to plumes of carbon dioxide and human skin odour. *Journal of Experimental Biology*, **214**, 3480-3494.
- Dekker, T., Geier, M. & Carde, R. T. (2005) Carbon dioxide instantly sensitizes female yellow fever mosquitoes to human skin odours. *Journal of Experimental Biology*, 208, 2963-2972.
- Dekker, T., Takken, W. & Cardé, R. T. (2001) Structure of host-odour plumes influences catch of Anopheles gambiae s.s. and Aedes aegypti in a dual-choice olfactomer. Physiological Entomology, 26(2), 124-134.
- Deletre, E., Martin, T., Duménil, C. & Chandre, F. (2019) Insecticide resistance modifies mosquito response to DEET and natural repellents. *Parasites and Vectors*, **12**, 89.
- Dennis, E. J., Goldman, O. V. & Vosshall, L. B. (2019) Aedes aegypti Mosquitoes Use Their Legs to Sense DEET on Contact. Current Biology, 29, 1551-1556 e1555.
- Dia, I., Diallo, D., Duchemin, J. B., Ba, Y., Konate, L., Costantini, C. & Diallo, M. (2005) Comparisons of human-landing catches and odor-baited entry traps for sampling malaria vectors in Senegal. *Journal of Medical Entomology*, 42, 104-109.

- Diaz, J. H. (2016) Chemical and Plant-Based Insect Repellents: Efficacy, Safety, and Toxicity. Wilderness & Environmental Medicine, 27, 153-163.
- Ditzen, M., Pellegrino, M. & Vosshall, L. B. (2008) Insect odorant receptors are molecular targets of the insect repellent DEET. *Science*, **319**, 1838-1842.
- Djouaka, R., Riveron, J. M., Yessoufou, A., Tchigossou, G., Akoton, R., Irving, H., Djegbe, I., Moutairou, K., Adeoti, R., Tamò, M., Manyong, V. & Wondji, C. S. (2016) Multiple insecticide resistance in an infected population of the malaria vector *Anopheles funestus* in Benin. *Parasites* and Vectors, 9, 453-453.
- Donnelly, B., Berrang-Ford, L., Ross, N. A. & Michel, P. (2015) A systematic, realist review of zooprophylaxis for malaria control. *Malaria Journal*, 14, 313.
- Dugassa, S., Lindh, J. M., Oyieke, F., Mukabana, W. R., Lindsay, S. W. & Fillinger, U. (2013) Development of a gravid trap for collecting live malaria vectors *Anopheles gambiae s.l. PLoS ONE*, 8, e68948-e68948.
- Durnez, L. & Coosemans, M. (2013) Residual Transmission of Malaria: An Old Issue for New Approaches.
- Eldakroory, S., Morsi, D. E., Abdel-Rahman, R., Roshdy, S., Gouida, M. & Khashaba, E. (2017) Correlation between toxic organochlorine pesticides and breast cancer. *Human & Experimental Toxicology*,**36**, 1326-1334.
- Evans, B. P., Clark, J. W., Barbara, K. A., Mundal, K. D., Furman, B. D., McAvin, J. C. & Richardson, J.
 H. (2009) Operational vector-borne disease surveillance and control: closing the capabilities gap through research at overseas military laboratories US Army Medical Department Journal, 2, 16-27.
- Feng, J., Zhang, F., Zhao, J., Guo, W. & Sun, J. (2018) An improved quantification method for 12 linear dimethylsiloxanes and 1 cyclic dimethylsiloxane in polydimethylsiloxane using gas chromatography-flame ionization detectorDevelopment strategy and accuracy. *Journal of Chromatography A*,1578, 112-116.
- Finda, M. F., Moshi, I. R., Monroe, A., Limwagu, A. J., Nyoni, A. P., Swai, J. K., Ngowo, H. S., Minja, E. G., Toe, L. P., Kaindoa, E. W., Coetzee, M., Manderson, L. & Okumu, F. O. (2019) Linking human behaviours and malaria vector biting risk in south-eastern Tanzania. PLoS ONE,14, e0217414-e0217414.
- Fornadel, C. M., Norris, L. C. & Norris, D. E. (2010) Centers for disease control light traps for monitoring *Anopheles arabiensis* human biting rates in an area with low vector density and high insecticide-treated bed net use. *America Journal of Tropical Medicine and Hygiene*,83.
- Foster, W. A. (2008) Phytochemicals as population sampling lures. *Journal of the American Mosquito Control Association*, **24**, 138-146, 139.

- Foster, W. A. & Takken, W. (2004) Nectar-related vs. human related volatiles: behavioural response and choice by female and male *Anopheles gambiae* (Diptera: Culicidae) between emergence and first feeding. *Bulletin of Entomological Research*, *Apr*;**94**(2), 145-157.
- Garcia-Rejon, J. E., Blitvich, B. J., Farfan-Ale, J. A., Lorono-Pino, M. A., Chi Chim, W. A., Flores-Flores, L. F., Rosado-Paredes, E., Baak-Baak, C., Perez-Mutul, J., Suarez-Solis, V., Fernandez-Salas, I. & Beaty, B. J. (2010) Host-feeding preference of the mosquito, Culex quinquefasciatus, in Yucatan State, Mexico. Journal of Insect Science (Online), 10, 32.
- Garrett-Jones, C., Magayuka, S. A. & Organization, W. H. (1975) Studies on the natural incidence of *Plasmodium* and *Wuchereria* infections in anopheles in rural East Africa. *World Health Organization*.
- Getawen, S. K., Ashine, T., Massebo, F., Woldeyes, D. & Lindtjorn, B. (2018) Exploring the impact of house screening intervention on entomological indices and incidence of malaria in Arba Minch town, southwest Ethiopia: A randomized control trial. *Acta Tropica*, **181**, 84-94.
- Ghaninia, M., Majeed, S., Dekker, T., Hill, S. R. & Ignell, R. (2019) Hold your breath Differential behavioral and sensory acuity of mosquitoes to acetone and carbon dioxide. *PLoS ONE*,14, e0226815.
- Ghebreyesus, T. A., Haile, M., Witten, K. H., Getachew, A., Yohannes, M., Lindsay, S. W. & Byass, P. (2000) Household risk factors for malaria among children in the Ethiopian highlands. *Transactions of The Royal Society of Tropical Medicine and Hygiene*, 94, 17-21.
- Gibson, G. & Torr, S. J. (1999) Visual and olfactory responses of haematophagous Diptera to host stimuli. *Medical and Veterinary Entomology*, **13**(1), 2-23.
- Gillies, M. T. (2009) The role of carbon dioxide in host-finding by mosquitoes (Diptera: Culicidae): a review. *Bulletin of Entomological Research*, **70**, 525-532.
- Gillies, M. T. & Coetzee, M. (1987) A supplement to the Anophelinae of Africa South of the Sahara. South African Institute for Medical Research, Johannesburg.
- Gleave, K., Lissenden, N., Richardson, M., Choi, L. & Ranson, H. (2018) Piperonyl butoxide (PBO) combined with pyrethroids in insecticide-treated nets to prevent malaria in Africa. *The Cochrane Database of Systematic Reviews*, **11**, CD012776.
- Glunt, K. D., Paaijmans, K. P., Read, A. F. & Thomas, M. B. (2014) Environmental temperatures significantly change the impact of insecticides measured using WHOPES protocols. *Malaria Journal*, 13, 350-350.
- González, M., Alarcón-Elbal, P. M., Valle-Mora, J. & Goldarazena, A. (2016) Comparison of different light sources for trapping Culicoides biting midges, mosquitoes and other dipterans. *Veterinary Parasitology*, 226, 44-49.

- Govella, N. J., Ogoma, S. B., Paliga, J., Chaki, P. P. & Killeen, G. (2015) Impregnating hessian strips with the volatile pyrethroid transfluthrin prevents outdoor exposure to vectors of malaria and lymphatic filariasis in urban Dar es Salaam, Tanzania. *Parasites and Vectors*, **8**, 322.
- Gressel, J. (2011) Low pesticide rates may hasten the evolution of resistance by increasing mutation frequencies. Pest Management Science, 67, 253-257.
- Grieco, J. P., Achee, N. L., Chareonviriyaphap, T., Suwonkerd, W., Chauhan, K., Sardelis, M. R. & Roberts, D. R. (2007) A new classification system for the actions of IRS chemicals traditionally used for malaria control. *PLoS ONE*, 2, e716.
- Grieco, J. P., Achee, N. L., Sardelis, M. R., Chuhan, K. R. & Roberts, D. R. (2005) A novel high-throuput screening system to evaluate the behavioral response of adult mosquitoes to chemicals. Journal of the American Mosquito Control Association,21, 404-411.
- Guedes, R. N. C., Walse, S. S. & Throne, J. E. (2017) Sublethal exposure, insecticide resistance, and community stress. *Current Opinion in Insect Science*, 21, 47-53.
- Guo, J., Shi, Y., Xu, C., Zhong, R., Zhang, F., Zhang, T., Niu, B. & Wang, J. (2016) Quantification of plasma myo-inositol using gas chromatography-mass spectrometry. *Clinica Chimica Acta*, 460, 88-92.
- Han, W., Tian, Y. & Shen, X. (2018) Human exposure to neonicotinoid insecticides and the evaluation of their potential toxicity: An overview. *Chemosphere*, 192, 59-65.
- Hancock, P. A., Hendriks, C. J. M., Tangena, J. A., Gibson, H., Hemingway, J., Coleman, M., Gething, P.
 W., Cameron, E., Bhatt, S. & Moyes, C. L. (2020) Mapping trends in insecticide resistance phenotypes in African malaria vectors. *PLoSBiology*, 18, e3000633.
- Hao, H., Wei, J., Dai, J. & Du, J. (2008) Host-Seeking and Blood-Feeding Behavior of Aedes albopictus (Diptera: Culicidae) Exposed to Vapors of Geraniol, Citral, Citronellal, Eugenol, or Anisaldehyde. *Journal of Medical Entomology*, 45, 533-539.
- Harwood, J. F., Richardson, A. G., Wright, J. A. & Obenauer, P. J. (2014) Field assessment of yeast- and oxalic Acid-generated carbon dioxide for mosquito surveillance. *Journal of the American Mosquito Control Association*, **30**, 275-283.
- Hasegawa, S., Azuma, M. & Takahashi, K. (2008) Enzymatic esterification of lactic acid, utilizing the basicity of particular polar organic solvents to suppress the acidity of lactic acid. *Journal of Chemical Technology and Biotechnology*, 83, 1503-1510.
- Hawkes, F. & Gibson, G. (2016) Seeing is believing: the nocturnal malarial mosquito *Anopheles coluzzii* responds to visual host-cues when odour indicates a host is nearby. *Parasites and Vectors*, **9**, 320.

- Hawkes, F. M., Dabire, R. K., Sawadogo, S. P., Torr, S. J. & Gibson, G. (2017) Exploiting *Anopheles* responses to thermal, odour and visual stimuli to improve surveillance and control of malaria. *Scientific Reports*, **7**, 17283.
- Hemingway, J. & Ranson, H. (2000) Insecticide resistance in insect vectors of human disease. Annual Review of Entomology, 45, 371-391.
- Hemingway, J., Shretta, R., Wells, T. N., Bell, D., Djimde, A. A., Achee, N. & Qi, G. (2016) Tools and Strategies for Malaria Control and Elimination: What Do We Need to Achieve a Grand Convergence in Malaria? *PLoSBiology*, 14, e1002380.
- Hershey, C. L., Florey, L. S., Ali, D., Bennett, A., Luhanga, M., Mathanga, D. P., Salgado, S. R., Nielsen, C. F., Troell, P., Jenda, G., Ye, Y. & Bhattarai, A. (2017) Malaria Control Interventions Contributed to Declines in Malaria Parasitemia, Severe Anemia, and All-Cause Mortality in Children Less Than 5 Years of Age in Malawi, 2000-2010. *The American Journal of Tropical Medicine and Hygiene*, 97, 76-88.
- Hill, N., Lenglet, A., Arnéz, A. M. & Carneiro, I. (2007) Plant based insect repellent and insecticide treated bed nets to protect against malaria in areas of early evening biting vectors: double blind randomised placebo controlled clinical trial in the Bolivian Amazon. *BMJ* **335**, 1023.
- Hill, N., Zhou, H. N., Wang, P., Guo, X., Carneiro, I. & Moore, S. J. (2014) A household randomized, controlled trial of the efficacy of 0.03% transfluthrin coils alone and in combination with longlasting insecticidal nets on the incidence of *Plasmodium falciparum* and *Plasmodium vivax* malaria in Western Yunnan Province, China. *Malaria Journal*, 13, 208.
- Hiscox, A., Homan, T., Mweresa, C. K., Maire, N., Di Pasquale, A., Masiga, D., Oria, P. A., Alaii, J., Leeuwis, C., Mukabana, W. R., Takken, W. & Smith, T. A. (2016) Mass mosquito trapping for malaria control in western Kenya: study protocol for a stepped wedge cluster-randomised trial. *Trials*, 17, 356.
- Hiscox, A., Otieno, B., Kibet, A., Mweresa, C. K., Omusula, P., Geier, M., Rose, A., Mukabana, W. R. & Takken, W. (2014) Development and optimization of the Suna trap as a tool for mosquito monitoring and control. *Malaria Journal*, 13, 257.
- Hodson, C. N., Yu, Y., Plettner, E. & Roitberg, B. D. (2016) New repellent effective against African malaria mosquito *Anopheles gambiae*: implications for vector control. *Medical and Veterinary Entomology*, 30, 369-376.
- Hogarh, J. N., Agyekum, T. P., Bempah, C. K., Owusu-Ansah, E. D. J., Avicor, S. W., Awandare, G. A., Fobil, J. N. & Obiri-Danso, K. (2018) Environmental health risks and benefits of the use of mosquito coils as malaria prevention and control strategy. *Malaria Journal*, 17, 265.

- Hogarh, J. N., Antwi-Agyei, P. & Obiri-Danso, K. (2016) Application of mosquito repellent coils and associated self-reported health issues in Ghana. *Malaria Journal*, **15**, 61.
- Homan, T., Hiscox, A., Mweresa, C. K., Masiga, D., Mukabana, W. R., Oria, P., Maire, N., Pasquale, A. D., Silkey, M., Alaii, J., Bousema, T., Leeuwis, C., Smith, T. A. & Takken, W. (2016) The effect of mass mosquito trapping on malaria transmission and disease burden (SolarMal): a stepped-wedge cluster-randomised trial. *The Lancet*, **388**, 1193-1201.
- Hu, X., Leming, M. T., Whaley, M. A. & O'Tousa, J. E. (2014) Rhodopsin coexpression in UV photoreceptors of *Aedes aegypti* and *Anopheles gambiae* mosquitoes. *Journal of Experimental Biology*, 217, 1003-1008.
- Huho, B., Briët, O., Seyoum, A., Sikaala, C., Bayoh, N., Gimnig, J. E., Okumu, F. O., Diallo, D., Abdulla, S., Smith, T. A. & Killeen, G. (2013) Consistently high estimates for the proportion of human exposure to malaria vector populations occurring indoors in rural Africa. *International Journal of Epidemiology*, 42, 235-247.
- Jatta, E., Jawara, M., Bradley, J., Jeffries, D., Kandeh, B., Knudsen, J. B., Wilson, A. L., Pinder, M., D'Alessandro, U. & Lindsay, S. W. (2018) How house design affects malaria mosquito density, temperature, and relative humidity: an experimental study in rural Gambia. *The Lancet. Planetary Health*, 2, e498-e508.
- Jawara, M., Smallegange, R. C., Jeffries, D., Nwakanma, D. C., Awolola, T. S., Knols, B. G., Takken, W.
 & Conway, D. J. (2009) Optimizing odor-baited trap methods for collecting mosquitoes during the malaria season in The Gambia. *PLoS ONE*, 4, e8167.
- Jensen, A. R., Spliid, N. H. & Svensmark, B. (2007) Determination of volatilization (dissipation) and secondary deposition of pesticides in a field study using passive dosimeters. *International Journal* of Environmental Analytical Chemistry, 87, 913-926.
- Jeyalakshmi, T., Shanmugasundaram, R., Kannadasan, J., Geetha, S., Saravanan, M. & Hilda, S. (2014) Efficacy of a commercial liquid vaporiser (Transfluthrin 0.88% (w/v)) under various room sizes against *Culex quinquefasciatus* Say *Journal of Entomology and Zoology Studies*, **2**, 220-224.
- Johnson, P. C., Barry, S. J., Ferguson, H. M. & Muller, P. (2015) Power analysis for generalized linear mixed models in ecology and evolution. *Methods in Ecology and Evolution*, 6, 133-142.
- Joshi, G. P., Fontaine, R. E. & Pradhan, G. D. (1975) The CDC battery-operated light trap for assessment of *An. gambiae* and *An. funestus* in a WHO stage VII insecticide trial, Kisumu, Kenya. *WHO/VBC*, **75**.
- Kaindoa, E. W., Finda, M., Kiplagat, J., Mkandawile, G., Nyoni, A., Coetzee, M. & Okumu, F. O. (2018) Housing gaps, mosquitoes and public viewpoints: a mixed methods assessment of relationships
between house characteristics, malaria vector biting risk and community perspectives in rural Tanzania. *Malaria Journal*, **17**, 298-298.

- Kampango, A., Bragança, M., Sousa, B. d. & Charlwood, J. D. (2013) Netting barriers to prevent mosquito entry into houses in southern Mozambique: a pilot study. *Malaria Journal*, 12, 99-99.
- Kawada, H., Ohashi, K., Dida, G. O., Sonye, G., Njenga, S. M., Mwandawiro, C. & Minakawa, N. (2014) Insecticidal and repellent activities of pyrethroids to the three major pyrethroid-resistant malaria vectors in western Kenya. *Parasites and Vectors*, 7, 208.
- Kayentao, K., Florey, L. S., Mihigo, J., Doumbia, A., Diallo, A., Koné, D., Doumbo, O. & Eckert, E. (2018) Impact evaluation of malaria control interventions on morbidity and all-cause child mortality in Mali, 2000-2012. *Malaria Journal*, 17, 424-424.
- Kenea, O., Balkew, M., Tekie, H., Gebre-Michael, T., Deressa, W., Loha, E., Lindtjorn, B. & Overgaard, H. J. (2017) Comparison of two adult mosquito sampling methods with human landing catches in south-central Ethiopia. *Malaria Journal*, 16, 30.
- Khan, Z., Midega, C. A., Hooper, A. & Pickett, J. (2016) Push-Pull: Chemical Ecology-Based Integrated Pest Management Technology. *Journal of Chemical Ecology*, 42, 689-697.
- Killeen, G. F. (2014) Characterizing, controlling and eliminating residual malaria transmission. *Malaria Journal*, **13**, 330.
- Killeen, G. F. & Moore, S. J. (2012) Target product profiles for protecting against outdoor malaria transmission. *Malaria Journal*, **11**, 17.
- King, A. M. & Aaron, C. K. (2015) Organophosphate and Carbamate Poisoning. *Emergency Medicine Clinics of North America*, 33, 133-151.
- Kirby, M. J., Ameh, D., Bottomley, C., Green, C., Jawara, M., Milligan, P. J., Snell, P. C., Conway, D. J. & Lindsay, S. W. (2009) Effect of two different house screening interventions on exposure to malaria vectors and on anaemia in children in The Gambia: a randomised controlled trial. *The Lancet*, 374, 998-1009.
- Kirby, M. J., Green, C., Milligan, P. M., Sismanidis, C., Jasseh, M., Conway, D. J. & Lindsay, S. W. (2008) Risk factors for house-entry by malaria vectors in a rural town and satellite villages in The Gambia. *Malaria Journal*, 7, 2.
- Kiware, S. S., Chitnis, N., Tatarsky, A., Wu, S., Castellanos, H. M. S., Gosling, R., Smith, D. & Marshall,
 J. M. (2017) Attacking the mosquito on multiple fronts: Insights from the Vector Control Optimization Model (VCOM) for malaria elimination. *PLoS ONE*, 12, e0187680-e0187680.
- Kline, D. L. (2007) Semiochemicals, traps/targets and mass trapping technology for mosquito management. *Journal of the American Mosquito Control Association*, 23, 241-251.

- Klun, J. A., Khrimian, A. & Debboun, M. (2006) Repellent and deterrent effects of SS220, Picaridin, and Deet suppress human blood feeding by *Aedes aegypti*, *Anopheles stephensi*, and *Phlebotomus papatasi*. *Journal of Medical Entomology*, **43**, 34-39.
- Kwan, M. W. C., Weisenseel, J. P., Giel, N., Bosak, A., Batich, C. D. & Willenberg, B. J. (2018) Detection and quantification of trace airborne transfluthrin concentrations via air sampling and thermal desorption gas chromatography-mass spectrometry. *Journal of Chromatography A*,1573, 156-160.
- Lalani, T., Yun, H., Tribble, D., Ganesan, A., Kunz, A., Fairchok, M., Schnaubelt, E., Fraser, J., Mitra, I., Kronmann, K. C., Burgess, T., Deiss, R. G., Riddle, M. S. & Johnson, M. D. (2016) A comparison of compliance rates with anti-vectorial protective measures during travel to regions with dengue or chikungunya activity, and regions endemic for *Plasmodium falciparum* malaria. *Journal of Travel Medicine*,23.
- LeClair, C., Cronery, J., Kessy, E., Tomás, E. V. E., Kulwa, Y., Mosha, F. W., Rowland, M., Protopopoff, N. & Derek Charlwood, J. (2017) 'Repel all biters': an enhanced collection of endophilic *Anopheles gambiae* and *Anopheles arabiensis* in CDC light-traps, from the Kagera Region of Tanzania, in the presence of a combination mosquito net impregnated with piperonyl butoxide and permethrin. *Malaria Journal*, 16, 336-336.
- Lee, J., Choi, D. B., Liu, F., Grieco, J. P. & Achee, N. L. (2018) Effect of the Topical Repellent para-Menthane-3,8-diol on Blood Feeding Behavior and Fecundiy of the Dengue Virus Vector Aedes eagypti. *Insects*,9, 60.
- Lehane, M. J. (2005) The Biology of Blood-Sucking in Insects Cambridge University Press.
- Li, C.-X., Smith, M. L., Fulcher, A., Kaufman, P. E., Zhao, T.-Y. & Xue, R.-D. (2015) Field Evaluation of Three New Mosquito Light Traps Against Two Standard Light Traps to Collect Mosquitoes (Diptera: Culicidae) and Non-Target Insects in Northeast Florida. *Florida Entomologist*, 98, 114-117, 114.
- Li, C., Smith, M. L., Fulcher, A., Kaufman, P. E., Zhao, T. & Xue, R. (2015) Field evaluation of three new mosquito light traps against two standard light traps to collect mosquitoes (Diptera: Culicidae) and non-target insects in northeast Florida. *Florida Entomologist*, June.
- Li, Z. X., Pickett, J. A., Field, L. & Zhou, J. J. (2005) Identification and expression of odorant-binding proteins of the malaria-carrying mosquitoes *Anopheles gambiae* and *Anopheles arabiensis*. *Archives of Insect Biochemistry and Physiology*, **58**, 175-189.
- Li, Z., Zhou, J. J., Shen, Z. & Field, L. (2004) Identification and expression profiling of putative odorantbinding proteins in the malaria mosquitoes, *Anopheles gambiae* and A. arabiensis. *Science China Life Sciences*, 47, 567-576.

- Limwagu, A. J., Kaindoa, E. W., Ngowo, H. S., Hape, E., Finda, M., Mkandawile, G., Kihonda, J., Kifungo, K., Njalambaha, R. M., Matoke-Muhia, D. & Okumu, F. O. (2019) Using a miniaturized double-net trap (DN-Mini) to assess relationships between indoor-outdoor biting preferences and physiological ages of two malaria vectors, *Anopheles arabiensis* and *Anopheles funestus*. *Malaria Journal*, 18, 282.
- Lindh, J. M., Okal, M. N., Herrera-Varela, M., Borg-Karlson, A.-K., Torto, B., Lindsay, S. W. & Fillinger, U. (2015) Discovery of an oviposition attractant for gravid malaria vectors of the *Anopheles gambiae* species complex. *Malaria Journal*, 14, 119.
- Lindsay, S. W., Emerson, P. M. & Charlwood, J. D. Reducing malaria by mosquito-proofing houses. *Trends in Parasitology*, **18**, 510-514.
- Lindsay, S. W., Jawara, M., Paine, K., Pinder, M., Walraven, G. E. & Emerson, P. M. (2003) Changes in house design reduce exposure to malaria mosquitoes. *Tropical Medicine & International Health*, 8, 512-517.
- Liu, M. Z. & Vosshall, L. B. (2019) General visual and contingent thermal cues interact to elicit attraction in female *Aedes aegypti* mosquitoes. *Current Biology*, 29(13), 2250-2257. e2254
- Liu, N. (2015) Insecticide Resistance in Mosquitoes: Impact, Mechanisms, and Research Directions. Annual Review of Entomology,60, 537-559.
- Liu, R., Ma, G., Meng, F. T. & Su, Z. G. (2005) Preparation of uniform-sized PLA microcapsules by combining Shirasu porous glass membrane emulsification technique and multiple emulsion-solvent evaporation method. *J Control Release*, **103**, 31-43.
- Lorenz, L. M., Keane, A., Moore, J. D., Munk, C. J., Seeholzer, L., Mseka, A., Simfukwe, E., Ligamba, J., Turner, E. L., Biswaro, L. R., Okumu, F. O., Killeen, G. F., Mukabana, W. R. & Moore, S. J. (2013) Taxis assays measure directional movement of mosquitoes to olfactory cues. *Parasites and Vectors*, 6, 131.
- Lu, T., Qiu, Y. T., Wang, G., Kwon, J. Y., Rutzler, M., Kwon, H. W., Pitts, R. J., van Loon, J. J. A., Takken, W., Carlson, J. R. & Zwiebel, L. J. (2007) Odor coding in the maxillary palp of the malaria vector mosquito *Anopheles gambiae*. *Current Biology*, **17**, 1533-1544.
- Lynch, P. A., & Boots, M. (2016) Using evolution to generate sustainable malaria control with spatial repellents. *eLIFE*.e15416
- Madhubabu, G. & Yenugu, S. (2017) Exposure to allethrin-based mosquito coil smoke during gestation and postnatal development affects reproductive function in male offspring of rat. *Inhalation Toxicology*, 29, 374-385.
- Mafra-Neto, A. & Dekker, T. (2019) Novel odor-based strategies for integrated management of vectors of disease. *Current Opinion in Insect Science*, 34, 105-111.

- Magbity, E. B., Lines, J. D., Marbiah, M. T., David, K. & Peterson, E. (2002) How reliable are light traps in estimating biting rates of adult *Anopheles gambiae* s.l. (Diptera: Culicidae) in the presence of treated bed nets? *Bulletin of Entomological Research*, 92.
- Maia, M. F., Kliner, M., Richardson, M., Lengeler, C. & Moore, S. J. (2018) Mosquito repellents for malaria prevention. *The Cochrane Database of Systematic Reviews*, 2, CD011595.
- Maia, M. F., Kreppel, K., Mbeyela, E., Roman, D., Mayagaya, V., Lobo, N. F., Ross, A. & Moore, S. J. (2016) A crossover study to evaluate the diversion of malaria vectors in a community with incomplete coverage of spatial repellents in the Kilombero Valley, Tanzania. *Parasites and Vectors*, 9, 451.
- Maia, M. F. & Moore, S. J. (2011) Plant-based insect repellents: a review of their efficacy, development and testing. *Malaria Journal*, **10**, S11.
- Manin, B. O., Ferguson, H. M., Vythilingam, I., Fornace, K., William, T., Torr, S. J., Drakeley, C. & Chua, T. H. (2016) Investigating the Contribution of Peri-domestic Transmission to Risk of Zoonotic Malaria Infection in Humans. *PLoS Neglected Tropical Diseases*, 10, e0005064-e0005064.
- Margus, A., Piiroinen, S., Lehmann, P., Tikka, S., Karvanen, J. & Lindstrom, L. (2019) Sublethal pyrethroid insecticide exposure carries positive fitness effects over generations in a pest insect. *Scientific Reports*, **9**, 11320.
- Martin, N. J., Nam, V. S., Lover, A. A., Phong, T. V., Tu, T. C. & Mendenhall, I. H. (2020) The impact of transfluthrin on the spatial repellency of the primary malaria mosquito vectors in Vietnam: *Anopheles dirus* and *Anopheles minimus*. *Malaria Journal*, 19, 9.
- Martin, N. J., Smith, P. A., Achee, N. L. & DeLong, G. T. (2013) Determining airborne concentrations of spatial repellent chemicals in mosquito behavior assay systems. *PLoS ONE*, 8, e71884.
- Masalu, J. P., Finda, M., Killeen, G. F., Ngowo, H. S., Pinda, P. G. & Okumu, F. O. (2020) Creating mosquito-free outdoor spaces using transfluthrin-treated chairs and ribbons. *Malaria Journal*, **19**, 109.
- Masalu, J. P., Finda, M., Okumu, F. O., Minja, E. G., Mmbando, A. S., Sikulu-Lord, M. T. & Ogoma, S. B. (2017) Efficacy and user acceptability of transfluthrin-treated sisal and hessian decorations for protecting against mosquito bites in outdoor bars. *Parasites and Vectors*, 10, 197.
- Masalu, J. P., Okumu, F. O., Mmbando, A. S., Sikulu-Lord, M. T. & Ogoma, S. B. (2018) Potential benefits of combining transfluthrin-treated sisal products and long-lasting insecticidal nets for controlling indoor-biting malaria vectors. *Parasites and Vectors*, 11, 231.
- Mathania, M. M., Kimera, S. I. & Silayo, R. S. (2016) Knowledge and awareness of malaria and mosquito biting behaviour in selected sites within Morogoro and Dodoma regions Tanzania. *Malaria Journal*, 15, 287-287.

- Mathenge, E. M., Misiani, G. O., Irungu, L. W., Ndegwa, P. N., Smith, L. W., Killeen, G. F. & Knols, B. G. J. (2005) Comparative performance of the Mbita trap, CDC light trap and the human landing catch in the sampling of *Anopheles arabiensis*: an funestus and culicine species in a rice irrigation in western Kenya. *Malaria Journal*, 4.
- Matowo, N. S., Moore, J., Mapua, S., Madumla, E. P., Moshi, I. R., Kaindoa, E. W., Mwangungulu, S. P., Kavishe, D. R., Sumaye, R. D., Lwetoijera, D. W. & Okumu, F. O. (2013) Using a new odourbaited device to explore options for luring and killing outdoor-biting malaria vectors: a report on design and field evaluation of the Mosquito Landing Box. *Parasites and Vectors*, 6, 137.
- Matowo, N. S., Munhenga, G., Tanner, M., Coetzee, M., Feringa, W. F., Ngowo, H. S., Koekemoer, L. L.
 & Okumu, F. O. (2017) Fine-scale spatial and temporal heterogeneities in insecticide resistance profiles of the malaria vector, *Anopheles arabiensis* in rural south-eastern Tanzania. *Wellcome Open Research*, 2, 96-96.
- Mbare, O., Lindsay, S. W. & Fillinger, U. (2013) Dose-response tests and semi-field evaluation of lethal and sub-lethal effects of slow release pyriproxyfen granules (Sumilarv(R)0.5G) for the control of the malaria vectors *Anopheles gambiae* sensu lato. *Malaria Journal*, **12**, 94.
- Mbare, O., Lindsay, S. W. & Fillinger, U. (2019) Testing a pyriproxyfen auto-dissemination station attractive to gravid *Anopheles gambiae* sensu stricto for the development of a novel attract-release and-kill strategy for malaria vector control. *BMC Infectious Diseases*, **19**, 800.
- Mboera, L. E. G., Knols, B. G. J., Braks, M. A. H. & Takken, W. (2000) Comparison of carbon dioxidebaited trapping systems for sampling outdoor mosquito populations in Tanzania. *Medical and Veterinary Entomology*,**14**(3):257-263.
- Mboera, L. E. G., Knols, B. G. J., Takken, W. & Della Torre, A. (1997) The response of *Anopheles gambiae s.l.* and *Anopheles funestus* (Diptera: Culicidae) to tents baited with human odour or carbon dioxide in Tanzania. *Bulletin of Entomological Research*, **87**(2), 173-178.
- Mbogo, C. N. M., Glass, G. E., Forster, D., Kabiru, E. W., Githure, J. I. & Ouma, J. H. (1993) Evaluation of light traps for sampling anopheline mosquitoes in Kilifi, Kenya. *Journal of the American Mosquito Control Association*, **9**.
- Mburu, M. M., Juurlink, M., Spitzen, J., Moraga, P., Hiscox, A., Mzilahowa, T., Takken, W. & McCann,
 R. S. (2018) Impact of partially and fully closed eaves on house entry rates by mosquitoes. *Parasites and Vectors*, 11, 383.
- Mburu, M. M., Mweresa, C. K., Omusula, P., Hiscox, A., Takken, W. & Mukabana, W. R. (2017) 2-Butanone as a carbon dioxide mimic in attractant blends for the Afrotropical malaria mosquitoes *Anopheles gambiae* and *Anopheles funestus. Malaria Journal*, **16**, 351.

- Mburu, M. M., Zembere, K., Hiscox, A., Banda, J., Phiri, K. S., van den Berg, H., Mzilahowa, T., Takken, W. & McCann, R. S. (2019) Assessment of the Suna trap for sampling mosquitoes indoors and outdoors. *Malaria Journal*, **18**, 51.
- McMeniman, C. J., Corfas, R. A., Matthews, B. J., Ritchie, S. A. & Vosshall, L. B. (2014) Multimodal integration of carbon dioxide and other sensory cues drives mosquito attraction to humans. *Cell*, **156**.
- McPhatter, L. P., Mischler, P. D., Webb, M. Z., Chauhan, K., Lindroth, E. J., Richardson, A. G. & Debboun, M. (2017) Laboratory and semi-field evaluations of two (transfluthrin) spatial repellent devices against *Aedes aegypti* (L.) (Diptera: Culicidae). US Army Medical Department Journal, 1-17, 13-22.
- Menger, D. J., Omusula, P., Holdinga, M., Homan, T., Carreira, A. S., Vandendaele, P., Derycke, J. L., Mweresa, C. K., Mukabana, W. R., van Loon, J. J. A. & Takken, W. (2015) Field evaluation of a push-pull system to reduce malaria transmission. *PLoS ONE*, **10**, e0123415.
- Menger, D. J., Omusula, P., Wouters, K., Oketch, C., Carreira, A. S., Durka, M., Derycke, J. L., Loy, D. E., Hahn, B. H., Mukabana, W. R., Mweresa, C. K., van Loon, J. J. A., Takken, W. & Hiscox, A. (2016) Eave screening and push-pull tactics to reduce house entry by vectors of malaria. *The American Journal of Tropical Medicine and Hygiene*, 94, 868-878.
- Menger, D. J., Otieno, B., de Rijk, M., Mukabana, W. R., van Loon, J. J. A. & Takken, W. (2014) A push-pull system to reduce house entry of malaria mosquitoes. *Malaria Journal*, **13**, 119.
- Meyers, J. I., Pathikonda, S., Popkin-Hall, Z. R., Medeiros, M. C., Fuseini, G., Matias, A., Garcia, G., Overgaard, H. J., Kulkarni, V., Reddy, V. P., Schwabe, C., Lines, J., Kleinschmidt, I. & Slotman, M. A. (2016) Increasing outdoor host-seeking in *Anopheles gambiae* over 6 years of vector control on Bioko Island. *Malaria Journal*, 15, 239.
- Miro Specos, M. M., Garcia, J. J., Gutierrez, A. C. & Hermida, L. G. (2016) Application of microencapsulated biopesticides to improve repellent finishing of cotton fabrics. *The Journal of The Textile Institute*, **108**, 1454-1460.
- Misni, N., Nor, Z. M. & Ahmad, R. (2017) Repellent effect of microencapsulated essential oil in lotion formulation against mosquito bites. *Journal of Vector Borne Diseases*, **54**, 44-53.
- Mmbando, A. S., Batista, E. P. A., Kilalangongono, M., Finda, M. F., Mwanga, E. P., Kaindoa, E. W., Kifungo, K., Njalambaha, R. M., Ngowo, H. S., Eiras, A. E. & Okumu, F. O. (2019) Evaluation of a push–pull system consisting of transfluthrin-treated eave ribbons and odour-baited traps for control of indoor- and outdoor-biting malaria vectors. *Malaria Journal*, 18, 87.

- Mmbando, A. S., Ngowo, H. S., Kilalangongono, M., Abbas, S., Matowo, N. S., Moore, S. J., et al. (2017) Small-scale field evaluation of push-pull system against early- and outdoor-biting malaria mosquitoes in an area of high pyrethroid resistance in Tanzania. Wellcome Open Research, 2, 112.
- Mmbando, A. S., Ngowo, H., Limwagu, A., Kilalangongono, M., Kifungo, K. & Okumu, F. O. (2018) Eave ribbons treated with the spatial repellent, transfluthrin, can effectively protect against indoorbiting and outdoor-biting malaria mosquitoes. *Malaria Journal*, 17, 368-368.
- Moiroux, N., Djènontin, A., Bio-Bangana, A. S., Chandre, F., Corbel, V. & Guis, H. (2014) Spatiotemporal analysis of abundances of three malaria vector species in southern Benin using zerotruncated models. *Parasites and Vectors*, 7, 103-103.
- Moore, S. J., Davies, C. R., Hill, N. & Cameron, M. M. (2007) Are mosquitoes diverted from repellentusing individuals to non-users? Results of a field study in Bolivia. *Tropical Medicine and International Health*, 12, 532-539.
- Moore, S. J., Mordue Luntz, A. J. & Logan, J. G. (2012) Insect bite prevention. *Infectious Disease Clinics* of North America, **26**, 655-673.
- Moore, S. J., Zunwei, D., Hongning, Z., Xuezhong, W., Hongbing, L., Yujiang, X. & Hill, N. (2001) The efficacy of different mosquito trapping methods in a forest-fringe village, Yunnan Province, Southern China. Southeast Asian Journal of Tropical Medicine and Public Health, 32, 282-289.
- Morton, F., Travis, B. & Linduska, J. (1947) Results of screening tests with materials evaluated as insecticides, miticides and repellents at the Orlando, Fla., laboratory : April 1942 to April 1947. United States.: Bureau of Entomology and Plant Quarantine, 246.
- Moshi, I. R., Manderson, L., Ngowo, H. S., Mlacha, Y. P., Okumu, F. O. & Mnyone, L. L. (2018) Outdoor malaria transmission risks and social life: a qualitative study in South-Eastern Tanzania. *Malaria Journal*, 17, 397.
- Moshi, I. R., Ngowo, H., Dillip, A., Msellemu, D., Madumla, E. P., Okumu, F. O., Coetzee, M., Mnyone,
 L. L. & Manderson, L. (2017) Community perceptions on outdoor malaria transmission in Kilombero Valley, Southern Tanzania. *Malaria Journal*, 16, 274.
- Mukabana, W. R., Mweresa, C. K., Otieno, B., Omusula, P., Smallegange, R. C., van Loon, J. J. A. & Takken, W. (2012) A novel synthetic odorant blend for trapping of malaria and other African mosquito species. *Journal of Chemical Ecology, Mar*; **38**(3), 235-244.
- Mukabana, W. R., Takken, W., Coe, R. & Knols, B. G. (2002) Host-specific cues cause differential attractiveness of Kenyan men to the African malaria vector *Anopheles gambiae*. *Malaria Journal*, **1**, 17.

- Murchie, A. K., Clawson, S., Rea, I., Forsythe, I. W. N., Gordon, A. W. & Jess, S. (2016) DEET (N,Ndiethyl-meta-toluamide)/PMD (para-menthane-3,8-diol) repellent-treated mesh increases Culicoides catches in light traps. *Parasitology Research*, **115**, 3543-3549.
- Murphy, E. J., Booth, J. C., Davrazou, F., Port, A. M. & Jones, D. N. (2013) Interactions of Anopheles gambiae odorant-binding proteins with a human-derived repellent: implications for the mode of action of n,n-diethyl-3-methylbenzamide (DEET). The Journal of Biological Chemistry,288, 4475-4485.
- Muturi, E. J., Muriu, S., Shililu, J., Mwangangi, J. M., Jacob, B. G., Mbogo, C., Githure, J. & Novak, R. J. (2008) Blood-feeding patterns of Culex quinquefasciatus and other culicines and implications for disease transmission in Mwea rice scheme, Kenya. *Parasitology Research*, **102**, 1329-1335.
- Mwanga, E. P., Mmbando, A. S., Mrosso, P. C., Stica, C., Mapua, S. A., Finda, M. F., Kifungo, K., Kafwenji, A., Monroe, A. C., Ogoma, S. B., Ngowo, H. S. & Okumu, F. O. (2019) Eave ribbons treated with transfluthrin can protect both users and non-users against malaria vectors. *Malaria Journal*, 18, 314.
- Mwanga, E. P., Ngowo, H. S., Mapua, S. A., Mmbando, A. S., Kaindoa, E. W., KIfungo, K. & Okumu, F.
 O. (2019) Evaluation of an ultraviolet LED trap for catching Anopheles and Culex mosquitoes in south-eastern Tanzania. *Parasites and Vectors*, 12, 418.
- Mweresa, C. K., Omusula, P., Otieno, B., van Loon, J. J. A., Takken, W. & Mukabana, W. R. (2014) Molasses as a source of carbon dioxide for attracting the malaria mosquitoes *Anopheles gambiae* and Anopheles funestus. *Malaria Journal*, **13**, 160.
- Mweresa, C. K., Otieno, B., Omusula, P., Weldegergis, B. T., Verhulst, N. O., Dicke, M., van Loon, J. J. A., Takken, W. & Mukabana, W. R. (2015) Understanding the long-lasting attraction of malaria mosquitoes to odor baits. *PLoS ONE*, **10**, e0121533.
- Nararak, J., Giorgio, C. D., Sukkanon, C., Mahiou-Leddet, V., Ollivier, E., Manguin, S. & Chareonviriyaphap, T. (2020) Excito-repellency and biological safety of β-caryophyllene oxide against *Aedes albopictus* and *Anopheles dirus* (Diptera: Culicidae). *Acta Tropica*,**210**, 105556.
- Nazimek, T., Wasak, M., Zgrajka, W. & Turski, W. A. (2011) Content of Transfluthrin in Indoor air during the use of electro-vaporizers. *Annals of Agricultural and Environmental Medicine*, 18, 85-88.
- Nchoutpouen, E., Talipouo, A., Djiappi-Tchamen, B., Djamouko-Djonkam, L., Kopya, E., Ngadjeu, C. S., *et al.* (2019) Culex species diversity, susceptibility to insecticides and role as potential vector of Lymphatic filariasis in the city of Yaounde, Cameroon. *PLoS Neglected Tropical Diseases*, 13, e0007229.

- Nentwig, G., Frohberger, S. & Sonneck, R. (2016) Evaluation of Clove Oil, Icaridin, and Transfluthrin for Spatial Repellent Effects in Three Tests Systems Against the Aedes aegypti (Diptera: Culicidae). *Journal of Medical Entomology*, **54**, 150-158.
- Ngadjeu, C. S., Doumbe-Belisse, P., Talipouo, A., Djamouko-Djonkam, L., Awono-Ambene, P., Kekeunou, S., Toussile, W., Wondji, C. S. & Antonio-Nkondjio, C. (2020) Influence of house characteristics on mosquito distribution and malaria transmission in the city of Yaoundé, Cameroon. *Malaria Journal*, 19, 53-53.
- Ng'ang'a, P. N., Mutunga, J., Oliech, G. & Mutero, C. M. (2019) Community knowledge and perceptions on malaria prevention and house screening in Nyabondo, Western Kenya. *BMC Public Health*, 19, 423.
- Njie, M., Dilger, E., Lindsay, S. W. & Kirby, M. J. (2009) Importance of eaves to house entry by anopheline, but not culicine, mosquitoes. *Journal of Medical Entomology*, **46**, 505-510.
- Njihia, T. N., Jaramillo, J., Murungi, L., Mwenda, D., Orindi, B., Poehling, H. M. & Torto, B. (2014) Spiroacetals in the colonization behaviour of the coffee berry borer: a 'push-pull' system. *PLoS ONE*,9, e111316.
- Njiru, B., Mukabana, W., Takken, W. & Knols, B. (2006) Trapping of the malaria vector *Anopheles* gambiae with odour-baited MM-X traps in semi-field conditions in western Kenya. *Malaria* Journal, **5**, 39.
- Njoroge, M. M., Tirados, I., Lindsay, S. W., Vale, G. A., Torr, S. J. & Fillinger, U. (2017) Exploring the potential of using cattle for malaria vector surveillance and control: a pilot study in western Kenya. *Parasites and Vectors*, **10**, 18.
- Nkya, T. E., Akhouayri, I., Poupardin, R., Batengana, B., Mosha, F., Magesa, S., Kisinza, W. & David, J.
 P. (2014) Insecticide resistance mechanisms associated with different environments in the malaria vector *Anopheles gambiae*: a case study in Tanzania. *Malaria Journal*, 13, 28
- Norris, E. J. & Coats, J. R. (2017) Current and future repellent technologies: the potential of spatial repellents and their place in mosquito-borne disease control. *International Journal of Environmental Research and Public Health*, **14**, 124.
- Nyasembe, V. O., Tchouassi, D. P., Pirk, C. W. W., Sole, C. L. & Torto, B. (2018) Host plant forensics and olfactory-based detection in Afro-tropical mosquito disease vectors. *PLoS Neglected Tropical Diseases*, 12, e0006185.
- Nyasembe, V. O., Teal, P. E. A., Mukabana, W. R., Tumlinson, J. H. & Torto, B. (2012) Behavioural response of the malaria vector *Anopheles gambiae* to host plant volatiles and synthetic blends. *Parasites and Vectors*, **5**, 234.

- Obermayr, U., Ruther, J., Bernier, U. R., Rose, A. & Geier, M. (2015) Evaluation of a push-pull approach for *Aedes aegypti* (L.) using a novel dispensing system for spatial repellents in the laboratory and in a semi-field environment. *PLoS ONE*, **10**, e0129878.
- Ogola, E. O., Fillinger, U., Ondiba, I. M., Villinger, J., Masiga, D. K., Torto, B. & Tchouassi, D. P. (2018) Insights into malaria transmission among *Anopheles funestus* mosquitoes, Kenya. *Parasites and Vectors*, 11, 577.
- Ogoma, S. B., Lorenz, L. M., Ngonyani, H., Sangusangu, R., Kitumbukile, M., Kilalangongono, M., Simfukwe, E. T., Mseka, A., Mbeyela, E., Roman, D., Moore, J., Kreppel, K., Maia, M. F. & Moore, S. J. (2014) An experimental hut study to quantify the effect of DDT and airborne pyrethroids on entomological parameters of malaria transmission. *Malaria Journal*, 13, 131.
- Ogoma, S. B., Lweitoijera, D. W., Ngonyani, H., Furer, B., Russell, T. L., Mukabana, W. R., Killeen, G. F. & Moore, S. J. (2010) Screening mosquito house entry points as a potential method for integrated control of endophagic filariasis, arbovirus and malaria vectors. *PLoS Neglected Tropical Diseases*, 4, e773.
- Ogoma, S. B., Mmando, A. S., Swai, J. K., Horstmann, S., Malone, D. & Killeen, G. F. (2017) A low technology emanator treated with the volatile pyrethroid transfluthrin confers long term protection against outdoor biting vectors of lymphatic filariasis, arboviruses and malaria. *PLoS Neglected Tropical Diseases*, **11**, e0005455.
- Ogoma, S. B., Moore, S. J. & Maia, M. F. (2012) A systematic review of mosquito coils and passive emanators: defining recommendations for spatial repellency testing methodologies. *Parasites and Vectors*, **5**, 287.
- Ogoma, S. B., Ngonyani, H., Simfukwe, E. T., Mseka, A., Moore, J. & Killeen, G. F. (2012) Spatial repellency of transfluthrin-treated hessian strips against laboratory-reared *Anopheles arabiensis* mosquitoes in a semi-field tunnel cage. *Parasites and Vectors*, **5**, 54.
- Ogoma, S. B., Ngonyani, H., Simfukwe, E. T., Mseka, A., Moore, J., Maia, M. F., Moore, S. J. & Lorenz, L. M. (2014) The mode of action of spatial repellents and their impact on vectorial capacity of *Anopheles gambiaesensu stricto. PLoS ONE*,9, e110433.
- Okal, M. N., Francis, B., Herrera-Varela, M., Fillinger, U. & Lindsay, S. W. (2013) Water vapour is a pre-oviposition attractant for the malaria vector *Anopheles gambiae* sensu stricto. *Malaria Journal*, 12, 365.
- Okal, M. N., Herrera-Varela, M., Ouma, P., Torto, B., Lindsay, S. W., Lindh, J. M. & Fillinger, U. (2015) Analysing chemical attraction of gravid *Anopheles gambiae sensu stricto* with modified BG-Sentinel traps. *Parasites and Vectors*, 8, 301.

- Okumu, F. O., Govella, N. J., Moore, S. J., Chitnis, N. & Killeen, G. F. (2010a) Potential benefits, limitations and target product-profiles of odor-baited mosquito traps for malaria control in Africa. *PLoS ONE*,**5**, e11573.
- Okumu, F. O., Killeen, G. F., Ogoma, S., Biswaro, L., Smallegange, R. C., Mbeyela, E., Titus, E., Munk, C., Ngonyani, H., Takken, W., Mshinda, H., Mukabana, W. R. & Moore, S. J. (2010a) Development and field evaluation of a synthetic mosquito lure that is more attractive than humans. *PLoS ONE*, 5, e8951.
- Okumu, F. O., Madumla, E., John, A., Lwetoijera, D. & Sumaye, R. (2010b) Attracting, trapping and killing disease-transmitting mosquitoes using odor-baited stations -The Ifakara odor-baited stations. *Parasites and Vectors*, **3**, 12.
- Okumu, F., Biswaro, L., Mbeleyela, E., Killeen, G. F., Mukabana, R. & Moore, S. J. (2010a) Using Nylon Strips to Dispense Mosquito Attractants for Sampling the Malaria Vector Anopheles gambiae s.s. Journal of Medical Entomology, 47, 274-282.
- Olanga, E. A., Okal, M. N., Mbadi, P. A., Kokwaro, E. D. & Mukabana, W. R. (2010) Attraction of *Anopheles gambiae* to odour baits augmented with heat and moisture. *Malaria Journal*,9, 6.
- Oli, K., Jeffery, J. & Vythilingam, I. (2005) A comparative study of adult mosquito trapping using dry ice and yeast generated carbon dioxide. *Tropical Biomedicine*, **22**(2):249-251.
- Oliveira, E. C. d., Muller, E. I., Abad, F., Dallarosa, J. & Adriano, C. (2010) Internal standard versus external standard calibration: an uncertainty case study of a liquid chromatography analysis. *Química Nova*,**33**, 984-987.
- Onder, E., Sarier, N. & Cimen, E. (2008) Encapsulation of phase change materials by complex coacervation to improve thermal performances of woven fabrics. *Thermochimica Acta*,**467**, 63-72.
- Paluch, G., Bartholomay, L. & Coats, J. (2010) Mosquito repellents: a review of chemical structure diversity and olfaction. *Pest Management Science*, 66, 925-935.
- Pates, H. V., Lines, J. D., Keto, A. J. & Miller, J. E. (2002) Personal protection against mosquitoes in Dar es Salaam, Tanzania, by using a kerosene oil lamp to vaporize transfluthrin. *Medical and Veterinary Entomology*, 16, 277-284.
- Patrascu, E., Rapeanu, G., Bonciu, C. & Hopulele, T. (2009) Bioethanol production from molasses by different strains of *Saccharomyces cerevisiae*, **3**(33), 49-56. In AUDJG
- Pettebone, M. S. (2014) Characterization of Transfluthrin emissions over time in an enclosed space over a range of discrete temperatures. *Faculty of the Occupational & Environmental Health Sciences Graduate Program.*
- Platt, R. B., Collins, C. L. & Witherspoon, J. P. (1957) Reactions of Anopheles quadrimaculatus Say to moisture, temperature, and light. *Ecological Monographs*, 27, 303-324.

- Ponlawat, A., Khongtak, P., Jaichapor, B., Pongsiri, A. & Evans, B. P. (2017) Field evaluation of two commercial mosquito traps baited with different attractants and colored lights for malaria vector surveillance in Thailand. *Parasites and Vectors*, **10**, 378.
- Pyke, B., Rice, M., Sabine, B. & Zalucki, M. (1987) The push-pull strategy-behavioural control of Heliothis. *Australian Cotton Grower*, 1987.
- Qiu, Y. T., Smallegange, R. C., Braak, C. J. F., Spitzen, J., Van Loon, J. J. A., Jawara, M., Milligan, P., Galimard, A. M., Van Beek, T. A., Knols, B. G. J. & Takken, W. (2007) Attractiveness of MM-X traps baited with human or synthetic odor to mosquitoes (Diptera: Culicidae) in The Gambia. *Journal of Medical Entomology*, 44.
- Qiu, Y. T. & van Loon, J. J. A. (2010) Olfactory physiology of blood-feeding vector mosquitoes. In Olfaction in vector-host interactions (ed. by W. Takken & B. G. J. Knols). Wageningen Academic Publishers, Wageningen.
- R Core Team. (2018) R: A language and environment for statistical computing. *R Foundation for Statistical Computing, Vienna, Austria.*
- Raji, J. I. & DeGennaro, M. (2017) Genetic Analysis of Mosquito Detection of Humans. *Current Opinion* in Insect Science, 20, 34-38.
- Rakotondranaivo, T., Randriamanarivo, S. F., Tanjona, M. R., Vigan-Womas, I., Randrianarivelojosia, M.
 & Ndiath, M. O. (2018) Evidence of insecticide resistance to pyrethroids and bendiocarb in *Anopheles funestus* from Tsararano, Marovoay District, Madagascar. *BioMed Research International*, 2018, 5806179.
- Ramesh, A. & Vijayalakshmi, A. (2001) Monitoring of allethrin, deltamethrin, esbiothrin, prallethrin and transfluthrin in air during the use of household mosquito repellents. *Journal of Environmental Monitoring*, **3**, 191-193.
- Ranson, H. & Lissenden, N. (2016) Insecticide Resistance in African Anopheles Mosquitoes: A Worsening Situation that Needs Urgent Action to Maintain Malaria Control. Trends in Parasitology, 32, 187-196.
- Ray, A. (2015) Reception of odors and repellents in mosquitoes. *Current Opinion in Neurobiology*, **34**, 158-164.
- Revay, E. E., Kline, D. L., Xue, R. D., Qualls, W. A., Bernier, U. R., Kravchenko, V. D., Ghattas, N., Pstygo, I. & Muller, G. C. (2013) Reduction of mosquito biting-pressure: spatial repellents or mosquito traps? A field comparison of seven commercially available products in Israel. *Acta Trop*, 127, 63-68.

- Riveron, J. M., Huijben, S., Tchapga, W., Tchouakui, M., Wondji, M. J., Tchoupo, M., Irving, H., Cuamba, N., Maquina, M., Paaijmans, K. & Wondji, C. S. (2019) Escalation of Pyrethroid Resistance in the Malaria Vector *Anopheles funestus* Induces a Loss of Efficacy of Piperonyl Butoxide-Based Insecticide-Treated Nets in Mozambique. *The Journal of infectious diseases*,220, 467-475.
- Rowland, M. (1989) Changes in the circadian flight activity of the mosquito *Anopheles stephensi* associated with insemination, blood-feeding, oviposition and nocturnal light intensity. *Physiological Entomology*,**14**(1):77-84
- Rubio-Palis, Y. (1996) Evaluation of light traps combined with carbon dioxide and 1-octen-3-ol to collect anophelines in Venezuela. *Journal of the American Mosquito Control Association*, **12**, 91-96.
- Russell, T. L., Govella, N. J., Azizi, S., Drakeley, C. J., Kachur, S. F. & Killeen, G. F. (2011) Increased proportions of outdoor feeding among residual malaria vector populations following increased use of insecticide-treated nets in rural Tanzania. *Malaria Journal*, **10**, 80
- Russell, T. L., Lwetoijera, D. W., Maliti, D., Chipwaza, B., Kihonda, J., Charlwood, J. D., Smith, T. A., Lengeler, C., Mwanyangala, M. A., Nathan, R., Knols, B. G. J., Takken, W. & Killeen, G. F. (2010) Impact of promoting longer-lasting insecticide treatment of bed nets upon malaria transmission in a rural Tanzanian setting with pre-existing high coverage of untreated nets. *Malaria Journal*, 9, 187.
- Saavedra, M. P., Conn, J. E., Alava, F., Carrasco-Escobar, G., Prussing, C., Bickersmith, S. A., Sangama, J. L., Fernandez-Minope, C., Guzman, M., Tong, C., Valderrama, C., Vinetz, J. M., Gamboa, D. & Moreno, M. (2019) Higher risk of malaria transmission outdoors than indoors by *Nyssorhynchus darlingi* in riverine communities in the Peruvian Amazon. *Parasites and Vectors*, 12, 374.
- Saini, R. K., Orindi, B. O., Mbahin, N., Andoke, J. A., Muasa, P. N., Mbuvi, D. M., Muya, C. M., Pickett, J. A. & Borgemeister, C. W. (2017) Protecting cows in small holder farms in East Africa from tsetse flies by mimicking the odor profile of a non-host bovid. *PLoS Neglected Tropical Diseases*, 11, e0005977.
- Saitoh, Y., Hattori, J., Chinone, S., Nihei, N., Tsuda, Y. & Kurahashi, H. (2004) Yeast generated CO₂ as a convenient source of carbon dioxide for adult mosquito sampling. *Journal of American Mosquito Control Association*, **20**(3), 261-264.
- Sampaio, V. S., Rivas, G., Kobylinski, K., Pinilla, Y. T., Pimenta, P. F. P., Lima, J. B. P., Bruno, R. V., Lacerda, M. V. G. & Monteiro, W. M. (2017) What does not kill it makes it weaker: effects of sublethal concentrations of ivermectin on the locomotor activity of *Anopheles aquasalis*. *Parasites and Vectors*, 10, 623.

- Sande, S., Zimba, M., Chinwada, P., Masendu, H. T., Mazando, S. & Makuwaza, A. (2015) The emergence of insecticide resistance in the major malaria vector Anopheles funestus (Diptera: Culicidae) from sentinel sites in Mutare and Mutasa Districts, Zimbabwe. *Malaria Journal*, 14, 466.
- Sangoro, O., Kelly, A. H., Mtali, S. & Moore, S. J. (2014a) Feasibility of repellent use in a context of increasing outdoor transmission: a qualitative study in rural Tanzania. *Malaria Journal*, 13, 347.
- Sangoro, O., Turner, E., Simfukwe, E., Miller, J. E. & Moore, S. J. (2014b) A cluster-randomized controlled trial to assess the effectiveness of using 15% DEET topical repellent with long-lasting insecticidal nets (LLINs) compared to a placebo lotion on malaria transmission. *Malaria Journal*, 13, 324.
- Sathantriphop, S., White, S. A., Achee, N. L., Sanguanpong, U. & Chareonviriyaphap, T. (2014) Behavioral responses of *Aedes aegypti*, *Aedes albopictus*, *Culex quinquefasciatus*, and *Anopheles minimus* against various synthetic and natural repellent compounds. *Journal of Vector Ecology*, 39, 328-339.
- Schreck, C. E., Gilbert, I. H., Weidhaas, D. E. & Posey, K. H. (1970) Spatial Action of Mosquito Repellents. *Journal of Economic Entomology*, 63, 1576-1578.
- Semmelhack, J. L. & Wang, J. W. (2009) Select Drosophila glomeruli mediate innate olfactory attraction and aversion. *Nature*, 459, 218-223.
- Sexton, J. D., Hobbs, J. H., St Jean, Y. & Jacques, J. R. (1986) Comparison of an experimental updraft ultraviolet light trap with the CDC miniature light trap and biting collections in sampling for *Anopheles albimanus* in Haiti. *Journal of the American Mosquito Control Association*, 2, 168-173.
- Shaalan, E. A., Canyon, D. V., Younes, M. W., Abdel-Wahab, H. & Mansour, A. H. (2005) Effects of sub-lethal concentrations of synthetic insecticides and Callitris glaucophylla extracts on the development of Aedes aegypti. *Journal of Vector Ecology*, 30, 295-298.
- Sheppard, A. D., Rund, S. S. C., George, G. F., Clark, E., Acri, D. J. & Duffield, G. E. (2017) Light manipulation of mosquito behaviour: acute and sustained photic suppression of biting activity in the *Anopheles gambiae* malaria mosquito. *Parasites and Vectors*, **10**, 255.
- Sherrard-Smith, E., Skarp, J. E., Beale, A. D., Fornadel, C., Norris, L. C., Moore, S. J., Mihreteab, S., Charlwood, J. D., Bhatt, S., Winskill, P., Griffin, J. T. & Churcher, T. S. (2019) Mosquito feeding behavior and how it influences residual malaria transmission across Africa. *Proceedings of the National Academy of Sciences of the United States of America*, **116**, 15086-15095.
- Short, I., Sahgal, A. & Hayduk, W. (1983) Solubility of Ammonia and Hydrogen Sulfide in Several Polar Solvents. *Journal of Chemical Engeneering Data*, 28.63-66
- Silberglied, R. E. (1979) Communication in the ultraviolet. *Annual Review of Ecology, Evolution, and Systematics*, **10**, 373-398.

- Smallegange, R. C., Schmied, W. H., van Roey, K. J., Verhulst, N. O., Spitzen, J., Mukabana, W. R. & Takken, W. (2010) Sugar-fermenting yeast as an organic source of carbon dioxide to attract the malaria mosquito *Anopheles gambiae*. *Malaria Journal*, 9, 292-292.
- Smith, D. L., McKenzie, F. E., Snow, R. W. & Hay, S. I. (2007) Revisiting the basic reproductive number for malaria and its implications for malaria control. *PLoS Biology*, **5**, e42.
- Snow, W. F. (1987) Studies of house-entering habits of mosquitoes in The Gambia, West Africa: experiments with prefabricated huts with varied wall apertures. *Medical and Veterinary Entomology*, **1**, 9-21.
- Sougoufara, S., Diédhiou, S. M., Doucouré, S., Diagne, N., Sembène, P. M., Harry, M., Trape, J.-F., Sokhna, C. & Ndiath, M. O. (2014) Biting by *Anopheles funestus* in broad daylight after use of long-lasting insecticidal nets: a new challenge to malaria elimination. *Malaria Journal*, 13, 125.
- Sougoufara, S., Ottih, E. C. & Tripet, F. (2020) The need for new vector control approaches targeting outdoor biting *Anopheline* malaria vector communities. *Parasites and Vectors*, **13**, 295-295.
- Spitzen, J., Koelewijn, T., Mukabana, W. R. & Takken, W. (2016) Visualization of house-entry behaviour of malaria mosquitoes. *Malaria Journal*, **15**, 233.
- Spitzen, J., Smallegange, R. C. & Takken, W. (2008) Effect of human odours and positioning of CO2 release point on trap catches of the malaria mosquito *Anopheles gambiae* sensu stricto in an olfactometer. *Physiological Entomology*,**33(2)**.116-122
- Spitzen, J., Spoor, C. W., Grieco, F., ter Braak, C., Beeuwkes, J., van Brugge, S. P., Kranenbarg, S., Noldus, L. P. J. J., van Leeuwen, J. L. & Takken, W. (2013) A 3D Analysis of Flight Behavior of *Anopheles gambiae* sensu stricto Malaria Mosquitoes in Response to Human Odor and Heat. *PLoS ONE*,8(5).e62995
- Sriwichai, P., Karl, S., Samung, Y., Sumruayphol, S., Kiattibutr, K., Payakkapol, A., Mueller, I., Yan, G., Cui, L. & Sattabongkot, J. (2015) Evaluation of CDC light traps for mosquito surveillance in a malaria endemic area on the Thai-Myanmar border. *Parasites and Vectors*, 8, 636.
- Stark, W. S., &Tan, K..E.W.P. (1982) Ultraviolet light: photosensitivity and other effects on the visual system. *Photochemistry and Photobiology*, 36, 371-380.
- Suh, E., Bohbot, J. D. & Zwiebel, L. J. (2014) Peripheral olfactory signaling in insects. Current Opinion in Insect Science, 6, 86-92.
- Sukumaran, D., Ponmariappan, S., Sharma, A. K., Jha, H. K. & Wasu, Y. H. (2016) Application of biogenic carbon dioxide produced by yeast with different carbon sources for attraction of mosquitoes towards adult mosquito traps. *Parasitology Research*, **115**, 1453-1462.
- Takken, W. (2010) Push-pull strategies for vector control. Malaria Journal, 9, 116.

- Takken, W. (2011) The Role of Olfaction in Host-Seeking of Mosquitoes: A Review. *International Journal of Tropical Insect Science*, **12**, 287-295.
- Takken, W., Dekker, T. & Wijnholds, Y. G. (1997) Odor-mediated flight behavior of *Anopheles gambiae* gilessensu stricto and *An. stephensi* liston in response to CO₂, acetone, and 1-octen-3-ol (Diptera: Culicidae). *Journal of Insect Behavior*, **10**, 395-407.
- Takken, W. & Kline, D. (1989) Carbon dioxide and 1-octen-3-ol as mosquito attractants. *Journal of American Mosquito Control Association*,**5**(3), 311-316.
- Takken, W. & Knols, B. G. (1999) Odor-mediated behavior of Afrotropical malaria mosquitoes. Annual Review of Entomology, 44, 131-157.
- Takken, W., Knols, B. G. J. & Otten, H. (1997) Interactions between physical and olfactory cues in the host-seeking behaviour of mosquitoes: the role of relative humidity. *Annals of Tropical Medicine & Parasitology*,91, S119-S120.
- Tangena, J. A., Thammavong, P., Chonephetsarath, S., Logan, J. G., Brey, P. T. & Lindsay, S. W. (2018) Field evaluation of personal protection methods against outdoor-biting mosquitoes in Lao PDR. *Parasites and Vectors*, 11, 661.
- Tanner, M. & de Savigny, D. (2008) Malaria eradication back on the table. *Bulletin of the World Health* Organization, **86**, 82.
- Tauxe, G. M., MacWilliam, D., Boyle, S. M., Guda, T. & Ray, A. (2013) Targeting a dual detector of skin and CO₂ to modify mosquito host seeking. *Cell*, 155, 1365-1379.
- Telang, A. & Skinner, J. (2019) Effects of host blood meal source on reproductive output, nutrient reserves and gut microbiome of West Nile virus vector Culex quinquefasciatus. *Journal of Insect Physiology*, 114, 15-22.
- Thievent, K., Hauser, G., Elaian, O. & Koella, J. C. (2019) The interaction between permethrin exposure and malaria infection affects the host-seeking behaviour of mosquitoes. *Malaria Journal*, **18**, 79.
- Tingle, C. C., Rother, J. A., Dewhurst, C. F., Lauer, S. & King, W. J. (2003) Fipronil: environmental fate, ecotoxicology, and human health concerns. *Reviews of Environmental Contamination and Toxicology*, **176**, 1-66.
- Tisch, M., Faulde, M. K. & Maier, H. (2005) Genotoxic effects of pentachlorophenol, lindane, transfluthrin, cyfluthrin, and natural pyrethrum on human mucosal cells of the inferior and middle nasal conchae. *American Journal of Rhinology & Allergy: SAGE Journals*, **19**, 141-151.
- Townson, H., Nathan, M. B., Zaim, M., Guillet, P., Manga, L., Bos, R. & Kindhauser, M. (2005) Exploiting the potential of vector control for disease prevention. *Bulletin of the World Health* Organization, 83, 942-947.

- Tsitoura, P., Koussis, K. & Iatrou, K. (2015) Inhibition of *Anopheles gambiae* odorant receptor function by mosquito repellents. *The Journal of Biological Chemistry*, **290**, 7961-7972.
- Turner, S. L., Li, N., Guda, T., Githure, J., Cardé, R. T. & Ray, A. (2011) Ultra-prolonged activation of CO₂-sensing neurons disorients mosquitoes. *Nature*, 474, 87-91.
- Tusting, L. S., Bottomley, C., Gibson, H., Kleinschmidt, I., Tatem, A. J., Lindsay, S. W. & Gething, P.
 W. (2017) Housing Improvements and Malaria Risk in Sub-Saharan Africa: A Multi-Country Analysis of Survey Data. *PLoS Medicine*, 14, e1002234-e1002234.
- Tzotzos, G., Iley, J. N. & Moore, E. A. (2018) New insights on repellent recognition by Anopheles gambiae odorant-binding protein 1. PLoS ONE, 13, e0194724.
- Uzzan, B., Konate, L., Diop, A., Nicolas, P., Dia, I., Dieng, Y. & Izri, A. (2009) Efficacy of four insect repellents against mosquito bites: a double-blind randomized placebo-controlled field study in Senegal. *Fundamental Clinical Pharmacology*,**23**, 589-594.
- van Breugel, F., Riffell, J., Fairhall, A. & Dickinson, M. H. (2015) Mosquitoes Use Vision to Associate Odor Plumes with Thermal Targets. *Current Biology* : *CB*, **25**, 2123-2129.
- van de Straat, B., Hiscox, A., Takken, W. & Burkot, T. R. (2019) Evaluating synthetic odours and trap designs for monitoring Anopheles farauti in Queensland, Australia. *Malaria Journal*, **18**, 299.
- van den Broek, I. V. & den Otter, C. J. (1999) Olfactory sensitivities of mosquitoes with different host preferences (*Anopheles gambiaes.s.*, *An. arabiensis*, *An. quadriannulatus*, *An. m. atroparvus*) to synthetic host odours. *Journal of Insect Physiology*, **45**, 1001-1010.
- Vaníčková, L., Canale, A. & Benelli, G. (2017) Sexual chemoecology of mosquitoes (Diptera, Culicidae): Current knowledge and implications for vector control programs. *Parasitology International*, 66, 190-195.
- Verhulst, N. O., Bakker, J. W. & Hiscox, A. (2015) Modification of the Suna Trap for Improved Survival and Quality of Mosquitoes in Support of Epidemiological Studies. J Am Mosq Control Assoc, 31, 223-232.
- Verhulst, N. O., Loonen, J. A. & Takken, W. (2013) Advances in methods for colour marking of mosquitoes. *Parasites and Vectors*, 6, 200.
- Verhulst, N. O., Mbadi, P. A., Kiss, G. B., Mukabana, W. R., van Loon, J. J. A., Takken, W. & Smallegange, R. C. (2011) Improvement of a synthetic lure for *Anopheles gambiae* using compounds produced by human skin microbiota. *Malaria Journal*, 10, 28.
- Wachira B.M., M. P. O., Okoth S., Ng'ang'a M.M., William J.M., Murilla G.A., Hassanali A. (2016) Responses of *Glossina pallidipes* and *Glossina morsitans morsitans* tsetse flies to analogues of δoctalactone and selected blends. *Acta Tropica*, **160**, 53-57.

- Wagman, J. M., Achee, N. L. & Grieco, J. P. (2015a) Insensitivity to the spatial repellent action of transfluthrin in Aedes aegypti: a heritable trait associated with decreased insecticide susceptibility. *PLoS Neglected Tropical Diseases*, 9, e0003726-e0003726.
- Wagman, J. M., Grieco, J. P., Bautista, K., Polanco, J., Briceno, I., King, R. & Achee, N. L. (2015) The field evaluation of a push-pull system to control malaria vectors in northern Belize, Central America. *Malaria Journal*, 14, 184.
- Wagman, J. M., Grieco, J. P., Bautista, K., Polanco, J., Briceno, I., King, R. & Achee, N. L. (2014) A comparison of two commercial mosquito traps for the capture of malaria vectors in Northern Belize, central America. *Journal of American Mosquito Control Association*, **30**, 175-183.
- Wamae, P. M., Githeko, A. K., Otieno, G. O., Kabiru, E. W. & Duombia, S. O. (2015) Early biting of the *Anopheles gambiae* s.s. and its challenges to vector control using insecticide treated nets in western Kenya highlands. *Acta Tropica*, **150**, 136-142.
- Wangdi, K., Furuya-Kanamori, L., Clark, J., Barendregt, J. J., Gatton, M. L., Banwell, C., Kelly, G. C., Doi, S. A. R. & Clements, A. C. A. (2018) Comparative effectiveness of malaria prevention measures: a systematic review and network meta-analysis. *Parasites and Vectors*, **11**, 210.
- Wanzirah, H., Tusting, L. S., Arinaitwe, E., Katureebe, A., Maxwell, K., Rek, J., Bottomley, C., Staedke, S. G., Kamya, M., Dorsey, G. & Lindsay, S. W. (2015) Mind the gap: house structure and the risk of malaria in Uganda. *PLoS ONE*, **10**, e0117396.
- Webster, B., Lacey, E. S. & Cardé, R. T. (2015) Waiting with Bated Breath: Opportunistic Orientation to Human Odor in the Malaria Mosquito, *Anopheles gambiae*, is Modulated by Minute Changes in Carbon Dioxide Concentration. *Journal of Chemical Ecology*, **41**, 59-66.
- Weinholt, A. (2016) The EU Biocidal Products Regulation (No. 528/2012). AeroSpace and Defence Industries Association of Europe.
- WHO. (2014) Malaria policy advisory committee to the WHO: conclusions and recommendations of fifth biannual meeting (March 2014). *Malaria Journal*, 13, 253.
- WHO. (1975) Manual on practical entomology in malaria: part II, methods and techniques. WHO Division of Malaria and other Parasitic Diseases, Geneva, Switzerland.
- WHO. (2013) Test procedures for insecticide resistance monitoring in malaria vector mosquitoes. Geneva:World Health Organization.
- WHO. (2019) World Malaria Report 2019. World Health Organization, Geneva.
- Wilson, A. L., Chen-Hussey, V., Logan, J. G. & Lindsay, S. W. (2014) Are topical insect repellents effective against malaria in endemic populations? A systematic review and meta-analysis. *Malaria Journal*, 13, 446-446.

- Wong, J., Bayoh, N., Olang, G., Killeen, G. F., Hamel, M. J., Vulule, J. M. & Gimnig, J. E. (2013) Standardizing operational vector sampling techniques for measuring malaria transmission intensity: evaluation of six mosquito collection methods in western Kenya. *Malaria Journal*, 12, 143.
- Yan, H., Zeng, J. & Zhong, G. (2015) The push-pull strategy for citrus psyllid control. *Pest Management Science*, 71, 893-896.
- Zhou, G., Lee, M.-C., Githeko, A. K., Atieli, H. E. & Yan, G. (2016) Insecticide-Treated Net Campaign and Malaria Transmission in Western Kenya: 2003-2015. *Frontiers in Public Health*, **4**, 153-153.

List of Abbreviations

CDC UV	Centre of Disease Control light trap with ultra violet light
DDT	Dichlorodiphenyltrichloroethane
DEET	N,N-Diethyl-meta-toluamide
GC-FID	Gas Chromatography coupled with Flame Ionisation detector
GEE	Generalized estimating equations
GLMER	Generalized linear mixed effects models
HLC	Human landing catches
IRS	Indoor residual spraying
LED	Light-emitting diode
LLINs	Long-lasting insecticide bed nets
MB5	Mbita blend 5
OR	Odds ratio
PMD	Para menthane-3,8-diol
WHO	World Health Organization

List of publications

- Njoroge M.M., Fillinger U., Saddler A., Moore S., Takken W., van Loon J.J.A., Hiscox A. Evaluating putative repellent 'push' and attractive 'pull' components for manipulating the odour orientation of host-seeking malaria vectors in the peri-domestic space. Parasites &Vectors. 2021 Jan 11; 14(1):42. doi: 10.1186/s13071-020-04556-7. PMID: 33430963; PMCID: PMC7802213.
- Njoroge MM, Tirados I, Lindsay SW, Vale GA, Torr SJ, Fillinger U. Exploring the potential of using cattle for malaria vector surveillance and control: a pilot study in western Kenya. Parasites & Vectors. 2017 Jan 10;10(1):18. doi: 10.1186/s13071-016-1957-8. PMID: 28069065; PMCID: PMC5223359.
- Denz A, Njoroge MM, Tambwe MM, Champagne C, Okumu F, van Loon JJA, Hiscox A, Saddler A, Fillinger U, Moore SJ, Chitnis N. Predicting the impact of outdoor vector control interventions on malaria transmission intensity from semi-field studies. Parasite & Vectors. 2021 Jan 20;14(1):64. doi: 10.1186/s13071-020-04560-x. PMID: 33472661; PMCID: PMC7819244.
- Njoroge MM, Irungu Mwangi V, Owalla TJ, Ozwara H, Egwang TG. Mice pups breastfed by foster mothers whose breast milk contains *Plasmodium falciparum* recombinant SE36 antigen develop specific antibodies. American Journal of Tropical Medicine & Hygiene. 2020 Dec 29. doi: 10.4269/ajtmh.20-0206. Epub ahead of print. PMID: 33377448.

Acknowledgements

To be at a point where the possibility of acquiring a doctorate degree is finally within grasp is surreal to me. When I first became fascinated by how things work, I never thought I would eventually pursue science as a career, although looking back; this has always been in my heart. To get to this point, I have an army of people who have held my hand, opened doors for me, believed in me, corrected and guided me, refused to give up on me and are responsible for getting me here.

I would like to first thank my Promoter, Prof. Willem Takken, for giving me an opportunity to pursue a doctorate degree. You took a lot of time and effort in ensuring everything entailing my PhD is done well. You tirelessly provided input on my work structure and write-up and you gave me many opportunities to step up and become better. I am privileged to be associated with you as you remain a world-class scientist with whom I am honoured to have as a Promoter. To my second promoter, Prof. Joop van Loon, special thanks for always being at hand to guide me and give me a lot of input even though your hands were already quite engaged. Thank you for your availability to discuss through whatever needed clarification and for the extensive input that you made in all my write ups. Special thanks to Dr. Alex Hiscox, my copromoter, who not only gave me the opportunity to work in her project, but also took my journey personally and guided me along the way even when she was not physically next to me. I am grateful to Prof. Marcel Dicke for having secured special funds to allow me to complete the writing process of my thesis. I reserve a special gratitude to Wageningen University &Research (WUR) for offering me a platform to get my doctorate degree and to International Vector Control Consortium (IVCC) for funding our work.

I would like to thank the management at International Centre of Insect Physiology and Ecology (*icipe*) through whom I have had a place to conduct all my research in. Very special thanks to Dr. Ulrike Fillinger, who can only be described as a super human. Uli, you have guided me and mentored me for more than six years and have borne most of the frustrations in getting me to this point. Thank you for never giving up on me even when I gave you many reasons to and for always believing in my abilities even when I could not believe in them myself. Special acknowledgement to Dr. Dan Masiga for making many efforts in ensuring I got this opportunity. Thank you because you invested in me, which is an honour. I also thank Prof. Baldwyn Torto for constantly encouraging me towards this point. Thank you to Capacity Building and Institutional Development (CBID) at *icipe*, through Dr. Rob Skilton, who have offered me many training opportunities and even gave me a platform to participate in training others. Truly appreciated.

I never worked alone. In truth, I have always had a multitude of people around me who have worked hard so that my name appears in this thesis.

At *icipe*-Mbita, there is a myriad of people that have been integral to my work. Thank you Elisha and Davie for ensuring my mosquito supplies were there even during the holidays, I will remain indebted to you. Thanks Bruno for being the hands behind the work, truly grateful. Fedinand, thank you for helping me to the very end and Charles for making sure our Ahero work was immaculately done. My pace-setters; Oscar, Mike and Manuelita for always being at hand to encourage me to the finish line, Paul Ouma, Rose, Liz and Kiche for being old guns from who I received a lot of technical expertise and assistance. To my work family Akbar, Getachew, Richie, Silas, Lillian, Godfrey, Mogaka, Eugene, Salim, Philip, Shem,

Boniface, Collins Mweresa, Annette, Mbaisi, Simiyu, Tracy, Lavender, Joshua, Mwangi, Martha, Motari, Jimmy Pitcharr, Dr. Francis and Mr. Nyongesa, thank you for being part of my journey. My friends; Rofas who has called me Dr. Mendi since he knew me, Abneel, Carol Manoti, Faith Kandie, Faith Utuku, Jenny, Martha, Mose, Naomi, Nzisa, Pauline, Paul, Billy and Ssuki thank you for the many tears you helped wipe from my eyes.

At WUR, I had a lot of people who, even though I spent the least time with them, they offered me an unbelievable amount of support and encouragement. Special thanks to Marieke, who became an integral support system and a lot of thanks to the Ento family; Steve, Tessa, Marilyn, Helen, Hans, Julian, Jeanine, Sandeep, Sander, Jeroen, Daan, Alessia, Alexander, Antonio, Bram, Davy, Els, Filippo, Gabriela, Jeroen, Shaphan, Jessica, Julia, Karol, Katherine, Kay, Luuk, Maite, Max, Monica, Mitchel, Peter, Pieter, Quint, Simone, Sivaprakaham, Stijn, Thibault and Yidong. You guys are amazing. When not in office, I was part of the WUR Kenyan community who I thank for feeding me and getting me a little tipsy sometimes. Thank you Joshua, Jared, Faith, Emma, Cate, Dan, Esther, Alex, Jamleck, Njoro, Ojos, Richard (chairman), Simon, Jane, Emily, Martin, Mercy, Bilhah, Monica, Afranina, John and the many more who reminded me the beauty of family no matter where you are in the world.

At my core is my family who I can not sufficiently thank. They have celebrated this journey even more than I have and have loved me more than I can describe. My mum, Josephine, for praying for me every single day, my sisters; Michelle and Ciru for anchoring me and loving me completely, my big brother, Peter for believing in me always, my nephews; Caleb and Njoro and my niece, Venetia, for being so adorable that their pictures and company fill my heart to the brim. And to my late dad, we talked about this PhD 11 years ago and it finally happened. I wish you could have been here for this. I dedicate this PhD to you. To my extended family who are waiting to slaughter a few goats in my honour; aunt Peris and uncle Ngaruiya and all my cousins from my dad's side of the family as well as all my aunties and uncles and cousins from my mum's side of the family, thank you for being at hand to contribute to and celebrate my achievements.

I truly cannot mention every single person who has been my support but I appreciate you all who have walked this journey with me.

Thank you.

Margaret Mendi Njoroge

Curriculum Vitae



Margaret Mendi Njoroge was born in Nairobi, Kenva on 9th June 1979. After undertaking both her primary and secondary education, she joined Maseno University for a Bachelors of Science in Biomedical sciences and technology and majored in Medical biotechnology. Immediately after this, Margaret received training at the Kemri-Wellcome Trust Research group on Plasmodium falciparum point mutations that lead to drug resistance. This was followed by an intensive 2-year work contract at the Med Biotech Laboratories in Kampala, Uganda where drug resistance mutations in eastern Uganda were studied. She undertook her masters' studies in Molecular Biology at the VrijeUniversiteit Brussel (VUB) in Belgium on a VLIR-UOS scholarship. Her masters' thesis was on Vaccination of promiscuous peptides and dynamics of myeloid cells in experimental models of murine schistomiasis. After her masters' studies, she worked as a project administrator at the African Academy of Sciences. Together with her former Ugandan supervisor, she secured a Grand Challenges Exploration grant on maternal immunization for the protection of neonates and infants from malaria at the Institute of Primate Research (IPR). Since 2012, she has been working at the International Centre of Insect Physiology and Ecology (*icipe*) as a research assistant on various human health projects including treatment of cattle with insecticide for control of mosquito vectors and most recently the East African collaboration on mosquito push-pull, which looked at combined use of spatial repellents and attractants to control outdoor-biting malaria vectors. This was also the project that sponsored her admission into the PhD graduate programme at the Wageningen University and Research (WUR). Her PhD thesis has been on understanding the host-seeking behaviour of Anopheles mosquitoes in response to olfactory and visual cues. Margaret is currently involved in a One-Health project on the use of biorational solutions for controlling human and animal disease vectors in Kwale and Busia counties in Kenya.

PE&RC Training and Education Statement

With the training and education activities listed below the PhD candidate has complied with the requirements set by the C.T. de Wit Graduate School for Production Ecology and Resource Conservation (PE&RC) which comprises of a minimum total of 32 ECTS (= 22 weeks of activities)



Review of literature (4.5 ECTS)

- Background literature on chapters of PhD thesis: host-seeking behaviour of *Anopheles* mosquitoes in response to olfactory and visual cues

Post-graduate courses (7 ECTS)

- R Statistics for arthropod research; ICIPE (2017)
- Vector ecology and disease training; ICIPE (2018)
- International Chemical Ecology (ICE) course; ICIPE/Max Planck/Penn State/SLU (2019)

Deficiency, refresh, brush-up courses (12 ECTS)

- Ecological aspects of biointeractions; WUR (2017)
- Ecological methods; WUR (2017)

Laboratory training and working visits (1.9 ECTS)

- Design of semi-field systems for testing push-pull mosquito control experiments; Ifakara Health Institute, Bagamoyo, Tanzania (2017)

Competence strengthening / skills courses (3.7 ECTS)

- Research proposal writing course; International Foundation for Science/ICIPE (2015)

Scientific integrity / ethics in science activity (1.8 ECTS)

- Online course on research ethics; Global Health Network (2019)

PE&RC Annual meetings, seminars and the PE&RC weekend (1 ECTS)

- PE&RC First years weekend (2017)
- WGS/PE&RC Last stretch of the PhD programme (2019)

National scientific meetings / local seminars / discussion groups (9.8 ECTS)

- Mbita science club meetings; ICIPE (2017-2019)

- One health entomology group meetings during periods spent; WUR (2017, 2019)
- Scientific meetings: malaria, human health and all themes; ICIPE (2017-2019)

International symposia, workshops and conferences (6.3 ECTS)

- 5th Pan African Mosquito Control Association (PAMCA); Victoria Falls, Zimbabwe (2018)
- 12th Neglected Tropical Diseases (NTD) conference; Nairobi, Kenya (2018)
- 11th European Congress of Tropical and International Medicine (ECTMIH); Liverpool, United Kingdom (2019)

Societally relevant exposure (0.3 ECTS)

- Wellcome-Trust malaria program for primary students documentary interview (2019)

Lecturing / supervision of practicals / tutorials (3.6 ECTS)

- World malaria day-community training (2017, 2019)
- Malaria introductory course for primary students (2019)

MSc thesis supervision (6 ECTS)

- Use of transfluthrin shields in the control of outdoor malaria transmission
- Inferiority trials on LUMIN8 light trap against standard CDC light traps on major malaria vectors in western Kenya
- Attractiveness of MB5 odour-bait in comparison to humans to the major malaria vectors in western Kenya
- Quantification of protection conferred against indoor-biting malaria vectors using controlled release design (insecticide) repellent blocks

The research described in this thesis was conducted under the East Africa Collaboration on mosquito push-pull that was financially supported by the Innovative Vector Control Consortium (IVCC), UK.

Cover design by Hans M. Smid and Margaret Mendi Njoroge Thesis layout by Margaret Mendi Njoroge Printed by GVO Printers & Designers, Ede, The Netherlands